Characterization of Process Variability and Robust Optimization of Analog Circuits

by

Daihyun Lim

Bachelor of Science, Seoul National University, February 1999 Master of Science, Massachusetts Institute of Technology, June 2004

Submitted to the Department of Electrical Engineering and Computer Science

in partial fulfillment of the requirements for the degree of

Doctor of Philosophy

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

September 2008

© Massachusetts Institute of Technology 2008. All rights reserved.

Author

Department of Electrical Engineering and Computer Science

Professor of Electrical Engineering and Computer Science Thesis Supervisor

Accepted by

Terry P. Orlando Professor of Electrical Engineering Graduate Officer

, A .

MASSACHUSETTS INSTITUTE OF TECHNOLOGY	
OCT 2 2 2008	ARCHIVES
LIBRARIES	

Characterization of Process Variability and Robust Optimization of Analog Circuits

by

Daihyun Lim

Submitted to the Department of Electrical Engineering and Computer Science on August 29, 2008, in partial fulfillment of the requirements for the degree of Doctor of Philosophy

Abstract

Continuous scaling of CMOS technology has enabled dramatic performance enhancement of CMOS devices and has provided speed, power, and density improvement in both digital and analog circuits. CMOS millimeter-wave applications operating at more than 50GHz frequencies has become viable in sub-100nm CMOS technologies, providing advantages in cost and high density integration compared to other heterogeneous technologies such as SiGe and III-V compound semiconductors.

However, as the operating frequency of CMOS circuits increases, it becomes more difficult to obtain sufficiently wide operating ranges for robust operation in essential analog building blocks such as voltage-controlled oscillators (VCOs) and frequency dividers. The fluctuations of circuit parameters caused by the random and systematic variations in key manufacturing steps become more significant in nano-scale technologies. The process variation of circuit performance is quickly becoming one of the main concerns in high performance analog design.

In this thesis, we show design and analysis of a VCO and frequency divider operating beyond 70GHz in a 65nm SOI CMOS technology. The VCO and frequency divider employ design techniques enlarging frequency operating ranges to improve the robustness of circuit operation. Circuit performance is measured from a number of die samples to identify the statistical properties of performance variation. A back-propagation of variation (BPV) scheme based on sensitivity analysis of circuit performance is proposed to extract critical circuit parameter variation using statistical measurement results of the frequency divider.

We analyze functional failure caused by performance variability, and propose dynamic and static optimization methods to improve parametric yield. An external bias control is utilized to dynamically tune the divider operating range and to compensate for performance variation. A novel time delay model of a differential CML buffer is proposed to functionally approximate the maximum operating frequency of the frequency divider, which dramatically reduces computational cost of parametric yield estimation. The functional approximation enables the optimization of the VCO and frequency divider parametric yield with a reasonable amount of simulation time. Thesis Supervisor: Duane Boning Title: Professor of Electrical Engineering and Computer Science

Acknowledgments

Remembering my first day at MIT, I was so nervous since I was totally misled by the sign "Welcome to Hell" at the Harvard Bridge. Of course, you can feel like being in the middle of Hell when there is 30 inches of snow all around you in January. However, certainly it cannot be Hell when you have so many talented people around you, and they are willing to help you do what you really want to do. I feel very fortunate and privileged to spend my past six years at MIT, and this conclusion would have never been possible without the help and support of great people who I have met during my PhD.

First of all, I would like to extend my wholehearted gratitude to my advisor, Professor Duane Boning, for his inspiring guidance and sincere mentoring from the beginning through the end of my PhD. This thesis would not have been possible without his support and encouragement. He provided me lots of good advice and clear guidance throughout my PhD, and I truly enjoyed all the group meetings and FCRP trips with him. I feel extremely fortunate to have him as my PhD advisor. Thank you, Duane.

I would like to thank my thesis committee members (and also RQE committee), Vladimir Stojanovic and Luca Daniel. Vladimir always answered my questions very quickly throughout my PhD, and gave me good feedback on my research in committee meetings. I appreciate his inspiring advice about the goal of PhD study after my RQE presentation, and he also provided nice lectures on high speed serial link design in summer 2008 that would be very helpful in my future career. Luca gave me good feedback on this thesis and I also had many inspiring meetings with Luca throughout my PhD. His lectures on numerical simulation in fall 2004 was one of the best classes I have taken at MIT, and the things I learned from the class were very helpful to conduct numerical analysis in this thesis.

I would like to thank all the IBM people I met during my two internships at IBM Fishkill. They gave me a tremendous amount of help in finishing my PhD. I cannot thank enough Jonghae Kim for his encouragement, research ideas, and sound advice about life as an engineer. I learned a lot from him about ISSCC paper writing. Jean-Olivier Plouchart was my manager during my first internship, and he gave me lots of help in designing the VCO and frequency divider. He also provided me the measurement of the input sensitivity curve of the frequency divider in this thesis. Daeik Kim gave me lots of help in the measurement and analysis of the VCO and divider. He also provided me a tremendous amount of help in publishing many papers. Choongyeun Cho provided me many nice plots in this thesis, and in particular, Section 3.4.2 is mostly based on the collaboration with him. Weipeng Li, my colleague during my first internship at IBM Fishkill, designed the VCO in this work. Robert Trzcinski helped me to setup the equipment for the measurement of the frequency divider. I would like to thank Ava Wan and Jae-eun Park who helped me to enjoy my internships at IBM as much as possible.

I would like to thank my fellow Boning group members: Karthik Balakrishnan, Albert Chang, Nigel Drego, Wei Fan, Joy Johnson, and Hayden Taylor. I owe numerous thanks especially to circuit subgroup members for countless helpful discussions and good company. I am so glad that I could share lots of great moments with them. Thank you Karthik, Nigel, and Albert. You are my great friends. I also thank former Boning group members: Karen Gonzalez Valentine Gettings, Ali Farahanchi, Xioalin Xie, Hong Cai, Ajay Somani, Kwaku Abrokwah, Mehdi Gazor, and Daniel Truque.

I thank all of my friends who have made my life at MIT truly enjoyable. I would like to thank my Korean MIT friends: Kevin Kyoungbeom Ryu, Hyemin Jung, Junghoon Lee, Jungwoo Joh, Jaekyu Lee, Jinwook Chung, Min Park, Seongmoo Heo, Junmo Kim, Yong-il Shin, Minkyu Kim, and Jungwon Kim.

I would like to thank my old CSG friends who enlightened my early days at MIT: Edward Suh, Jaewook Lee, and Charlie O'Donell. Special thanks to my Master's thesis advisor Srini Devadas who gave me solid background to perform my PhD research.

My golf buddies provided me great chances to escape from real world frustrations. Thanks go to: Hyungwoo Lee, Kwanhong Lee, Woosik Kim, Hyuksang Kwon, Tairin Hahn, and Dogeun Kim. I would like to thank my KGSA basketball friends with whom I shared great moments with lots of sweats. Thank you Jounggeun Lim, Sungjoo Bae, Jongchul Moon, Taeyoon Kim, Daegeun Hwang, Younggun Go, and Jungeun Park.

I would like to thank Korea Foundation of Advanced Study (KFAS) for the funding in my first year and financial support for five years of my PhD.

This thesis could not have been possible without my mom's unconditional love and support. As I have been studying abroad, I have not been able to do my duties as a son. I am so indebted for the love and care she gave to me. Thank you mom.

Last, I would like to thank my wife, Jerin Gu, for her sincere love, encouragement, and support. We have been through countless happy and tough moments together, and I am so glad that both of us survived MIT successfully! Thank for being with me Jerin.

This research has been supported by Focus Center Research Program (FCRP) Center for Circuits, Systems, and Solutions (C2S2).

Contents

1	Intr	roduction	21
	1.1	Motivation: Process Variation in millimeter-Wave CMOS Design	25
	1.2	Sources of Variation	27
		1.2.1 Front End Variation	29
		1.2.2 Back End Variation	37
		1.2.3 Environmental Variation	39
	1.3	Statistical Design of Analog Circuits	40
		1.3.1 Statistical Analysis of Performance Variation	41
		1.3.2 Variation-Aware Circuit Design	45
	1.4	Contributions and Organization of This Thesis	48
	1.5	Summary	50
2	\mathbf{Des}	ign of Manufacturable mm-Wave PLL Building Blocks	51
2	Des 2.1	sign of Manufacturable mm-Wave PLL Building Blocks Previous Work on mm-Wave VCO and Frequency Divider	51 54
2	Des 2.1 2.2	Sign of Manufacturable mm-Wave PLL Building Blocks Previous Work on mm-Wave VCO and Frequency Divider Design of mm-Wave Voltage Controlled Oscillator	51 54 57
2	Des 2.1 2.2 2.3	Sign of Manufacturable mm-Wave PLL Building Blocks Previous Work on mm-Wave VCO and Frequency Divider Design of mm-Wave Voltage Controlled Oscillator Design of mm-Wave Frequency Divider	51 54 57 60
2	Des 2.1 2.2 2.3	sign of Manufacturable mm-Wave PLL Building Blocks Previous Work on mm-Wave VCO and Frequency Divider	51 54 57 60 60
2	Des 2.1 2.2 2.3	Sign of Manufacturable mm-Wave PLL Building Blocks Previous Work on mm-Wave VCO and Frequency Divider	51 54 57 60 60 63
2	Des 2.1 2.2 2.3 2.4	Sign of Manufacturable mm-Wave PLL Building Blocks Previous Work on mm-Wave VCO and Frequency Divider	51 54 57 60 60 63 69
2	Des 2.1 2.2 2.3 2.4 2.5	sign of Manufacturable mm-Wave PLL Building Blocks Previous Work on mm-Wave VCO and Frequency Divider Design of mm-Wave Voltage Controlled Oscillator Design of mm-Wave Frequency Divider 2.3.1 CML Based D-FF Architecture 2.3.2 Analysis of Self-Oscillation Frequency and Locking Range Implementation of PLL Front-End Circuit Summary	 51 54 57 60 60 63 69 72
2	Des 2.1 2.2 2.3 2.4 2.5 Stat	sign of Manufacturable mm-Wave PLL Building Blocks Previous Work on mm-Wave VCO and Frequency Divider Design of mm-Wave Voltage Controlled Oscillator Design of mm-Wave Frequency Divider 2.3.1 CML Based D-FF Architecture 2.3.2 Analysis of Self-Oscillation Frequency and Locking Range Implementation of PLL Front-End Circuit Summary tistical Measurement and Analysis	 51 54 57 60 60 63 69 72 75

		3.1.1	Measurement Setup for the mm-Wave PLL Front-End	77
	3.2	Nomin	al Performance Measurement of PLL Front-End	78
		3.2.1	Measurement of VCO Performance	79
		3.2.2	Measurement of Frequency Divider Performance	81
	3.3	Variat	ion of VCO and Frequency Divider Performance	82
		3.3.1	Variation of VCO Performance	84
		3.3.2	Variation of Frequency Divider Performance	85
	3.4	Statist	ical Analysis of Measurement Data	89
		3.4.1	Back-Propagation of Variation	92
		3.4.2	Decomposition of Frequency Divider Variation	96
	3.5	Summ	ary	100
4	Rob	oust Oj	ptimization of Analog Circuits	103
	4.1	Backg	round: Yield Optimization of Analog Integrated Circuits	105
		4.1.1	Dynamic Optimization of Analog Circuit Yield	107
		4.1.2	Static Optimization of Analog Circuit Yield	108
	4.2	Dynan	nic Yield Optimization of PLL Front-End	109
		4.2.1	Failure Mechanism of PLL Front-End Circuit	112
		4.2.2	Optimal Bias Condition for Maximum Yield	114
	4.3	Static	Yield Optimization of PLL Front-End	117
		4.3.1	Estimation of Frequency Divider Performance	119
		4.3.2	Estimation of Parametric Yield of PLL Front-End	126
		4.3.3	Comparison of Dynamic and Static Yield Optimization \ldots	130
	4.4	Summ	ary	131
5	Con	clusio	n and Future Work	135
	5.1	Summ	ary of Thesis	135
	5.2	Future	e Work	137
		5.2.1	Future Trend of Process Variation in Analog Circuits	138
		5.2.2	Technology Benchmarking for Analog Circuit Design	140
		5.2.3	Digitally-Assisted Robust Analog Design	143

A Pro	cess Variability and Statistical Yield Analysis of High-Speed	
Rin	g Oscillators 1	L 4 5
A.1	Analog Design for RF Test	147
A.2	Measurement Setup	150
A.3	Statistical RF Test	155
A.4	Yield Analysis	156
A.5	Summary	159

List of Figures

1-1	Trend of mm-wave CMOS performance in near future [8]	22
1-2	Design for manufacturability of analog circuits by the interaction be-	
	tween variation characterization, yield assessment, and robust opti-	
	mization	23
1-3	Millimeter-wave applications	26
1-4	High-speed building blocks in a phase-locked loop (PLL). The PLL	
	front-end consists of the VCO and frequency divider, circled at left. $% \mathcal{A} = \mathcal{A}$.	27
1-5	Sources of process variation in front-end of line	30
1-6	Normalized threshold voltage variation of NMOS device in 65nm CMOS	
	technology over 13 wafer (15 dies per wafer)	32
1-7	Sub-threshold current-based within-die threshold voltage variation mea-	
	surement [34]	34
1-8	Spatial correlation model of channel length variation [40]	35
1-9	Pattern-dependent CMP variation of metal and oxide thickness in pla-	
	narization steps.	38
1-10	Variation of contact resistivity identifying parametric- and open- fail-	
	ure of contacts and vias.	39
1-11	Worst case analysis of circuit performance based on the corners of	
	circuit parameters. Corners may be based on statistical models, e.g.	
	$\pm 3\sigma$ bound, or $\pm 1.5\sigma$ bounds as pictured here	42
1-12	Improvement in a higher order RSM, through the translation of circuit	
	parameters into intermediate parameters by understanding underlying	
	physical circuit behaviors.	44

2-1	VCO structures	52
2-2	Frequency divider structures	53
2-3	PLL high-speed front-end circuit consisting of a VCO, a frequency	
	divider, and a buffer amplifier	54
2-4	Schematic of complementary LC-VCO design using an AMOS varactor	
	$(M_7 \text{ and } M_8)$ and a C-shape inductor (L)	59
2-5	Divide-by-two using a D-FF based on CML latches	61
2-6	Schematic of the 90GHz frequency divider based on a static CML ar-	
	chitecture	62
2-7	Input sensitivity curve of a frequency divider indicating self-oscillation	
	frequency and maximum operating frequency for a given input signal	
	power requirement as set by the available VCO output power. \ldots	63
2-8	Small signal analysis of the static frequency divider in the self-oscillation	
	mode	65
2-9	Behavioral model of the injection locking oscillator with frequency in-	
	jection through a mixer.	66
2-10	Simulation results of the divider frequency operating ranges at different	
	V_{BIAS} conditions	68
2-11	Simulated phase noise of the divided output signal at different bias	
	conditions	69
2-12	Cross sectional diagram of 65 nm SOI CMOS technology with 10 metal	
	layers. The technology provides a Vertical Natural Capacitor (VN-	
	CAP) for high-capacitance density and symmetric characteristic	70
2-13	Layout and die photo of the frequency divider circuit in 65nm SOI	
	CMOS technology.	71
2-14	Layout and die photo of the VCO and frequency divider circuit in 65nm	
	SOI CMOS technology.	72
3-1	Ground and signal pad assignment of test circuits with physical dimen-	
	sion	77

3-2	Automated high-frequency measurement setup for VCO and frequency	
	divider samples	78
3-3	Frequency tuning range of the mm-wave VCO over a full range of V_{CTRL}	. 79
3-4	Phase noise characteristic of the VCO and the phase noise measure-	
	ment set up	80
3-5	Measured performance characteristics of the frequency divider	83
3-6	Statistical distribution of the VCO output frequency over a 300mm	
	wafer (57 die samples).	85
3-7	Minimum and maximum frequency of the VCO over a 300mm wafer.	86
3-8	Variation of self-oscillation frequency of three frequency divider alter-	
	natives at a bias condition $V_{DD}=1.6$ V and $V_{BIAS}=0.75$ V	87
3-9	Wafer-level die-to-die variation pattern of the self-oscillation frequency	
	of the divider over one 300mm wafer	88
3-10	Variation of self-oscillation frequency of the frequency divider over 11	
	wafers. Symbol represents the wafer number	89
3-11	Scatter plots of f_{SO} , for corresponding dividers, between each pair of	
	wafers	90
3-12	Statistical distribution of the self-oscillation frequency variation over	
	each wafer and the entire wafer set	91
3-13	Extracted variation of V_{TH} , load resistance, and capacitance and their	
	correlations	95
3-14	Comparison of traditional PCA (a) vs CPCA (b)	98
3-15	Comparison of cumulative variance explained by PCA and CPCA, for	
	in-line DC test data.	99
3-16	Decomposition of frequency divider self-oscillation frequency variation	
	based on constrained principal component analysis	100
4-1	Robust optimization of analog circuits considering process variability	
	and parametric yield constraints.	104
4-2	Design centering methodologies for indirect yield improvement	106

4-3	Comparison of the variation of VCO minimum frequencies and fre-	
	quency divider self-oscillation frequencies over 65 dies on one 300mm	
	wafer	110
4-4	Correlation between the variation of the VCO and frequency divider	
	over 65 dies on one 300mm wafer	111
4-5	Failure analysis of the PLL front-end circuit in the presence of signifi-	
	cant divider variation.	113
4-6	Divided output frequency of the PLL front-end circuit at different bias	
	conditions of the frequency divider.	114
4-7	VCO-divider outputs from 76 die samples over a wafer	115
4-8	Functional yield of the PLL front-end circuit at different bias conditions	
	of the frequency divider over four VCO and divider design alternatives.	116
4-9	Evaluation of maximum operating frequency of the frequency divider.	
	Sequence of transient simulation with a fine frequency step is needed	
	to evaluate the maximum operating frequency with high accuracy	118
4-10	Binary search scheme of the maximum operating frequency of the divider.	119
4-11	A CML differential delay buffer	120
4-12	Transconductance function $g(V)$ of the differential pair in Equation (4.4).	.121
4-13	Timed input and output signals of the CML buffer, and physical time	
	delay components.	122
4-14	Quadratic modeling fitting of the oscillation frequency of a CML buffer-	
	based ring oscillator.	124
4-15	Quadratic modeling fitting of the oscillation frequency of a CML buffer-	
	based ring oscillator.	126
4-16	Estimation of the parametric yield of the frequency divider based on	
	the statistics of variation parameters.	127
4-17	Procedure of yield estimation using DC Monte Carlo simulation results	
	and functional approximation.	128
4-18	Scattering plot of estimated and simulated self-oscillation and maxi-	
	mum operating frequencies in 500 Monte Carlo runs.	129

4-19	Quantile-quantile normal plots of the simulated self-oscillation and	
	maximum operating frequencies from 500 Monte Carlo runs	130
4-20	Functional yield of the frequency divider over the design space for the	
	given specifications at $V_{BIAS} = 1.05V$ and $V_{BIAS} = 1.20V$	132
4-21	Parametric yield of the frequency divider over the design space with	
	dynamic V_{BIAS} tuning.	133
5-1	Digital feedback control of V_{CTRL} and V_{BIAS} and divided frequency	
	measurement to set optimized V_{BIAS} voltage	144
A-1	Process flow diagram of analog design for RF test platform	146
A-2	A schematic diagram of 11-stage CML VCO implemented in 130nm	
	CMOS	149
A-3	VCO layout with testing environment considering RF DFT. \ldots .	149
A-4	Measurement setup diagram for the VCO.	151
A-5	Manual screen captures of VCO output spectrum with 50MHz span	
	and the phase noise at 6.94GHz	151
A-6	VCO output frequency for input current I_B and the output signal power	.152
A-7	VCO phase noise output with input current I_B sweep	153
A-8	VCO phase noise at 1, 2, 5, and 10MHz offset with I_B sweep	154
A-9	VCO phase noise at $I_B = 1$ mA and 15 mA	154
A-10	Statistical measurement on VCO maximum and minimum tunable fre-	
	quency	155
A-11	Statistical measurement on VCO phase noise with $I_B=2$ mA and 12 mA	
	@10MHz offset	157
A-12	2 VCO F_{max} ($I_B=12$ mA) and F_{min} ($I_B=2$ mA) versus phase noise distri-	
	bution	158
A-13	Variation model including wafer-level die-to-die variation and random	
	wafer-to-wafer variation.	159

List of Tables

2.1	Test vehicles for frequency dividers	71
2.2	Test vehicles for the PLL front-end	72
3.1	Comparison with state-of-the-art CMOS VCOs	81
3.2	Comparison with state-of-the-art frequency dividers operating over	
	50GHz	82
3.3	Statistics of VCO performance variation.	84
3.4	Sensitivity of the self-oscillation frequency to circuit parameters at dif-	
	ferent bias conditions.	94
3.5	Categories of in-line parameters used in the CPCA analysis	98
4.1	Test vehicles for the PLL front-end	116
4.2	Performance constraints of a 10GHz frequency divider in $0.18 \mu \mathrm{m}$ CMOS	130
5.1	Common metrics for the characterization of a digital microprocessor	
	manufacturing process.	141
A.1	Digital DFT vs. RF DFT	147
A.2	VCO Performance specification for RF DFT	150
A.3	CML VCO RF Performance Statistics	156

Chapter 1

Introduction

Over the past few decades, CMOS technology has been the mainstream in digital integrated circuit (IC) design, and in recent years CMOS technology has become the dominant path for analog IC design as well. Delay and active power consumption in digital CMOS circuits have been continuously improved by scaling of MOS transistor feature size and mobility enhancement techniques. Technology scaling has also had significant impact on analog circuit design, greatly improving the operating frequency of the active and passive components in analog building blocks in particular [9].

Figure 1-1 shows the trend of the unity current gain frequency (cutoff frequency) f_T , maximum oscillation frequency (unity power gain frequency) f_{MAX} , and effective gate length L of the CMOS devices in current and near future technologies in the ITRS road [8]. Since f_T is inversely proportional to L, as shown in Equation (1.1) below [70], the high frequency performance of the CMOS device has improved with feature size scaling.

$$f_T \approx \frac{g_m}{C_{gs}} \approx \frac{v_{sat}}{2\pi L}.$$
(1.1)

In Equation (1.1), v_{sat} is saturation carrier velocity, g_m is the maximum transconductance, and C_{gs} is gain-to-source capacitance of a CMOS device. In modern sub-100nm technology nodes, the cutoff frequency of CMOS devices is beyond several hundred giga-hertz. For example, NMOS f_T greater than 330GHz was reported recently in a 65nm technology node [61]. As a result, CMOS technology is quickly replacing SiGe



Figure 1-1: Trend of mm-wave CMOS performance in near future [8]

bipolar and III-V technologies for the implementation of analog building blocks for millimeter-wave (mm-wave) applications operating over 50GHz, because of intrinsic advantages in CMOS manufacturing cost and high density integration.

However, not every scaling impact is positive in analog circuit design. Though the cutoff frequency has increased dramatically, it has become more difficult to achieve high signal-to-noise ratio (SNR), intrinsic gain, and linearity because of reduced allowable voltage swing caused by V_{DD} scaling, degraded intrinsic gain g_m/g_{ds} and noise characteristics in sub-100nm CMOS technologies [41]. In particular, the process variability in the device and interconnect parameters has become one of the main concerns in analog circuit design in nano-scale technologies. Dynamic operating ranges of analog building blocks become smaller as nominal operating frequency increases, and therefore, parametric yield loss caused by process variation in high frequency analog building blocks has become a critical limiting factor for the yield of mixed-signal System-on-Chip (SoC) applications.

For the improvement of manufacturing yield in high performance analog circuits, new approaches for the characterization of variability in circuit parameters, the assessment of circuit performance variation, and design methodologies to maximize the parametric yield are required, as summarized in Figure 1-2. Efficient test structures are needed to capture the variability of important circuit parameters in analog circuit



Figure 1-2: Design for manufacturability of analog circuits by the interaction between variation characterization, yield assessment, and robust optimization.

design. A good test structure provides a solid information link between design and technology by supplying feedback to manufacturing to investigate control problems in manufacturing steps such as oxidation, ion implantation, and lithography, or by providing parameter statistics to designers for the estimation of parametric yield.

An accurate and computationally efficient yield assessment method for complex analog circuits is one of the important requirements for yield improvement. Estimating the performance variation caused by inter-die and within-die variation of circuit parameters is a time-consuming task, as it typically requires a large number of simulation runs to obtain a sufficient set of statistics. Computationally efficient evaluation of performance variables to accelerate simulation speed and preserve estimation accuracy is strongly needed for yield assessment.

Last, robust optimization methodologies to directly or indirectly improve the parametric yield of analog circuits by finding optimal design points and circuit condition are required. Conventionally, analog circuit design has depended intensively on a designer's circuit intuition. However, to consider complex variation impacts in design optimization, computer-aided design optimization methodologies must support design procedures to correct the causes of parametric yield loss and to reduce the risk of costly re-spins in advanced technologies.

This thesis contributes new approaches for the characterization of process vari-

ability and methodologies for the robust optimization of high-speed analog circuit design. To develop and demonstrate these approaches, the design and implementation of a manufacturable mm-wave voltage-controlled oscillator (VCO) and frequency divider as high-speed building blocks of a phase-locked loop (PLL) are presented. The design optimization of the VCO and frequency divider is focused on achieving wide frequency operating ranges to improve manufacturability against process variation.

We show that the VCO and frequency divider circuits can be used as high-speed analog benchmarking circuits to characterize the statistical properties of analog circuit performance and critical circuit parameters in a manufacturing process. This is demonstrated through the statistical measurement of the VCO and frequency divider performance over multiple wafers. We present statistical analysis of the VCO and frequency divider performance variation to characterize important systematic and random variation patterns and the primary reasons of parametric yield loss. A sensitivity-based parameter extraction scheme is suggested and tested using the performance measurement data to estimate the statistical variation in critical circuit parameters.

Finally, a key ingredient of our methodology is robust optimization and compensation for improved yield. Dynamic compensation of performance fluctuation by active circuit tuning is a general trend in the robust optimization of digital and analog circuit performance. This thesis proves that the parametric yield of the VCO and frequency divider circuits can be improved by tuning the bias condition of the frequency divider to compensate for the divider performance variation. In addition, we suggest a computationally efficient yield estimation method for the frequency divider circuit, which allows a designer to explore the entire design space and to find the maximum yielding design point within a reasonable amount of design time and computational resource. Statistical simulation results show that the parametric yield can be maximized for given statistical distributions of important circuit parameters by the yield estimation.

In this chapter, we introduce the motivation for addressing process variability in high performance IC design in Section 1.1. In Section 1.2, we review the important sources of variation in semiconductor manufacturing processes. Statistical methodologies for the analysis of variation impact on circuit performance and variation-aware design methodologies are discussed and reviewed in Section 1.3. The organization of this thesis is shown and the contributions are summarized in Section 1.4.

1.1 Motivation: Process Variation in millimeter-Wave CMOS Design

In recent years, mm-wave applications for high data-rate communication and passive imaging using the unlicensed 7GHz bandwidth around 60GHz and the 77GHz automotive radar band, respectively, have drawn significant attention; examples are pictured in Figure 1-3. Traditionally, mm-wave monolithic microwave integrated circuits (MMICs) have been implemented in heterogeneous III-IV technologies such as GaAs and InP because of their excellent high frequency performance due to higher electron mobility and higher breakdown voltages. Though CMOS devices provided superior cost effectiveness and higher levels of integration, the device performance could not provide sufficient speed to enable mm-wave operations in silicon. However, the continuous scaling of the silicon device feature size below 100nm, stress-induced mobility enhancement, and the improvement in circuit design techniques have enabled the implementation of mm-wave applications in CMOS technologies [33].

A PLL is an essential building block for high frequency clock generation in RF/mmwave and digital applications. A typical second-order PLL circuit consists of a phase and frequency detector, a charge pump, a low pass filter (LPF), a VCO, and a frequency divider. We define a *PLL front-end* as containing a VCO and a 2:1 frequency divider ¹ as depicted in Figure 1-4. Since the PLL front-end circuit operates at the highest frequency in the entire system, speed, power consumption, and noise characteristics of the PLL front-end must be considered carefully in the early stages of design.

In particular, in the mm-wave frequency regime over 50GHz, an LC-tank based VCO is a better choice compared to a ring-oscillator based VCO, due to its high

¹called as *pre-scaler* in some references



(a) 77GHz automotive radars for cruise control [1] (b) 60GHz short-range private networks [2]



(c) 60GHz high date-rate links between portable electronicsFigure 1-3: Millimeter-wave applications

frequency operation and lower phase noise characteristics. However, the frequency tuning range of an LC-VCO is significantly narrowed by the reduced capacitance ratio between a varactor and parasitics present in deeply scaled CMOS devices. Moreover, the operating range of the frequency divider becomes limited for a reduced input power due to the power loss in the interconnect between the VCO and frequency divider. Without sufficient performance margin in VCO and frequency divider designs, process variation can cause a mismatch between the frequency tuning range of the VCO and the operating range of the frequency divider, which can result in a serious functional yield problem during the total mm-wave PLL integration.

A typical high-speed frequency divider generates a self-oscillation frequency when the input of the divider is biased to DC (discussed in detail in Section 2.3.2). The selfoscillation frequency is sensitive to circuit parameters such as the threshold voltage, channel length, parasitic capacitance, and load resistance. For example, 23.9% of $3\sigma/\mu$ variation in the self-oscillation frequency of CML static frequency dividers has been reported in a 65nm SOI CMOS technology [67]. Previously, 21.5% of $3\sigma/\mu$



Figure 1-4: High-speed building blocks in a phase-locked loop (PLL). The PLL frontend consists of the VCO and frequency divider, circled at left.

in the self-oscillation frequency was reported in a 90nm SOI CMOS for a similar circuit implementation [87]. Even though systematic components of the performance variation can be reduced by circuit techniques and process control improvement, the impact of unavoidable random fluctuation in circuit parameters becomes more significant as circuit performance improvement is more dependent on the feature size scaling of the device and interconnects.

1.2 Sources of Variation

One of the key drawbacks of nanometer scale CMOS technology is the increasing degree of variability in device and interconnect parameters, which affects the performance and power consumption of digital and analog circuits. The process variability issues in high-performance CMOS in the sub-100nm regime have been studied intensively in recent publications (e.g., [10], [16], and [15]). A modern semiconductor manufacturing process can be classified into front-end of line (FEOL) and back-end of line (BEOL) clusters. The FEOL consists of manufacturing steps for active devices: lithography, implantation, oxidation, shallow-trench-isolation, etching, etc. On the other hand, the BEOL comprises the steps for creating interconnects: deposition, etching, chemical-mechanical polishing, etc. Variability can be involved in every man-

ufacturing step in front- and back-end processing.

The impact of each variation source on circuit performance is highly dependent on circuit application, and therefore, it is difficult to say which group of variation sources is providing a dominant impact. For example, when a circuit is sensitive to mismatch between device characteristics (e.g., SRAM cell static noise margin), the front-end variability is a critical concern for overall parametric yield. Close to 90% of canonical path delay is also caused by device variability instead of variability in interconnect parameters [82]. On the other hand, in global signal paths such as clock distribution nets, the variability in interconnect wires contributes to overall clock skew. In high frequency VCO design, the variability in metal inductors is a critical contributor to the variation in resonant frequency.

The classification of process variation based on a spatial range is also important. The total variation of a parameter P can be divided into lot-to-lot, waferto-wafer, across-wafer (die-to-die), within-die, and random components as shown in Equation (1.2).

$$\Delta P = \Delta P_{L2L} + \Delta P_{W2W} + \Delta P_{D2D} + \Delta P_{WD} + \Delta P_R \tag{1.2}$$

Each circuit parameter has a different portion of variance from each range of variation depending on the physical characteristics of major variability contributors. For example, a significant portion of the variation in the threshold voltage is random since the threshold voltage variation is contributed by the intrinsic randomness of dopant fluctuation in the channel. On the other hand, the variation in interconnect parameters is highly systematic since the variation in BEOL is highly dependent on layout pattern-density and wafer-level non-uniformity of manufacturing conditions.

In this section, we classify the sources of variation in IC performance by the variation in FEOL, BEOL, and environmental changes. Physical characteristics of each variation source are reviewed individually. The variation sources in FEOL include threshold voltage, channel length, and miscellaneous sources such as channel stress, oxide thickness, and aging. In BEOL, chemical-mechanical polishing (CMP) is the main source of variation in the physical dimension of metal wires. Lithographyinduced variation and the fluctuation of contact resistance are also contributors to performance variation in the BEOL. The IC performance is also sensitive to the variation of operating environment such as power supply voltage and surrounding temperature. The impact of the environmental variability on circuit performance must be characterized precisely to devise an adaptive compensation scheme for supply- or temperature-induced performance variation.

1.2.1 Front End Variation

Figure 1-5 shows the physical manifestations of the front-end variability in device characteristics. The primary causes of device variability are: random dopant fluctuation, oxide thickness variation, and line-edge roughness of gate polysilicon. For designers, the physical sources of variation from various manufacturing steps in FEOL such as photo-lithography, etching, and ion-implantation can be aggregated and expressed by the variation in two critical device parameters used in analog and digital circuit simulation: threshold voltage and channel length. We discuss physical causes and suggested models of variation in the threshold voltage and channel length.

Threshold Voltage Variation

The major process parameters affecting the threshold voltage of a MOS device are the doping concentration in the channel of the device and of the gate material, the channel length, and the thickness of the silicon-dioxide insulating film. As a result of the continuous scaling of MOSFET feature size, the threshold voltage variation due to random dopant fluctuation has become more significant. Manufacturing steps, such as ion implantation and annealing, have intrinsic random characteristics since the dopant atoms penetrate and diffuse through the crystal lattice in a random fashion. As a result, the final number and location of the dopant atoms are random variables. Variation due to the limited number of dopant atoms becomes a significant cause of the threshold voltage fluctuation.



Figure 1-5: Sources of process variation in front-end of line

The variability of threshold voltage becomes a serious issue in digital circuit design since power consumption due to leakage current has increased significantly in high performance CMOS devices. Since the leakage current is exponentially dependent on the threshold voltage of CMOS devices, total power becomes more sensitive to the increased variability of the threshold voltage [31]. In analog design, the within-die variation of the threshold voltage causes mismatch in device pairs, which degrades the circuit performance of SRAM cells, differential amplifiers, comparators, and other sensitive circuits. Mean-value shift of the threshold voltage between dies, or inter-die variation, can also affect operation of circuits that depend on parameter values rather than matching in circuit performance.

The intrinsic variation of the threshold voltage associated with depletion charge, implantation, fixed oxide charge, and oxide granularity was predicted and treated analytically in the classical Pelgrom's model [85]. The feature size dependency of the standard deviation of the threshold voltage mismatch was experimentally confirmed and widely used in the prediction of variability in circuit performance. Based on the theoretically predicted $(L_{eff}W_{eff})^{-0.5}$ dependency, a useful empirical expression of $\sigma(V_{TH})$ in sub-100nm gate length devices was developed in [7] as Equation (1.3). Model coefficients were fitted to the results from atomistic simulation considering the discrete random number of dopant atoms and their random positioning over a three-dimensional channel space.

$$\sigma(V_{TH}) = 3.19 \times 10^{-8} \frac{t_{OX} N^{0.4}}{\sqrt{L_{eff} W_{eff}}} [V], \qquad (1.3)$$

where t_{OX} is the thickness of the silicon dioxide film, and N is the number of dopant atoms in the channel.

While the within-die variation becomes more significant in scaled processes, dieto-die and wafer-to-wafer variation components are still the main contributors in overall threshold voltage variability. Figure 1-6 shows the threshold voltage variation of minimum channel length devices in a 65nm CMOS technology over 13 wafers. From each wafer, 15 dies are sampled for threshold voltage measurement. The threshold voltage is measured based on a threshold current method, giving the gate-to-source voltage (V_{GS}) when the drain current crosses a reference current level while the drainto-source voltage (V_{DS}) is close to zero (e.g., at 0.1V). In Figure 1-6(a), each curve shows the die-to-die variation in each wafer. The similarity of each curve shows strong wafer-to-wafer spatial correlation of the threshold voltage variation. The highly systematic wafer-level die-to-die variation results in the strong correlation between two devices within the same die, as shown in Figure 1-6(b).

While the threshold current method for the measurement of the threshold voltage is convenient, it is difficult to avoid including also some impact of from channel length variation to achieve the measurement of pure threshold voltage variation. A V_{TH} measurement scheme indifferent to ΔL to capture the stochastic behavior of the within-die threshold voltage variation has been presented in [34]. The method in [34] measures a sub-threshold leakage current ratio between CMOS devices at a bias condition where the ratio is highly sensitive to the threshold voltage difference while the channel length variation impact is negligible. Figure 1-7(a) shows the ratio between the sensitivities of sub-threshold current to threshold voltage change and



(b) Correlation between two devices

Figure 1-6: Normalized threshold voltage variation of NMOS device in 65nm CMOS technology over 13 wafer (15 dies per wafer).

channel length change at different V_{GS} voltages. The results in Figure 1-7(b) show a highly stochastic trend of the within-die variation of the threshold voltage, with no notable spatial correlation between the devices. The correlation coefficient between two devices is close to zero and indifferent to separation distance between the devices.

Channel Length Variation

The variation in the effective channel length of the devices directly affects the saturation current in strong inversion. Channel length variation is induced by various manufacturing steps, including mask printing, exposure in photo-lithography, etching, spacer definition, and source and drain implantation.

The channel length variation can be divided into systematic and random portions. As a dominant factor of the channel length variation, the variability in photolithography is known to be highly systematic. As a result, wafer-level channel length variation patterns induced by temperature and material thickness non-uniformity in photo-lithography are likely to be systematic [81]. The systematic variation components can be decomposed using sophisticated statistical methods introduced in [97]. The randomness of the channel length variation is highly affected by line-edge roughness (LER) in gate printing. The reasons for the increased LER in the deeply scaled processes include the random variation in the number of incoming photons during exposure and the contrast of aerial images, as well as variation in the absorption rate, chemical reactivity, and molecular composition of the photo-resist. The impact of LER is certainly growing as the minimum channel length of the devices shrinks down below 100nm where sub-wavelength resolution assistant features (SRAFs) are in use to overcome the systematic variations due to limitations of the patterning processes [80] [30].

The within-die variation component of the channel length variation is quickly becoming a significant factor in performance variation, in particular, for mismatchsensitive circuit examples such as SRAM cells and differential amplifiers. Intuitively, the transistors laid out close to each other are likely to have similar characteristics



(a) Sensitivity ratio between threshold voltage and channel length variation



(b) Spatial correlation as a function of separation distance

Figure 1-7: Sub-threshold current-based within-die threshold voltage variation measurement [34].



Figure 1-8: Spatial correlation model of channel length variation [40].

compared to the transistors far apart. Various formula have been introduced to model the spatial correlation of the channel length variation in terms of the distance between the transistors. An early empirical form of the spatial correlation in channel length variation was suggested in Pelgrom's model [85]. An empirical piece-wise linear form of spatial correlation with a few parameters was introduced, as shown in Figure 1-8 [40]. In this model, the correlation coefficient as a function of separation distance xis defined as in Equation (1.4):

$$\rho(x) = \begin{cases} 1 - \frac{x}{X_L} (1 - \rho_B), & x \le X_L \\ \rho_B & x \ge X_L \end{cases}$$
(1.4)

where X_L and ρ_B are model parameters.

Theoretically, the piece-wise linear form as a spatial correlation function cannot result in the valid generation of random numbers satisfying any distance-based correlation matrix from layout. Theoretically valid forms of the spatial correlation function and a robust extraction method of the model parameters for the spatial correlation function were studied in [107]. A general family of spatial correlation functions is as follows,

$$\rho(v) = 2\left(\frac{bv}{2}\right)^{s-1} K_{s-1}(bv)\Gamma(s-1)^{-1},$$
(1.5)

where K is the modified Bessel function of the second kind, Γ is the gamma function, b and s are fitting parameters that decide the shape of the correlation function, and v is the separation distance between the devices.

Miscellaneous Front-End Variation Sources

Historically, silicon dioxide film thickness is extremely tightly controlled in a thermal oxidation process. However, as the feature size and oxide thickness scaling has continued, the oxide thickness has reached an atomic-level scale, near 10Å, which corresponds to approximately five atomic layers of silicon dioxide. Physical limitations such as interface roughness and oxide layer non-uniformity lead to increased variability of the effective oxide thickness.

Variation in the oxide thickness can affect the electrical properties of the device such as the threshold voltage and carrier mobility. In particular, the gate leakage current of a thin film CMOS device by quantum-mechanical carrier tunneling through the gate oxide is exponentially sensitive to the dielectric thickness. Beyond the 65nm technology node, the gate leakage current is comparable to the sub-threshold leakage through the channel, and the variability in the gate leakage current is quickly becoming a growing source of circuit performance variation [24]. The impact of the oxide thickness variation on device electrical properties can be characterized by atomic-level 3D device simulation [7].

Electron mobility improvement by mechanical stress has been another driving force for device performance enhancement, in addition to feature size scaling. Shallow trench isolation (STI) is one dominant source of mechanical stress in a MOSFET channel. The mismatch in thermal expansion coefficients of different materials and the volume expansion of silicon dioxide cause the channel stress in the oxidation step that follows the formation of STI. The STI-induced stress and its impact on electron and hole mobility is highly dependent on the layout, in particular, on the size of the active area and the distance to the STI edge [81]. Therefore, the variability of the electrical characteristics of the devices by mechanical stress is highly systematic and layout-dependent.
1.2.2 Back End Variation

As CMOS technology scales, the contribution of interconnect parameters to the overall circuit performance becomes more important. Accurate modeling of parasitic capacitance and resistance of interconnect wires becomes essential to achieve accurate estimation of system performance. In particular, modeling the variation of the interconnect performance is critical to estimate the manufacturing yield of high performance digital and analog circuits. The recent survey in [79] shows that the overall time delay variation increases, and in particular, the portion of the within-die variation in the interconnect parameters becomes more significant.

Variation in the BEOL cluster of manufacturing causes the variability of the interconnect parameters. Modern sub-100nm process nodes provide more than 10 metal layers in their BEOL stacks. The BEOL flow consists of a repeated sequence of metal and dielectric layer deposition steps. After the deposition of a dielectric layer, photolithography and etching steps follow to build holes and trenches for vias and metal lines. After a thin barrier is deposited, electroplating fills the spaces for the vias and wires with copper. The electroplating leaves unwanted over-filled copper, and a chemical-mechanical polishing (CMP) step follows to remove the over-filled metal and planarize the surface for the following metallization.

In previous studies, CMP is found to be a critical source of systematic withindie variation in thickness of metal wires [71]. Figure 1-9 portrays the variation in thickness of metal wires by dishing and erosion after CMP, which is highly dependent on the feature size and pattern density of the layout. Pattern-dependency can also be observed in the CMP of dielectric materials for the planarization of shallowtrench isolation (STI) and inter-dielectric layers (ILDs). Physical understanding and a pattern-dependent fitting model of the dielectric CMP variation is provided in [106].

Furthermore, the topography of over-filled metal after electroplating is highly correlated to after-CMP topography, since the amount of excess metal impacts clearing time and degree of over-polish. The pattern-dependent variation of electroplating



Figure 1-9: Pattern-dependent CMP variation of metal and oxide thickness in planarization steps.

topography and its impact on CMP was studied in [20]. To improve the planarity of the post-CMP surface, the pattern density of every metal layer in the layout must be regulated by strict design rule checks (DRCs) at the design stage. Insertion of CMP dummies improves the planarity significantly, but the parasitic capacitance added by the metal dummies must be considered in post-layout simulation for accurate performance estimation.

Since CMP variation is highly layout pattern-dependent, a significant portion of interconnect variation is systematic. Therefore, interconnect performance variation can be predicted using the pre-processing of interconnect parameters, such as parasitic resistance and capacitance, based on the CMP variation models. The impact of metal wire and ILD thickness variation on time delay was investigated in [72], where simulations with the CMP pre-processor showed 30% increase in the delay of 5mm bus lines due to copper CMP variation.

Corresponding with the scaling of device feature size, the physical dimensions of interconnect have been shrunk down as well, to fully exploit the increased density of active devices. The resistivity of contacts and vias becomes one of the critical factors in circuit performance, since the number of metal layers, contacts, and vias increases as the circuit complexity increases in the advanced process nodes. Openfailure in the contact and via is a common problem caused by defects. In recent technologies, the parametric variation of the contact resistivity has become one of the important variation factors in circuit performance estimation, as shown in Figure 1-10. Efficient test structures must be introduced to characterize the statistical properties of the contact resistance variation and its layout-dependency in deeply scaled CMOS processes and the measured characteristics must be included in circuit simulation for better accuracy in interconnect modeling [19].



Figure 1-10: Variation of contact resistivity identifying parametric- and open- failure of contacts and vias.

1.2.3 Environmental Variation

Another aspect of IC statistical modeling is to include the effect of environmental parameters such as temperature and power supply voltages, which contribute significantly to circuit performance variability. The effect of supply voltage fluctuations can easily be modeled in circuit simulators (e.g., SPICE) since power sources are always explicitly defined in the circuit models. The modeling of temperature impact depends on the circuit simulator and device model used in the simulation. If a circuit simulator has a built-in temperature model, then temperature can be treated as one of the circuit parameters. Otherwise, an external temperature model has to be defined in addition to the circuit simulator, to pre-calculate the simulation parameters influenced by temperature.

An empirical variation model of the delay, active power consumption, and stand-by leakage of a ring oscillator to power supply and temperature fluctuation is suggested and evaluated in [98]. While the delay and active power consumption are fairly linear to the power supply and temperature variation, the leakage power shows an exponential dependency on the environmental variations. Linear fitting of the frequency, active power variation, and log(leakage) of a seven-stage ring oscillator example and excellent fitting accuracy are shown in [81].

Typically, both supply voltage and temperature can be defined as independent random variables, which influence all components in a circuit. Thus temperature and supply voltages can be directly added to the group of variability factors in statistical circuit models [109]. However, while physical variability has random and systematic components, environmental variability is largely deterministic since it depends on the dynamic operation of circuit building blocks. Therefore, the study of environmental variation naturally focuses on the efficient prediction and bounding of the variation in work load, power consumption, and operating frequency of circuit building blocks.

As a dynamic compensation scheme for environmental variations, techniques such as frequency and power supply scaling, localized toggling, and migrating computation to different hardware units can be utilized for dynamic thermal management in multicore microprocessor application [96].

1.3 Statistical Design of Analog Circuits

Process variation is certainly not a newfound issue in semiconductor manufacturing; it was a primary concern in the early days (1970's) of IC manufacturing [36] [84]. The relationship between the control of manufacturing steps and the variability of relevant transistor parameters was investigated, particularly to help guide the improvement in the control of the manufacturing conditions in order to reduce the fluctuation of critical circuit parameters. Recently, process variation has drawn significant attention, resulting in a sharp increase in the number of publications in various circuit and computer-aided-design (CAD) conferences and journals. Process variation impact on the performance of circuits and micro-architectures, and mitigation methodologies for the variation impact, have been heavily discussed in many references (e.g., [16] and [15]). Compared to the process variation studies in the 1970's, today's variation research is typically focused on design-related issues: efficient characterization of variation parameters, development of variation models in critical circuit parameters, assessment of parametric yield at design time, and the statistical optimization of circuit performance.

In this section, we review parametric yield assessment techniques and variationaware design methodologies to mitigate the variation impact and to improve the manufacturing yield.

1.3.1 Statistical Analysis of Performance Variation

Statistical circuit analysis requires both the characterization of the variability in device and interconnect behaviors, and the ability to translate the device and interconnect variability to circuit performance variability, i.e., statistical circuit simulation. Worst-case corner based circuit simulation and performance optimization have been the most commonly used method to achieve the robustness of analog circuit design [32]. The purpose of the corner-based analysis is to identify the corners in parameter space which corresponds to the worst circuit performance. The tolerance of circuit parameters are finite and bounded, and the simulations at each corner of the parameter space can estimate the boundary of the circuit performance space as shown in Figure 1-11.

The corner-based analysis leads circuit designers to a well-posed problem: meeting the performance specifications at every possible corner of active and passive component models. However, there are several practical issues that limit the usage of worst-case based design optimization.

• The worst-case design optimization often results in a pessimistic over-design.



Figure 1-11: Worst case analysis of circuit performance based on the corners of circuit parameters. Corners may be based on statistical models, e.g. $\pm 3\sigma$ bound, or $\pm 1.5\sigma$ bounds as pictured here.

A subset of circuit parameters may be tightly correlated with each other since they are affected by the same manufacturing steps, and therefore, a subset of the variation corner have a negligible likelihood of occurrence in reality. The design optimization over all possible parameter corners can lead to the degradation of nominal circuit performance at more realistic corners, and the reduction of overall parametric yield.

- Computational cost of considering all possible parameter corners is too expensive. In particular, if the within-die variation of circuit components becomes a primary concern, the combinational increase in the number of parameter corners is exponential and it becomes impractical to consider all parameter corners for each circuit component.
- Non-linearity of circuit behavior cannot guarantee the one-to-one matching of circuit parameter space corners and performance space corners.

Alternatively, Monte Carlo analysis is commonly used for the characterization of the performance distribution, based on the given statistics of circuit parameters. In addition to a nominal parameter set of device and interconnect models (e.g., BSIM4), the additional variables for critical variation parameters are sampled from given statistics for each iteration of circuit simulation. The accuracy of the estimates from Monte

Carlo analysis depends on the number of iterations, since the confidence intervals of the estimates are a function of the number of Monte Carlo runs. Monte Carlo analysis captures the non-linearity of complex circuit behaviors and provides a realistic distribution of performance variables to determine the parametric yield in design stages.

Modern device and interconnect models include numerous parameters for finetuning of the models to match the measured hardware characteristics. As the number of model parameters increases, it becomes more difficult to consider the joint probability density function of the overall parameter set in Monte Carlo analysis. Principal component analysis is a commonly used technique to preserve the correlation between device and interconnect parameters while reducing the number of independent parameters by representing all the parameters as a linear combination of orthonormal principal components. For example, in [73], from the characterization data of 192 parameters of an individual test device, six principal components can be extracted to reduce the number of sampling variables significantly while explaining 96% of the variance and preserving the correlation in the entire parameter set.

Even though Monte Carlo analysis results in accurate estimates of performance variation, it is computationally expensive to repeat the circuit simulation a sufficient number of times to achieve accuracy in the statistical estimates. Recently, a number of alternative simulation methods have been proposed to achieve the accuracy of full Monte Carlo simulation while reducing the computational cost dramatically [64] [105]. Even though the methods show good error performance in digital logic simulation, it is difficult to explain the variation in complex analog circuits which have a number of non-linear performance variables.

A response surface model (RSM) is a commonly used functional approximation for the estimation of circuit performance to bypass the full circuit simulation in Monte Carlo analysis [17]. RSMs are constructed by computing the coefficients of linear or quadratic polynomials which represent the performance variables so as to minimize the mean-square errors at sampled simulation points. The number of the sampled evaluation points must be equal to or larger than the number of coefficients to obtain a unique or approximated solution. Various methods to generate response surface models and the evaluation of model accuracy are described in [17].

The accuracy of the RSM-based yield estimation can be improved simply by using a higher order fitting model. In particular, when the variation ranges of the circuit parameters become large, the linear assumption of output variables over the circuit parameter range is no longer valid. By increasing the order of the RSM, the nonlinearity of circuit behaviors can be captured to improve the fitting accuracy. A risk of over-fitting is an issue, potentially affecting the stability of model coefficient calculation in higher order model fitting.

Alternatively, the estimation accuracy of the RSM can be improved by understanding the underlying physical meaning of circuit parameters, and introducing translation variables between the circuit parameters and output variables. For example, as shown in Figure 1-12, when the underlying non-linear or discontinuous circuit behavior is known, conducting lower order RSM fitting to intermediate parameters and evaluating the output variables based on the translated intermediate parameters can improve both accuracy and stability of the model fitting. In Chapter 4, we show the accuracy enhancement for the RSM of current-mode logic (CML) buffer time delay and frequency divider performance by introducing intermediate variables which translate fundamental circuit parameters such as sheet resistance, threshold voltage, and channel length to physical time delay components.



Figure 1-12: Improvement in a higher order RSM, through the translation of circuit parameters into intermediate parameters by understanding underlying physical circuit behaviors.

1.3.2 Variation-Aware Circuit Design

The goal of the analysis of variation impacts and yield estimation techniques is to maximize the parametric yield of digital and analog circuits by tuning design parameters appropriately. From the analysis of performance variation, circuit designers must have optimization strategies for minimizing the impact of parameter variation on the circuit performance and maximizing the yield. In this section, we summarize several trends in variation-aware design methodologies in analog circuits and largescale micro-architectures: equation-based robust optimization, post-silicon tuning, and regular fabric design style.

The equation-based robust optimization proceeds by converting the yield optimization problem to a classical non-linear programming problem based on appropriate approximation of circuit behaviors to special functions [108]. A standard formulation of robust optimization problems can be as follows.

minimize
$$\sup_{u \in U} f_0(y, u)$$

subject to $\sup_{u \in U} f_i(y, u) \le 0, \ i = 1, \dots, m,$ (1.6)

where y is the design variable, u represents the variation variables, and the set U is a space describing the uncertainty of u. An optimization technique called *geometric programming* (GP) formulates the circuit behavior by *posynomial* functions whose general form is as follows [18]:

$$f(x) = \sum_{k=1}^{K} c_k x_1^{a_{1k}} x_2^{a_{2k}} \dots x_n^{a_{nk}}, \qquad (1.7)$$

where $c_k > 0$, $a_{ik} \in \mathbf{R}$, and x_1, \ldots, x_n are real positive variables. It is known that GPbased problems can be solved efficiently and accurately due to recent optimization theory progress [46], and the robust optimization problem size based on GP grows linearly with the number of uncertainty parameters [108]. The equation-based robust optimization can find an optimal design point without scanning through the entire design space. However, the posynomial fitting error of non-linear performance variables for various analog circuits cab pose significant limitations on the optimization accuracy.

Post-silicon tuning techniques can be used to compensate for circuit performance variation after fabrication. Compensation of the variation in time delay and leakage power consumption of digital logic using adaptive body bias (ABB) voltage control was studied in [103]. Bidirectional ABB is used for both NMOS and PMOS devices to increase the percentage of dies that meet both frequency requirements and leakage constraints. The control of NMOS and PMOS body bias values individually for each die can reduce die-to-die frequency variations (σ/μ) by an order of magnitude, and 100% of the sampled dies are shown to become acceptable in meeting the clock frequency and leakage power specifications. Furthermore, a theoretical foundation for joint design-time and post-silicon optimization using adaptive body-bias control considering the variation of delay and leakage power in digital circuits was studied in [69]. The problem is cast as an adjustable robust linear program and solved in a computationally efficient way. Experimental results indicate that a designer can greatly benefit from the synergy of combining the design-time and post-silicon optimization techniques, due to the ability of post silicon optimization solutions to tune the circuit parameters to the realization of uncertain data which is not possible to know at design time.

Alternatively, it is widely recognized that physical regularity in layout improves the control of variability, and regularity in IC design has received extensive attention in recent years. In sub-100nm CMOS technologies, the design rule check (DRC) enforces regularity in layout by restricting the pattern density and routing direction of polysilicon gates and interconnect wires. Various forms of regular design styles have been proposed, including restrictive design rules, PLA-based fabrics, homogeneous gate-array-like fabrics, and heterogeneous regular fabrics [100]. One way to systematically enhance the design regularity is to implement ICs using a library of regular logic bricks. The advantage of the design style based on the regular logic bricks for the manufacturability of digital circuits has been studied in [86]. Micro-architecture design based on a pre-characterized brick library can provide not only the reduction of the variability in speed and power consumption, but also more predictable performance distribution at the design stage by using the pre-characterized data. The trade-off between the improvement of parametric yield and design costs such as area, power, and complexity introduced by regularity must be investigated.

In this thesis, we specifically focus on variation characterization coupled with robust optimization methods for high performance analog circuits. We classify the robust optimization of analog circuits into two categories in Chapter 4: static and dynamic optimization. In the static optimization, design-time optimization in other words, designers optimize the design variables, such as device sizes and resistance and capacitance values, at the design stage to achieve a maximized estimated parametric yield. The static optimization assumes that the statistical properties of variation parameters (e.g., V_{TH} , ΔL , and R_{SH}) are given from the pre-characterization. Computational cost of statistical simulation (e.g., Monte Carlo) is the most serious issue in the static optimization strategy. To consider the complex behavior of process variation such as within-die variation and its spatial correlation in the yield estimation, there must be a sufficient number of iterations in Monte Carlo simulation. In particular, the number of performance variables in analog circuits is typically much larger than that of digital circuits, and thus yield estimation demands more Monte Carlo runs. Furthermore, since the variability of circuit-level parameters is dependent on the design variables, the distribution of the performance variation and parametric yield must be evaluated at each design point.

The dynamic or post-silicon optimization, on the other hand, finds tunable knobs in a creatively designed circuit to compensate for the performance variability after fabrication. Increasing the degree of freedom in output characteristic tuning involves additional circuit-level design techniques and feedback control issues. The dynamic optimization technique is highly dependent on circuit specifics, and each circuit needs a different strategy to tune output characteristics. For example, to improve the matching properties in differential pairs of circuit components, redundant design and proper selection mechanism that finds closely matched pairs can be introduced [65]. A power supply voltage, a reference current, and a bias voltage can be used as a tunable knob for many analog circuits such as oscillators and amplifiers. However, since the performance variables of an analog circuit are tightly correlated with each other, the trade-off between the performance variables must be carefully considered in the dynamic optimization schemes.

1.4 Contributions and Organization of This Thesis

This thesis is organized as follows. Chapter 1 enumerates the sources of variation in VLSI manufacturing process, their impact on high-speed analog circuit performance. Statistical approaches for robust IC design including the statistical analysis of circuit performance variation and variation-aware design methodologies are shown in this chapter.

Chapter 2 describes the design issues in key mm-wave building blocks, the VCO and frequency divider, considering variability of circuit parameters to improve the manufacturability of PLL design. A 70GHz PLL front-end circuit consisting of the VCO and frequency divider has been implemented in a 65nm SOI CMOS technology. The device and circuit are optimized to provide sufficiently wide operating ranges of the PLL building blocks in a mm-wave frequency regime to improve manufacturability in the deeply scaled CMOS process.

Chapter 3 shows statistical measurement results of the mm-wave building blocks. The operating frequency and power consumption of PLL front-end circuits have been measured in an automated mm-wave measurement setup over 300mm wafers. We report individual performance variation characteristics of the VCO and frequency divider. Furthermore, a back propagation of variation (BPV) method to characterize the variation in important circuit parameters is presented based on the measurement data, and the statistical properties of the performance variation over multiple wafers are analyzed using a constrained principal component analysis (CPCA) to identify a dominant factor of the performance variation.

Chapter 4 analyzes the statistical properties of the VCO and frequency divider circuits and shows dynamic and static robust optimization approaches to achieve a maximum parametric yield of the PLL front-end circuit. Statistical measurement results show that the frequency divider variation is the major concern for overall yield of the PLL front-end, and we provide analysis of the failure mechanism caused by process variation. The bias condition of the frequency divider can be dynamically tuned to change the operating range of the divider and maximize the yield of the PLL front-end, while satisfying output phase noise and power specifications.

Furthermore, we provide an empirical quadratic time-delay model based on physical time delay and frequency locking range components using DC circuit parameters, which shows excellent fitting accuracy to the maximum operating frequency variation over design migration and local parameter variation. Experimental results show that the parametric yield of the PLL front-end can be estimated over the entire design space in a reasonable amount of simulation time to achieve the optimum design point for the maximum yield.

Chapter 5 concludes this thesis and provides suggests for process variation research in mm-wave circuit design. In Appendix A, process variability of a high-speed ringoscillator based VCO and statistical analysis of experimental results are discussed.

The primary contributions of this thesis can be summarized as follows.

- Design and implementation of a manufacturable VCO and a frequency divider operating up to 90GHz in CMOS technology.
- Measurement and analysis of manufacturing variation in VCO and frequency divider performance and power.
- Extraction of critical circuit parameters based on the statistical measurement results of the frequency divider at multiple bias conditions.
- Dynamic optimization of the VCO and frequency divider parametric yield by bias tuning of the frequency divider.
- Suggestion of a computationally efficient time delay model for the fast yield estimation of the PLL front-end and statistical optimization.

1.5 Summary

In this chapter, we introduced the process variability issues in high-speed analog circuit design. The operating frequency of CMOS applications has reached the mmwave regime over 50GHz mainly by continuous scaling of the MOS transistor feature size. The aggressive scaling has introduced a parameter variability issue in digital and analog circuit design. Various manufacturing steps contribute to the variation of CMOS device and interconnect parameters and overall circuit performance. Therefore, a benchmarking task to characterize the performance level variation properties is strongly needed to provide useful guidelines for circuit designer to optimize the parametric yield of designs. To optimize the parametric performance yield, statistical design methodologies are used, whose goal is not only to meet the design specifications for nominal circuit performance, but also to achieve desired parametric yield despite the existence of parameter variation.

Chapter 2

Design of Manufacturable mm-Wave PLL Building Blocks

A PLL is an essential building block to generate a reference clock signal for most RF/mm-wave applications and micro-architectures [11]. The PLL needs a VCO and a frequency divider for local clock generation and output frequency leverage from an input reference frequency, respectively. Typically, in an integer-N PLL with a deep frequency division ratio greater than 1000, the frequency divider consists of a *prescaler* providing an initial divide-by-two function followed by a sub-divider for deep sub-division. Since the VCO and pre-scaler are operating at the highest frequency in the system, they draw significant design attention in order to meet the specifications of frequency operating range, phase noise, power consumption, and other requirements.

The operating frequency of a VCO achievable in CMOS has been continuously improved by the scaling of the device feature size. Generally, a designer has two choices for the implementation of the VCO, depending on required performance specifications: an LC-tank based VCO (LC-VCO) and a ring-oscillator based VCO (ring-VCO) as shown in Figure 2-1. The LC-VCO utilizes the resonance of an LC-tank with the cancellation of parasitic resistance by active devices for the oscillation frequency generation (Figure 2-1(a)). Since the LC-tank consists of passive components, the LC-VCO provides high operating frequency and low phase noise, which make the LC-VCO suitable for RF/mm-wave applications. However, the LC-VCO has a nar-



Figure 2-1: VCO structures

row frequency tuning range, large area for passives, and static power consumption as disadvantages.

The ring-VCO consists of a ring of variable delay stages with proper inversion as shown in Figure 2-1(b). Each delay stage can be implemented by a single-ended or differential inverter with load resistance or bias current control to change the time delay. Typically, the ring-VCO provides a wide frequency tuning range, a multiphase output, small power and area consumption, and easy implementation. However, the main disadvantage of the ring-VCO is larger phase noise compared to the LC-VCO with the same power budget. Furthermore, the osciilation frequency and power consumption of the ring-VCO is highly sensitive to process variability of the devices, which is relatively much larger than that of the passive components in the LC-VCO. Appendix A shows the experimental results of a ring-VCO performance variation, including oscillation frequency, power consumption, and output phase noise, in a 90nm SOI CMOS technology. As a result, the ring-VCO is suitable for clock generation for digital logic and micro-processors, which have tight area and power budgets and loose phase noise specifications.

There are various design methodologies for the divide-by-two frequency divider implementation. The most commonly used topology is a master and slave latch based architecture as shown in Figure 2-2(a). From a digital circuit point of view, this structure operates as a flip-flop to change its state at a rising or falling edge of an input clock, to output half of the input clock frequency. For mm-wave frequency division, this structure can be used as a frequency divider based on an injection-locking ring



Figure 2-2: Frequency divider structures

oscillator to provide a wide frequency operating range and easy implementation. The detailed operation of this structure as an injection-locking ring oscillator is described in Section 2.3. Similarly, an LC-tank based injection locking oscillator can be used as a mm-wave frequency divider (Figure 2-2(b)). The output frequency of the LC-tank is locked to half of the frequency of an injected current source through the mixing behavior of active devices [59]. The pros and cons of the LC injection-locking frequency divider are similar to those of the LC-VCO: high and accurate frequency operating range with excellent phase noise performance but a narrow operating range with large area and power consumption.

In this chapter, we describe the design and analysis of a manufacturable mm-wave PLL front-end circuit implemented in 65nm SOI CMOS technology. The mm-wave PLL front-end consists of an LC-VCO, a 2:1 current-mode-logic (CML) frequency divider in a master and slave latch based architecture, and a cascoded buffer amplifier between the VCO and frequency divider as shown in Figure 2-3. One of the challenging issues in mm-wave CMOS VCO and frequency divider design is manufacturability; a wide VCO frequency tuning range (FTR) is required in order to cover the desired frequency range in the presence of process variation. In addition, the frequency operating range of the divider must be sufficiently wide for proper frequency division over the entire VCO tuning range. Conventionally, the frequency divider has a significant amount of performance variation since the circuit performance is based on the electrical properties of the CMOS devices with minimum feature size. Therefore, the frequency operating range variation of the divider circuit must be precisely



Figure 2-3: PLL high-speed front-end circuit consisting of a VCO, a frequency divider, and a buffer amplifier.

characterized and optimized during design.

Section 2.1 shows previous design examples of a mm-wave VCO and frequency divider and their output characteristics. Section 2.2 and 2.3 describe design details of the mm-wave VCO and frequency divider examined in this work. Section 2.4 depicts the actual implementation of the VCO and frequency divider test vehicles in a 65nm CMOS technology.

2.1 Previous Work on mm-Wave VCO and Frequency Divider

High frequency VCOs and frequency dividers are essential building blocks for most mm-wave applications. Typically, SiGe and III-V technologies have been appropriate selections for the implementation of VCOs and frequency dividers operating above 50GHz frequency [62] [104]. However, as the minimum feature size of CMOS devices has shrunk down below 100nm, VCOs and frequency dividers operating at over 50GHz have been reported in CMOS technologies. Here, we review several design examples of mm-wave CMOS LC-VCOs and frequency dividers and discuss their output characteristics in terms of frequency operating ranges, which are important from a process variability perspective.

The output frequency of an LC-VCO is mainly determined by the resonance frequency of the LC-tank, and therefore, a mm-wave output frequency is still achievable by precise design of passive components without using sub-100nm technologies. However, to obtain a reasonable frequency tuning range, the parasitic capacitance in active devices must be minimized since the frequency tuning range is determined by the ratio between the varactor capacitance tuning range and the lumped capacitance at the VCO output nodes. In [99], Tiebout et al. reported a 1V 51GHz VCO in a 0.12μ m bulk CMOS technology with low-k dielectric. The area of the VCO circuit was minimized to reduce the parasitic capacitance and to constrain the component dimensions well below the wavelength of a 50GHz signal (6mm) so that the lumped model assumptions are valid. An excellent power consumption of 1mW was achieved, while the frequency tuning range is from 50.89 to 51.64GHz at $I_{TAIL} = 1mA$, resulting in an FTR of only 1.46%.

A 63GHz VCO was implemented in a 0.25μ m CMOS technology in [68] by Liu et al. A push-push cross-coupled topology was employed to achieve high frequency operation, high output power, and good phase noise. Coplanar waveguides and asymmetrical coplanar strips were used to reduce substrate loss, which results in the degradation of the inductor Q. A MOS varactor is used for frequency tuning. Due to the limitation by the parasitic capacitance at output nodes, the achieved FTR of 2.5GHz (3.97%) is still very small.

A 40GHz CMOS VCO in a 90nm SOI CMOS technology was reported in [37] by Fong et al. This LC-VCO employed a complementary topology for phase noise improvement and power consumption reduction. The VCO used an accumulation-mode MOS varactor for a wide capacitance tuning ratio, and a single-loop copper inductor with high-Q to achieve good phase noise. The measured FTR of this VCO is 15.8% (from 34.9 to 40.9GHz), which provided excellent manufacturability for commercial production. The VCO in this thesis inherited key design features of this VCO to achieve 77GHz center frequency with a wide FTR and reasonably good phase noise in a 65nm SOI CMOS technology.

For the sound joint operation of the VCO and frequency divider, the frequency operating range of the frequency divider must fully cover the frequency tuning range of the VCO with sufficient margin. To accommodate the increase of the system operating frequency, the requirement of the maximum operating frequency for the frequency divider becomes extremely high in mm-wave applications. Furthermore, as the operating frequency increases, the attenuation of the input signal from the VCO to the frequency divider becomes worse, and the operating range of the divider become narrower due to the reduced input signal power. The reduced operating range of the frequency divider quickly becomes a serious issue for manufacturability in commercial production.

Typically, an LC-resonator based injection-locking frequency divider (LC-ILFD) is used to achieve a high operating frequency [89]. However, the locking range of the LC-ILFD is generally narrow, and the limited operating range causes a parametric yield loss by process and environmental variations. On the other hand, a tail-injection ring oscillator-based ILFD (ring-ILFD) has a wider locking range, occupies smaller chip area, and has lower power consumption [74]. Here, we introduce several design examples of both LC-ILFDs and ring-ILFDs operating at multi-gigahertz range in CMOS technologies.

A 40GHz regenerative divider with an operating range of 2.3GHz (5.75%) was implemented in 0.18μ m CMOS by Lee et al. in [59]. The frequency divider uses a Miller divider scheme with inductive peaking, which provided both high frequency operation and wider locking range compared to a conventional injection-locking frequency divider.

A CML frequency divider with static architecture was implemented in 0.12μ m and 90nm SOI CMOS technologies in [88] and [87], respectively. Even though the divider employed a static divider architecture, the bias condition and load resistance values were configured to generate a self-oscillation frequency without an input signal. At around twice the frequency of the self-oscillation signal, the divider operates as a ring-oscillation based ILFD. Because of the multiple injection sources of an input signal, the ring-oscillation based ILFD provides a wide frequency locking range compared to the LC-resonator based ILFD. The detailed analysis of the locking range characteristics of the ring-oscillator based ILFD is presented in [75].

2.2 Design of mm-Wave Voltage Controlled Oscillator

Achieving both high frequency operation and a wide frequency tuning range are difficult challenges in mm-wave VCO design. In general, it is desirable for the resonator to include a large multi-turn inductor for better Q to achieve better phase noise performance, and a large varactor with high C_{max}/C_{min} ratio for a wider frequency tuning range. The frequency tuning range (FTR) of an LC-VCO is proportional to a capacitance tuning range of the LC-tank. The capacitance tuning range (CTR) can be defined using the varactor capacitance and fixed parasitic capacitance at output nodes as follows [38]:

$$FTR \propto CTR = \frac{C_{v,max} + C_{fix}}{C_{v,min} + C_{fix}},$$
(2.1)

where C_v is the varactor capacitance and C_{fix} is the fixed parasitic capacitance.

However, at high frequency as in the mm-wave regime, both inductance L and varactor capacitance C_v in the resonator are limited to small values, because the oscillation frequency f_O is determined by

$$f_O = \frac{1}{2\pi\sqrt{L(C_v + \Delta C)}} \tag{2.2}$$

where ΔC is the change in capacitance achievable by varactor tuning. When the center frequency $f_{O,C}$ ($\Delta C=0$) is fixed, L must be minimized to achieve a large frequency tuning range since the capacitance tuning range increases monotonically as C_v increases. However, to reduce output phase noise, L must be maximized to

achieve a high Q of the inductor, since the quality factor of the inductor dominates the overall quality factor of the LC tank. This imposes a design trade-off between frequency tuning range and phase noise.

The VCO design in this work focuses on achieving a wide frequency tuning range for improved manufacturability against process variation. Several circuit components are employed to improve the frequency tuning range.

- A large accumulation-mode MOS (AMOS) varactor with a good C_{max}/C_{min} ratio and Q.
- Relaxed pitch SOI transistors to reduce parasitic capacitance (C_{fix}) by 30% compared with normal pitch bulk CMOS devices.
- A single-turn top-level-only inductor achieving small inductance with high Q.

Figure 2-4 shows the schematic diagram of the LC-tank based VCO used in this work. The VCO circuit employs a complementary design to achieve the wide FTR and power reduction. It has a symmetric complementary negative g_m (NFET for M_1 and M_2 , and PFET for M_3 and M_4), and an LC tank with a C-shape inductor L and AMOS varactors M_7 and M_8 . The VCO outputs V_{OUT+} and V_{OUT-} are buffered by source followers M_5 and M_6 . The oscillating output node of the LC-VCO is DC-biased at around half of V_{DD} by proper sizing of the PFET and NFET. This biasing provides maximum capacitance tuning range to the AMOS whose capacitance is controlled by the voltage between V_{CTRL} and the biased output nodes. V_{CTRL} can be changed from zero to V_{DD} to fully tune the AMOS capacitance.

In this VCO, we use SOI CMOS devices which have lower parasitic diffusion-tosubstrate capacitance due to buried oxide layer isolation. The transistors are multifingered, and the gate pitch of the fingers are stretched to enhance carrier mobility and to reduce parasitic capacitance between gate and source (drain) terminals. The reduced parasitic capacitance at output nodes obtained by using the relaxed pitch SOI CMOS devices provides a large capacitance tuning range for the LC-tank, and as a result, a wide frequency tuning range of the VCO is achievable.



Figure 2-4: Schematic of complementary LC-VCO design using an AMOS varactor $(M_7 \text{ and } M_8)$ and a C-shape inductor (L).

The use of a small inductor allows larger varactors for wider capacitance tuning. The smaller size inductor also results in lower substrate capacitance due to smaller physical area. A single-turn structure in this VCO eliminates inter-winding capacitance. The inductor has been implemented in a 1.2μ m thick copper top-level metal layer to reduce parasitic capacitance to substrate, and to achieve highest possible Qin this CMOS technology.

However, since the skin depth of copper at 80GHz is 0.16μ m which is more than six times thinner than the top metal layer, the resistance of the inductor is likely to be decided by the skin depth. Stacking lower level metals reduces substrate isolation without improving series resistance due to the skin effect, and hence would also degrade the Q of the inductor. The degraded Q by the skin effect limits the phase noise performance of the VCO in mm-wave operation.

While the wide capacitance tuning range of the AMOS is achievable by using PFETs for proper biasing, the PFETs contribute additional parasitic capacitance to output nodes, lowering the maximum attainable oscillation frequency. However, the complementary design using PFETs consumes less power and provides better phase noise compared to a design using cross-coupled NFETs with inductive loads [44].

2.3 Design of mm-Wave Frequency Divider

Achieving both high frequency division and a wide frequency operating range of the frequency divider requires proper selection of divider architecture, device-level optimization, and circuit parameter tuning. An injection-locking frequency divider (ILFD) is the most commonly used architecture for mm-wave frequency division. Conventionally, the injection-locking mechanism is based on either an LC oscillator or ring oscillator depending on system requirements. The LC-ILFD provides higher frequency operation compared to the ring-ILFD by using the resonant frequency of an LC-tank. The ring-ILFD has an advantage in frequency operating range since the locking range of the ring-ILFD is much wider than that of the LC-ILFD within the same power budget.

By using a 65nm SOI CMOS technology whose NFET cut-off frequency is more than 300GHz, mm-wave frequency division over 50GHz is achievable in a ring-ILFD architecture. A wide operating range of the ring-ILFD is beneficial for improving the manufacturability of the divider against process variation. In this work, we employ a ring-ILFD as the 90GHz frequency divider. The ring-ILFD can be implemented using a conventional D-FF architecture with variety of features for speed enhancement. The detailed description of the divider architecture and analysis of divider behavior are in following sections.

2.3.1 CML Based D-FF Architecture

The 2:1 frequency divider in this work uses a master and slave latch architecture to construct the D-FF, as shown in Figure 2-5. From a digital circuit point of view, the D-FF is toggled at a rising or falling edge of an input clock to output half of the input clock frequency. The maximum operating frequency is determined mainly by the load



Figure 2-5: Divide-by-two using a D-FF based on CML latches

capacitance at output nodes and their driving currents. To achieve faster operation, a current-mode-logic (CML) latch is employed. A CML circuit uses resistor loads instead of conventional PMOS loads, or the polysilicon resistor loads provide smaller parasitic capacitance for high-speed operation compared to PMOS loads. Additional speed benefits can be obtained by using shunt inductive peaking at output nodes of CML circuits.

Figure 2-6 shows a schematic diagram of the mm-wave CMOS frequency divider used in this work. The bottom transistors $(M_9 \text{ to } M_{12})$ are sized to $20\mu\text{m}$ to drive up to 20mA bias and dynamic currents, and to satisfy reliability requirements. The transistors for differential pairs $(M_1, M_2, M_5, \text{ and } M_6)$ are sized to $10\mu\text{m}$ to obtain an appropriate g_m for the desired frequency range. The size of the latch transistors $(M_3, M_4, M_7, \text{ and } M_8)$ is reduced to $8\mu\text{m}$ to decrease capacitive loading at output nodes while satisfying the self-oscillation condition given by Equation (2.3) with 230 Ω unsilicided polysilicon resistors [95]:

$$g_{m,L} \cdot R_L > 1, \tag{2.3}$$



Figure 2-6: Schematic of the 90GHz frequency divider based on a static CML architecture.

where $g_{m,L}$ is the transconductance of the latch devices, and R_L is the resistance of the output loads. Together, these circuit and technology enhancements enable exceptionally high speed operation of the frequency divider up to 90GHz.

A cascoded amplifier with 100 Ω poly resistor loads is inserted as a buffer between the VCO and the frequency divider. A 200fF state-of-the-art vertical-naturalcapacitor (VNCAP) is used for AC coupling of the VCO output signals to divider inputs. The VNCAP provides large capacitance density and highly symmetric configuration suitable for RF applications [25]. The buffer amplifier isolates the VCO output from large capacitive output loading including VNCAP parasitic capacitance to substrate (~20fF) and divider input capacitance (~10fF), and provides a gain of four at the VCO center frequency for power loss compensation.

The on-chip AC coupling by the VNCAP enables V_{BIAS} to be controlled externally to change the bias currents in the CML latches and tune the operating range and output signal swing of the frequency divider. We show the advantage of tuning the operating range by the V_{BIAS} control in Chapter 4.



Figure 2-7: Input sensitivity curve of a frequency divider indicating self-oscillation frequency and maximum operating frequency for a given input signal power requirement as set by the available VCO output power.

2.3.2 Analysis of Self-Oscillation Frequency and Locking Range

The performance of a frequency divider can be fully characterized by measuring an input sensitivity curve as shown in Figure 2-7. At a certain input frequency of the divider, the corresponding point of the sensitivity curve indicates the minimum required power of an input signal to lock the divider output to the desired frequency (half of the input frequency). An injection-locking frequency divider generates a natural frequency, or self-oscillation frequency (f_{SO}) , when there is no injected input signal. When the offset between the input frequency and $2 \cdot f_{SO}$ increases, the minimum required input signal to overcome the natural frequency of the circuit and drive to the desired output frequency also increases. As a result, the input sensitivity curve becomes V-shaped, and the bottom of the V-shape curve indicates $2 \cdot f_{SO}$. When the input signal power is fixed or specified, the minimum and maximum operating frequencies of the divider can be characterized by the input sensitivity curve as in Figure 2-7.

To set the frequency operating range of the frequency divider to a desired region, an analytic form of the self-oscillation frequency is strongly needed. When the input of the divider is tied to DC, a properly biased CML divider generates the self-oscillation frequency by constructing a four stage oscillation loop. In Figure 2-8, clock inputs are tied to ground and each device is approximated as a linear element based on a small signal assumption. Each stage consists of a transconductance component (g_{mD}) , load resistance (R_L) , output capacitance (C_P) , and negative resistance $(-1/g_{mL})$ by the latch device. To satisfy Barkhausen's criterion for oscillation, the loop gain must be unity with 360 degree phase shift, and therefore, each delay stage must provide a 90 degree phase shift, assuming that all stages are identical. To result in 90 degree phase shift at each stage, the negative resistance must effectively cancel out the load resistance, and the output load impedance must become purely capacitive. Utilizing the oscillation condition, a first-order estimate of the self-oscillation frequency ω_{SO} is as follows.

$$\omega_{SO} = \sqrt[4]{\frac{g_{m1}g_{m2}g_{m3}g_{m4}}{C_{p1}C_{p2}C_{p3}C_{p4}}} \approx \frac{1}{R_L C_P} \cdot \frac{g_{mD}}{g_{mL}}$$
(2.4)

In Equation (2.4), g_{mi} is a transconductance of each differential pair, and C_{pi} is the lumped capacitance at each output node. The impact of mismatch in differential pairs on the self-oscillation frequency is reduced by averaging. By assuming that mismatch between stages is negligible, the self-oscillation frequency can be approximated as a function of effective load resistance R_L , lumped output capacitance C_P , and the ratio between transconductance of differential pair and latch devices g_{mD}/g_{mL} as in Equation (2.4).

The calculation of the self-oscillation frequency above is based on a small-signal and linear assumption. However, when the output signal swing becomes large as I_{TAIL} increases, the output signal starts to be rectified at V_{DD} and $V_{DD} - I_{TAIL} \cdot R_L$, and the linear assumption does not hold anymore. In transitions, the output capacitor is charged up through the load resistor R_L and the transition time is is largely governed by $R_L \cdot C_P$ delay. Therefore, the self-oscillation frequency is highly dependent on the circuit parameters I_{TAIL} , g_m/C_p , and $R_L \cdot C_P$, and their relationship is non-linear and



Figure 2-8: Small signal analysis of the static frequency divider in the self-oscillation mode

complex. In Section 4.3.1, we suggest a quadratic response surface model based on the parameters above to predict the self-oscillation frequency of the divider at each design point and bias condition.

The CML frequency divider operates as a free-running ring oscillator in the self-oscillation mode. When an external oscillating signal is injected into the self-oscillating loop, injection pulling and locking phenomena occur to change the oscillation frequency of the loop, which has become strongly correlated with the injected frequency depending on the signal power of the injection signal [4]. The injection locking scheme becomes the most commonly used frequency division technique in mm-wave frequency generation, due to its high frequency capability.

Figure 2-9 explains how an injection-locking oscillator works as a frequency divider. When there is no injection signal, the loop is in a steady-state satisfying Barkhausen's oscillation condition at ω_O . When an external signal is injected through the mixer, the loop oscillation frequency changes depending on the power of the in-



Figure 2-9: Behavioral model of the injection locking oscillator with frequency injection through a mixer.

jected signal. A weak injection signal pulls the loop oscillation without locking it. A pulled but unlocked oscillator traces a rich set of dynamics, which are of theoretical interest but of little practical consequence [75]. The oscillation loop settles at a different steady-state when the injection signal is sufficiently large to lock the loop. The output frequency of the mixer is both the sum and difference of the injected and locked output frequencies. Through a low-pass filter H(jf), the higher frequency term $(\omega_O + \omega_I)$ is eliminated, and the locked output frequency is equal to the difference $\omega_O - \omega_I$, and as a result, the locked output frequency becomes half of the injected signal frequency.

The frequency locking range (FLR) ω_L of the injection locking divider is the maximum frequency offset of the injected input frequency ω_I from $2 \cdot \omega_{SO}$, that continues to lock the divider output at half of the injected frequency for a given injected input power. The FLR is an important characteristic since it determines the maximum and minimum operating frequencies of the divider. Typically, ω_L must be greater than three or four times of the frequency tuning range of the VCO to guarantee the robustness of the VCO-divider operation considering process and environmental variations.

The FLR of both LC-ILFD and ring-ILFD architecture has been intensively stud-

ied in many references (e.g., [75], [90], [92], and [13]), as the injection locking divider has become more important as a feasible solution of mm-wave frequency division in CMOS technologies. A general analytic form of ω_L was suggested in [92] and [13] as follows:

$$\omega_L \approx k \cdot \frac{\omega_{SO}}{2Q} \cdot \frac{I_{INJ}}{I_{OSC}}.$$
(2.5)

In Equation (2.5), Q is the quality factor of the oscillation loop, I_{INJ} is the peak value of the injection current, I_{OSC} is the peak value of the self-oscillation current (equal to the bias current I_{TAIL}), and k is a constant depending on circuit configuration. For an LC-tank based injection-locking divider, Q is the quality factor of the LC-resonant tank. In an injection-locking ring oscillator, Q is the slope of the phase versus frequency characteristic of the loop transfer function H(f) in Figure 2-9, as follows:

$$Q = \frac{d\measuredangle H(j\omega)}{d\omega}\big|_{\omega=\omega_{SO}}$$
(2.6)

Since ω_L is inversely proportional to the quality factor Q, the LC-tank based injectionlocking divider has much narrower FLR compared to the injection-locking ring oscillator that has a lower Q.

From Equation (2.4), the self-oscillation frequency is proportional to the transconductance of differential pair devices, which is proportional to $\sqrt{I_{TAIL}}$. From Equation (2.5), ω_L is roughly proportional to $1/\sqrt{I_{TAIL}}$, and therefore, a larger bias current results in the reduction of the locking range of the divider while increasing the selfoscillation frequency.

In Figure 2-10, the simulated operating ranges of the frequency divider at varying V_{BIAS} voltage from 0.5 to 0.9V shows that the divider operating range is highly sensitive to V_{BIAS} . Figure 2-10(a) shows the change in sensitivity curves as V_{BIAS} changes. Figure 2-10(b) highlights that increasing V_{BIAS} reduces the maximum operating frequency of the divider for a given input signal power. As a result, the frequency operating range of the divider can be externally tuned dynamically by V_{BIAS} . This tunability is used to compensate for the performance variation of the divider.

On the other hand, the output voltage swing V_{SW} of the divider is proportional



(b) Frequency operating range

Figure 2-10: Simulation results of the divider frequency operating ranges at different V_{BIAS} conditions.



Figure 2-11: Simulated phase noise of the divided output signal at different bias conditions.

to I_{TAIL} , and for better signal-to-noise ratio (SNR) of the frequency divider output, a larger V_{SW} is preferred. Therefore, there is a trade-off between the operating range and the noise characteristic of the CML frequency divider. The bias condition (V_{BIAS}) can be optimized based on the specific requirements of phase noise. Figure 2-11 shows the change of the simulated output phase noise of the frequency divider over a sweep of V_{BIAS} . The output phase noise decreases as V_{BIAS} increases when the devices whose gates are connected to the clock input are in saturation. Beyond the boundary of saturation and triode regions of the clock devices, the phase noise starts to degrade, since the drain to source resistance of the clock devices becomes finite and the drain current noise of the differential pair and latch devices starts to appear in output signals. This characteristic provides a clear boundary of the phase noise enhancement achievable by V_{BIAS} increase, and is to be used as an optimization constraint in dynamic and static optimization of the circuit yield in Chapter 4.

2.4 Implementation of PLL Front-End Circuit

Test vehicles of the mm-wave PLL front-end circuit are implemented in 65nm SOI CMOS technology. Figure 2-12 shows the cross-sectional diagram of the SOI CMOS



Figure 2-12: Cross sectional diagram of 65nm SOI CMOS technology with 10 metal layers. The technology provides a Vertical Natural Capacitor (VNCAP) for high-capacitance density and symmetric characteristic.

technology used in this work. Compared to bulk CMOS, FETs in SOI CMOS technology have lower parasitic diffusion-to-substrate capacitance due to buried oxide layer isolation. The reduced parasitic capacitance is advantageous for both VCO and frequency divider design to achieve high operating frequencies [54]. The technology has 10 metal layers, and the top metal layer is eight times thicker than M1 and M2 metal layers, which is suitable for the implementation of high-Q inductors and high frequency pads.

Figure 2-13 shows the layout and die photo of the frequency divider circuit. The placement of the ten transistors is optimized to minimize the area of metal wires to reduce the parasitic capacitance to substrate. Calculation from technology information shows that the interconnection through the top metal layer provides the optimum RC delay even though it adds parasitic resistance and capacitance from the vias though the metal stack. The shunt-peaking inductors are implemented using 300μ m interconnect wires working as slab inductors by way of eddy current, to provide 330pH inductance for bandwidth enhancement.

We have three different vehicles for testing the frequency divider as listed in Table 2.1. DIV_1 has an initial layout with optimized devices sizes and relaxed placement and routing. DIV_2 has the same device sizing, and the area of interconnect wires is minimized to reduce wiring capacitance. On top of the DIV_2 layout, DIV_3 has 300



Figure 2-13: Layout and die photo of the frequency divider circuit in 65nm SOI CMOS technology.

Type	Description
DIV ₁	Parasitic-aware initial design
DIV ₂	Compact placement and routing
	for reduced wiring area
DIV ₃	With inductive shunt peaking

Table 2.1: Test vehicles for frequency dividers.

 μ m slab inductors at each of the four output nodes to enhance the output bandwidth. Each vehicle has a pair of GSGSGSGSG pad sets for high frequency measurement.

We implement four different types of test vehicles for the PLL front-end containing both the VCO and frequency divider, as summarized in Table 2.2. The test vehicles have two different types of VCOs: VCO₁ and VCO₂. Both VCOs have the same LC-tank, but different sizing of NFETs and PFETs such that NFET/PFET sizes of VCO₁ and VCO₂ are 14μ m/28 μ m and 17μ m/34 μ m, respectively. Because of the difference in the parasitic capacitance of the devices, the center frequencies of VCO₁ and VCO₂ are 65 and 71GHz, respectively.

Figure 2-14 shows the layout and die photo of the combination of the VCO and frequency divider with a buffer amplifier in between. The physical dimension of the circuit is $65\mu m \times 85\mu m$. A number of VNCAPs are inserted between DC signals and

Type	Description
$VCODIV_1$	$VCO_1 + DIV_2$
$VCODIV_2$	$VCO_1 + DIV_2$
VCODIV ₃	$VCO_2 + DIV_2$
$VCODIV_4$	$VCO_2 + DIV_3$

Table 2.2: Test vehicles for the PLL front-end.



Figure 2-14: Layout and die photo of the VCO and frequency divider circuit in 65nm SOI CMOS technology.

ground to reduce noise at reference voltages and power supply.

2.5 Summary

In this chapter, we describe the design and analysis of a mm-wave PLL front-end circuit, which contains a VCO and frequency divider as high-speed building blocks of a PLL. A 77GHz LC-VCO is designed to achieve a wide frequency tuning range for improved manufacturability against process variation. The VCO employs complementary topology for phase noise improvement and power reduction, an accumulation-mode varactor for a wide capacitance tuning ratio, relaxed pitch devices for reduced parasitic capacitance, and a thick metal one-turn high Q inductor for improved phase noise performance. A 2:1 frequency divider is designed to operate up to 90GHz input
frequency. The divider employs a ring-oscillator based injection-locking frequency divider scheme for a wide frequency locking range. The divider is implemented using a D-FF architecture with CML latches and inductive shunt peaking for bandwidth enhancement. The self-oscillation frequency and frequency locking range of the divider are analyzed to obtain analytic forms of the divider performance. The bias condition of the CML divider can be tuned externally to change the frequency operating range. The PLL front-end circuit is implemented in a 65nm SOI CMOS technology that provides various features for high-speed analog circuit design including a vertical-natural capacitor, a 10-layer metal stack with an extra-thick top metal layer, and relaxed pitch devices for reduced parasitic capacitance.

Chapter 3

Statistical Measurement and Analysis

In this chapter, we show the measurement results of the VCO and frequency divider designed and presented in Chapter 2. We characterize nominal performance of the VCO including frequency tuning range, power consumption, and output phase noise. The variability of the minimum and maximum frequencies of the VCO over a 300mm wafer is also measured to estimate parametric yield of the VCO. For the frequency divider, we focus on the statistical characterization of the self-oscillation frequency at different bias conditions of V_{DD} and V_{BIAS} . Though the input sensitivity curve of a frequency divider provides detailed characteristics of the divider, the measurement of the input sensitivity curve is an extremely time-consuming task, requiring sweeping of both input power and frequency. The self-oscillation frequency of the divider, on the other hand, is an easy-to-measure performance variable and provides an excellent characterization of process variation in the frequency divider.

To obtain more information about parameter variability from statistical performance measurement data, we develop an extraction scheme in which circuit parameter variation is extracted from the self-oscillation frequency measurement of the divider at multiple bias conditions. The method is based on the sensitivity analysis of performance variables to the deviation of fundamental circuit parameters, using a backpropagation of variation (BPV) approach, which is a modified form of the method shown in [35]. Experimental results presented here show the statistics of each circuit parameter and important correlation between the parameters. In addition, we present a simple yet effective method to analyze process variations using the statistics from manufacturing in-line data drawn from an extensive set of primitive test structures of devices and ring oscillators, without assuming any explicit underlying model for process variation. Based on a variant of principal component analysis (PCA), we are able to reveal systematic variation patterns existing on die-to-die and wafer-to-wafer levels individually.

Section 3.1 depicts the measurement setup and design-for-test features for automated performance measurement of mm-wave analog circuits. Section 3.2 shows the nominal performance of the VCO and frequency divider, verifying their functionalities. Section 3.3 provides the statistical measurement results of the VCO and frequency divider over 300mm wafers. Section 3.4 presents statistical analysis of the measured data of the VCO and frequency divider to identify variation in these key PLL front-end components.

3.1 Statistical Measurement for High-Speed CMOS Characterization

As operating frequencies of CMOS applications have reached the mm-wave regime, it has become more and more difficult for process engineers and circuit designers to characterize the device and circuit performance at such high frequencies. In particular, since process variability of circuit performance becomes a critical issue for parametric yield loss, statistical characterization of device and circuit performance by measuring a large set of circuit samples becomes essential to capture the variation properties of CMOS circuits in advanced technologies.

An on-wafer measurement scheme is necessary to enable measurement of a large number of die samples in the high frequency regime. On-board testing of a large number of circuit samples after packaging is extremely time-consuming, and demands



Figure 3-1: Ground and signal pad assignment of test circuits with physical dimension.

proper de-embedding techniques to exclude the effect of bonding wires and board traces on the circuit performance. For on-wafer measurement of mm-wave circuit test vehicles, pad assignment and interconnect length must be carefully considered during design to reduce overall area, IR drop in power supply wires, transmission-line effect for high frequency signals, and parasitic capacitance to substrate. The test structures in [29] and [27], for example, show highly optimized pad assignment and compact signal routing approaches to improve the measurement accuracy.

Figure 3-1 shows the pad assignment and physical dimensions of the test vehicles of the VCO and frequency divider in this work.¹ In the GSGSGSGSG² pad set, high frequency pads are centered and the test circuit is placed close to the high frequency pads to reduce the length of high frequency interconnects. The high frequency pads utilize only the two top metal layers to reduce parasitic capacitance to substrate, and to provide mechanical strength for repeated contacts by signal probes.

3.1.1 Measurement Setup for the mm-Wave PLL Front-End

Figure 3-2 shows the test setup for the on-wafer measurement of die samples on a 300mm wafer. A W-band measurement system is used to measure performance characteristics up to 110GHz frequency. The position of the probe set is automatically controlled to scan all die samples on each wafer. Circuit conditions for the test vehicles are programmed into a database, and measurement data of performance variables

¹The pad marked by 'S^{*}' indicates a high frequency signal pad.

²G and S indicate ground and signal pads, respectively.



Figure 3-2: Automated high-frequency measurement setup for VCO and frequency divider samples.

(e.g., oscillation frequency, phase noise, average current, etc.) is collected through the general-purpose-interface-bus (GPIB) of a spectrum analyzer and a current meter. We measure 12 sample wafers processed from the same lot, and each wafer contains 76 die samples. Statistical properties of the performance variation in the VCO and frequency divider are characterized over the wafer set.

3.2 Nominal Performance Measurement of PLL Front-End

Nominal performance of the VCO and frequency divider is measured to verify the proper operation of the circuits in desired frequency ranges. For the VCO, center frequency, frequency tuning range, phase noise, and power dissipation are important performance characteristics to be measured. In addition, we define a new figure-of-merit (FOM_T) that includes the frequency tuning range, to enable comparison of the measured performance of this VCO to other state-of-the-art CMOS VCOs operating in similar frequency ranges.



Figure 3-3: Frequency tuning range of the mm-wave VCO over a full range of V_{CTRL} .

For the frequency divider, we measure the full input sensitivity curve to characterize the divider performance. We also measure the self-oscillation frequency of the divider at different bias conditions to check hardware-to-model correlation. Power dissipation is measured to compare the performance of the divider with other stateof-the-art frequency dividers operating in the mm-wave frequency regime.

3.2.1 Measurement of VCO Performance

Figure 3-3 shows the output frequency change over the full tuning range of V_{CTRL} from zero to V_{DD} , for a representative VCO. The VCO is tunable from 66.8GHz to 73.5GHz, and the center frequency is 70.1GHz. As a result, the FTR is 6.68GHz, or 9.55% of the center frequency, which is relatively large compared to the design examples available in the literature for mm-wave CMOS VCOs shown in Table 3.1. The maximum VCO gain (Kv) is 9.09GHz/V at the center frequency, and the minimum Kv is 2.37GHz/V at the minimum frequency.



Figure 3-4: Phase noise characteristic of the VCO and the phase noise measurement set up.

Figure 3-4 shows the output phase noise characteristic of the VCO, and the test setup used for phase noise measurement. Since the VCO phase noise is not directly measurable due to limited output signal power and loss over 50GHz in the measurement equipment, the VCO output is divided by two using a frequency divider, and the divided output phase noise is measured by a spectrum analyzer assuming that the divider adds only a small constant amount of phase noise. The output phase noise was measured when the VCO output frequency is at 63.86GHz and the divided output frequency is 31.93GHz. Measured phase noise of the divided output is -112.14dBc/Hz at 10MHz offset. After 6dB compensation of the phase noise improvement by frequency division, the estimated VCO phase noise is -106.14dBc/Hz at 10MHz offset.

Table 3.1 shows the comparison of the performance of this VCO with other stateof-the-art CMOS VCOs operating at similar frequency ranges. We define FOM_T to compare the VCO performance including the impact of FTR as:

$$FOM_T = PN - 20 \log\left(\frac{f_O}{\Delta f} \cdot \frac{\text{FTR}}{10}\right) + 10 \log\left(\frac{P_{DISS}}{1\text{mW}}\right). \tag{3.1}$$

where PN is a measured phase noise, f_O is a center frequency of the VCO, FTR is

F_O	FTR	Phase	P _{diss}	FOM	FOM_T	Technology
(GHz)	(%)	Noise	(mW)	(dBc/Hz)	(dBc/Hz)	
``´´	. ,	(dBc/Hz)				
51.2	1.39	-85	1.0	-179.19	-162.03	120nm CMOS [99]
56.5	10.27	-108	9.8	-173.13	-173.36	130nm CMOS [22]
64.3	7.00	-85	118.8	-160.41	-157.32	250nm CMOS [68]
89.7	2.68	-106	15.8	-173.08	-161.63	130nm CMOS [21]
98.5	2.54	-102.7	7.0	-174.12	-162.21	130nm CMOS [22]
103.9	1.92	-94	180	-151.78	-137.47	90nm CMOS [39]
105.2	0.19	-97.5	7.2	-169.37	-134.95	130nm CMOS [22]
114.0	2.11	-107.6	8.4	-179.50	-165.96	130nm CMOS [49]
130.9	1.68	-108.4	20	-177.73	-162.24	90nm CMOS [48]
192.1	0.68	-100	16.5	-173.49	-150.10	130nm CMOS [23]
70.2	9.55	-106.14	5.4	-175.76	-175.36	This Work

Table 3.1: Comparison with state-of-the-art CMOS VCOs.

a frequency tuning range in percentage, Δf is the frequency offset for the measured phase noise, and P_{DISS} is total power dissipation. The VCO in this work outperforms other CMOS mm-wave VCOs in FOM_T because of the wide frequency tuning range while achieving reasonably good phase noise and power consumption through use of a complementary topology.

3.2.2 Measurement of Frequency Divider Performance

Figure 3-5(a) shows the sensitivity curves of the frequency divider (with inductive peaking) in the fastest die sample at four different bias conditions of V_{DD} and V_{BIAS} . Because of the bandwidth limitation of the equipment used in this measurement, the divider has been biased to work over the input frequency range from 75GHz to 110GHz. The divider biased at $V_{DD}=2.2$ V and $V_{BIAS}=1.5$ V operates up to 100.2GHz for a 4.87 dBm differential input.³

The sensitivity curves at different bias conditions show that the operating range of this divider can be dynamically adjusted to the tuning range of the VCO by changing V_{DD} and V_{BIAS} . The lowest points of the sensitivity curves indicate the self-oscillation frequency of the divider. The measured power consumption is 52.4

³This is a 1.52X improvement in frequency performance over the fastest reported frequency divider in 90 nm CMOS [87].

mW and the switching energy at the self-oscillation frequency of 46GHz is 0.57 pJ.

Figure 3-5(b) shows the comparison of the simulation and measurement results for the self-oscillation frequency at various V_{DD} and V_{BIAS} conditions. The strong correlation between the simulated and measured self-oscillation frequency data verifies the integrity of the simulation model.

Table 3.2 shows the comparison of the measured performance with other stateof-the-art frequency dividers operating above 50GHz in various technologies. The maximum operating frequency of the CML frequency divider in this work is comparable with the dividers in compound semiconductor technologies such as SiGe and InP HBTs, while the power consumption is much smaller. Table 3.2 shows that the reduction in the switching energy at the self-oscillation frequency is at least 42% compared to other state-of-the-art dividers.

F_{SO}	F_{MAX}	Power	V _{DD}	Switching	Switching	Technology
(GHz)	(GHz)	(mW)	(V)	Energy at	Energy at	
				$F_{SO}~(\mathrm{pJ})$	F_{MAX} (pJ)	
48	66	80	1.8	1.667	1.212	90nm CMOS [87]
95	143.6	90	N/A	0.947	0.627	$400 { m GHz} \; f_T \; { m InP} \; [45]$
77	100	122	3.3V	1.584	1.22	$230 \mathrm{GHz}$ f_T SiGe
						HBT [57]
33	100	750	N/A	22.727	7.5	$135 \text{GHz} \ f_T \ \text{InP} \ [76]$
65	110	1350	-5.2	20.769	12.273	225GHz f_T SiGe:C
						HBT [101]
71	96	770	-5	10.845	8.021	210GHz f_T SiGe
						HBT [93]
92.4	100	52.38	2.2	0.567	0.524	This Work
85.97	91	25.13	1.6	0.292	0.273	This Work

Table 3.2: Comparison with state-of-the-art frequency dividers operating over 50GHz.

3.3 Variation of VCO and Frequency Divider Performance

Statistical properties of the performance variation in the VCO and frequency divider must be characterized to build a realistic variation model and to calculate the para-



(a) Sensitivity curves at different V_{BIAS} and V_{DD} conditions



(b) Hardware-to-model correlation

Figure 3-5: Measured performance characteristics of the frequency divider.

	Average	Std.	. Normalized		Min.
		Dev.	Deviation		
			(%)		
Maximum frequency (GHz)	71.0	0.82	1.15	73.5	69.9
Minimum frequency (GHz)	64.8	0.76	1.18	66.9	63.8
Center frequency (GHz)	67.9	0.79	1.16	70.1	66.9
FTR (GHz)	6.14	0.13	2.11	6.68	5.91
FTR(%)	9.05	0.17	1.90	9.53	8.72
Full range VCO gain (GHz/V)	5.11	0.11	2.11	5.57	4.93
Maximum VCO gain (GHz/V)	7.96	0.43	5.36	9.38	7.25
Minimum VCO gain (GHz/V)	1.72	0.43	25.0	2.37	0.11
$I_{DD}(\mathrm{mW})$ $V_{CTRL}=0$ V	4.48	0.51	11.4	5.15	3.59

Table 3.3: Statistics of VCO performance variation.

metric yield. In this section, we show the statistical measurement data of the VCO and frequency divider performance and their statistics over 300mm wafers. Systematic and random patterns of the performance variation are identified and the causes of the variation patterns are analyzed.

3.3.1 Variation of VCO Performance

The VCO output characteristics are measured at 57 dies in a 300mm wafer, and the statistics of performance variables are summarized in Table 3.3. Figure 3-6(a) shows the histogram of the VCO output frequency. The quantile-quantile normal plot in Figure 3-6(b) shows that the distribution of the VCO output frequency cannot be accurately modeled as Gaussian since it deviates from a normal distribution significantly in the lower tail. Since the output frequency of an LC-VCO is determined by the electrical parameters of an inductor and varactor whose physical dimension is large, the variation of the output frequency is relatively small compared to other performance variables affected by device characteristics. As a result, $3\sigma/\mu$ of the VCO output frequency is only 3.45% while the $3\sigma/\mu$ variation of I_{DD} is 34.2%. An average supply current I_{DD} is 4.48 mA when $V_{DD}=1.2$ V, which results in 5.37 mW of power dissipation in the VCO core. Output buffers attached to the VCO core consume 3.96 mA of supply current at $V_{DD}=1.2$ V.



Figure 3-6: Statistical distribution of the VCO output frequency over a 300mm wafer (57 die samples).

Figure 3-7 shows the minimum and maximum output frequency of the VCO in one 300mm wafer. A common FTR shared by all VCO samples exists from 66.9 to 69.9GHz, which means that when a FTR requirement is within the common FTR, all the designed VCOs can satisfy the FTR specification. When we set the maximum frequency requirement to 70GHz, only three VCO samples fail to meet the maximum frequency requirement to result in 94.7% parametric yield in this wafer.

3.3.2 Variation of Frequency Divider Performance

To estimate the parametric yield of the frequency divider, measuring the maximum operating frequency (F_{MAX}) at a given input power is important. However, the F_{MAX} measurement is costly since it requires input frequency sweep and the detection of proper division. On the other hand, the self-oscillation frequency of the divider is more efficient to measure, and a large portion of the F_{MAX} variation can be explained from analysis of the self-oscillation frequency. For example, when the divider variation is sensitive to the variation in load resistance and capacitance of the divider output nodes, the self-oscillation frequency and F_{MAX} track each other with strong correlation. When the divider is sensitive to the variation in the bias current, the self-oscillation frequency and F_{MAX} vary to opposite directions with each other. The



Figure 3-7: Minimum and maximum frequency of the VCO over a 300mm wafer.

correlation between the self-oscillation frequency and F_{MAX} at different bias conditions of the divider is shown in [55]. Therefore, we conduct statistical measurement of the self-oscillation frequency over multiple wafers, and investigate the systematic and random variation patterns of the self-oscillation frequency.

Figure 3-8 shows the variation of the self-oscillation frequency when $V_{DD}=1.8V$ and $V_{BIAS}=1.5V$ across a 300mm wafer. DIV₁ and DIV₂ are identically sized but DIV₂ has more compact placement and routing for decreased wire capacitance as explained in Section 2.4. DIV₃ has inductive peaking in addition to the layout of DIV₂. For the three different types of dividers, the self-oscillation frequency measurements show $3\sigma/\mu$ of 16.7% and a 22% range between the fastest and slowest dies. The difference between the performance of DIV₁ and DIV₂ is negligible; the self-oscillation frequency of DIV₂ is only 0.39GHz (1.3%) higher than that of DIV₁ on average. Therefore, the reduction of wire capacitance by compact placement and routing is not a major means for overall output capacitance reduction. The self-oscillation frequency of DIV₃ is consistently higher than DIV₂ by 2.57GHz due to the inductive shunt peaking.

In Figure 3-9(a), we show the wafer-level pattern of the self-oscillation frequency in a 300mm wafer. The fastest corner exists on the left-hand side of the wafer when the notch is aligned to the right. The wafer-level pattern shows a mixture of a circular



Figure 3-8: Variation of self-oscillation frequency of three frequency divider alternatives at a bias condition $V_{DD}=1.6$ V and $V_{BIAS}=0.75$ V.

shape trend and horizontal direction dependency. We note that the variation pattern across all wafers is highly systematic. Figure 3-9(b) shows the comparison of the wafer-level variation pattern in four different wafers. The repetition of the similar wafer-level die-to-die variation pattern can be verified visually in the four sampled wafers.

For more accurate comparison, Figure 3-10 shows the within-wafer variation for 11 different wafers at a fixed bias condition (V_{DD} =1.8V and V_{BIAS} =1.5V). The site index is sorted (where the sort is based on f_{SO} average across all 11 wafers, and the sorted die number represents the same location in all wafers) to visualize a common ascending variation pattern in each wafer-level map, which is similarly repeating over the entire wafer set. Each wafer shows a similar pattern with a parallel shift: the cross correlation of the spatial pattern between different wafers is 92% on average. Figure 3-11 shows scatter plots between each pair of wafers to verify the strong cross-correlation in spatial variation patterns across all wafer. Based on this result, with a pre-characterized profile of a reference wafer, we can estimate the divider performance of all dies in a random wafer with 1.5% average error by measuring a single die and checking the difference with the reference wafer profile.



(b) Comparison of multiple wafers

Figure 3-9: Wafer-level die-to-die variation pattern of the self-oscillation frequency of the divider over one 300mm wafer.



Figure 3-10: Variation of self-oscillation frequency of the frequency divider over 11 wafers. Symbol represents the wafer number.

Figure 3-12 depicts the distributions of the measured self-oscillation frequency over each of our 11 wafers. In Figure 3-12(a), the distribution of each wafer is fitted to a Gaussian distribution to visualize the trend of mean and standard deviation values. To check the Gaussian assumption of the wafer-level variation, Figure 3-12(b) shows a quantile-quantile normal plot of the distribution for each wafer and for the entire wafer set. As with the VCO center frequency previously shown in Figure 3-6, we see that the lower tail for divider self-oscillation frequency is non-Gaussian, with somewhat truncated f_{SO} at lower values. The mean value of each wafer spreads from 29.9GHz to 35.8GHz, and $3\sigma/\mu$ of each wafer ranges from 15% to 18%. Considering all 803 dies, $3\sigma/\mu$ total variation (within-wafer and wafer-to-wafer combined) is 23.8%.

3.4 Statistical Analysis of Measurement Data

While it is convenient to measure the self-oscillation frequency of the divider at various bias conditions of V_{DD} and V_{BIAS} , it is difficult to extrapolate the measured statistics of the self-oscillation frequency to similar circuit examples with different sizing and



Figure 3-11: Scatter plots of f_{SO} , for corresponding dividers, between each pair of wafers.

topologies. The statistics of more fundamental circuit parameters (e.g., threshold voltage, channel length, parasitic capacitance, etc.) must be estimated from the measured performance statistics to assess the performance variation of general circuit examples. In this section, we present a back-propagation of variation method which extracts the deviation of fundamental circuit parameters, based on the measured selfoscillation frequencies at multiple bias conditions and on sensitivity analysis from model or simulation.

Alternatively, we also present a simple and practical method to decompose the process variability of complex analog circuits using the statistics of manufacturing inline measurement data, without assuming any underlying model for process variation. The experimental results show that a significant portion of the performance variation in complex RF/mm-wave analog circuits can be explained using the measurement data of primitive inline test structures (e.g., FETs, resistors, capacitors, ring oscillators, etc.).



(b) Quantile-quantile normal plots

Figure 3-12: Statistical distribution of the self-oscillation frequency variation over each wafer and the entire wafer set.

3.4.1 Back-Propagation of Variation

For accurate estimation of circuit performance variation and robust optimization, the variability of critical circuit and process parameters must be precisely characterized and synchronized with simulation tools. Output frequency of a ring oscillator (RO) is a common performance metric to assess the inter-die and long range intra-die variation of delay and power consumption in digital circuits, motivated in part by the convenience of RO measurement [42]. However, since each ring oscillator generates one output frequency whose variation is a composition of various parameter fluctuations, it is difficult to extract the variation of individual circuit parameters for the measured oscillation frequencies.

Sensitivity-based parameter de-embedding is a useful technique to extract the fluctuation of individual parameters from measured output variables. Using parameter sensitivity calculated from device and interconnect models, the deviation of critical circuit parameters from their nominal values can be extracted. To determine the deviation of multiple circuit parameters, the same or larger number of output variables is needed. As an example, the measurement data for ring oscillator frequency f_O and leakage current I_{LEAK} are used to extract the variation of channel length ΔL and oxide thickness T_{ox} in [83].

For the CML frequency divider in this work, the main circuit parameters affecting the self-oscillation frequency are bias current, parasitic capacitance and load resistance. The bias current is a function of the threshold voltage of input transistors. Through simulation, the sensitivity of the self-oscillation frequency with respect to the change of the threshold voltage, parasitic capacitance and load resistance can be calculated at different bias conditions. In combination with these sensitivities, the measurement statistics of the self-oscillation frequency at multiple bias conditions can then be used to extract circuit parameter variations. This method is based on a first-order response surface model of the self-oscillation frequency as follows:

$$\Delta f_{SO} = \alpha \cdot \Delta V_{TH} + \beta \cdot \Delta R_L + \gamma \cdot \Delta C_L \tag{3.2}$$

where α , β , and γ are the sensitivity values of the self-oscillation frequency to each circuit parameter. The deviation of the circuit parameters from their nominal values can be calculated using the parameter sensitivity values (α , β , and γ) extracted from a circuit simulation model and the measurement data of self-oscillation frequency at multiple bias conditions.⁴

We denote α_i , β_i , and γ_i as the sensitivity values at the *i*th bias condition. $\Delta f_{SO,i}$ is the deviation of the measured self-oscillation frequency from its nominal (average) value at the *i*th bias condition. We can build a system of linear equations such that

$$\begin{bmatrix} \Delta f_{SO,1} \\ \Delta f_{SO,2} \\ \vdots \\ \Delta f_{SO,n} \end{bmatrix} = \begin{bmatrix} \alpha_1 & \beta_1 & \gamma_1 \\ \alpha_2 & \beta_2 & \gamma_2 \\ \vdots & \vdots & \vdots \\ \alpha_n & \beta_n & \gamma_n \end{bmatrix} \cdot \begin{bmatrix} \Delta V_{TH} \\ \Delta R_L \\ \Delta C_L \end{bmatrix} = \mathbf{S} \cdot \begin{bmatrix} \Delta V_{TH} \\ \Delta R_L \\ \Delta C_L \end{bmatrix}.$$
(3.3)

The least mean-square error solution of Equation (3.3) can be found by pseudoinversion of the non-square matrix **S** as follows.

$$\begin{bmatrix} \Delta V_{TH} \\ \Delta R_L \\ \Delta C_L \end{bmatrix} = \mathbf{S}^T \cdot (\mathbf{S} \cdot \mathbf{S}^T)^{-1} \cdot \begin{bmatrix} \Delta f_{SO,1} \\ \Delta f_{SO,2} \\ \vdots \\ \Delta f_{SO,n} \end{bmatrix}$$
(3.4)

For the stable inversion of the sensitivity matrix \mathbf{S} , we re-condition \mathbf{S} by multiplying different factors to each column to regularize the mean of each column. The final values of V_{TH} , R_L and C_L are re-scaled by the multiplication factors. Furthermore, the condition number of \mathbf{S} become dramatically improved by increasing the number of bias conditions (the number of rows in \mathbf{S}). In the experiment, we use 12 bias conditions to reduce the singularity of \mathbf{S} to achieve a robust sensitivity matrix inversion.

Additionally, when the amount of parameter variation increases, the first-order model cannot explain the performance deviation accurately in the presence of non-

 $^{^{4}}$ A similar technique directly calculating the standard deviation of the parameters was shown in [35].

Bias Condition (BC)	$\Delta V_{TH} (\mathrm{GHz/mV})$	$\Delta R_L \; (\mathrm{GHz}/\Omega)$	$\Delta C_P ~(\mathrm{GHz/fF})$
BC1	-0.0312	-0.1145	-0.8565
BC2	-0.0401	-0.1159	-0.8648
BC3	-0.0458	-0.1165	-0.8918
BC4	-0.0291	-0.1252	-0.8872
BC5	-0.0445	-0.1151	-1.0340
BC6	-0.0544	-0.1162	-1.0902
BC7	-0.0272	-0.1369	-0.9047
BC8	-0.0440	-0.1292	-1.0348
BC9	-0.0322	-0.1345	-1.0764

Table 3.4: Sensitivity of the self-oscillation frequency to circuit parameters at different bias conditions.

linear circuit performance. A quadratic expansion of the response surface model can provide higher accuracy over local performance space as shown in [66]. Equation (3.3) can be expanded to a quadratic form:

$$\Delta f_{SO,i} = \mathbf{x}^T \cdot \mathbf{A}_i \cdot \mathbf{x} + \mathbf{b}_i \cdot \mathbf{x} + c_i \tag{3.5}$$

where \mathbf{A}_i , \mathbf{b}_i , and c_i are from quadratic response model fitting at each bias condition, and $\mathbf{x} = [V_{TH} \ R_L \ C_L]$. A non-linear equation solver such as fmincon in MATLAB can find the least mean-square solution of \mathbf{x} for a set of Equation (3.5) for $i = 1, \ldots, n$.

Experimental Results of Parameter Extraction

The parameter extraction scheme for threshold voltage, load resistance, and lumped parasitic capacitance are tested using the divider self-oscillation frequency measurement results. The sensitivity matrix is calculated from SPICE simulations with device, passive, and interconnect models. The model-to-hardware correlation is verified Section 3.2.2, as shown in Figure 3-5(b), to demonstrate reasonable accuracy of the SPICE simulation results. The self-oscillation frequency is measured at all combinations of 21 V_{BIAS} steps and four V_{DD} steps. We select nine bias conditions that provide the most non-singular matrix **S** for the calculation in Equation (3.4). The sensitivity matrix of the nine bias condition is shown in Table 3.4.

Figure 3-13 summarizes the statistics of extracted parameter fluctuations. The



Figure 3-13: Extracted variation of V_{TH} , load resistance, and capacitance and their correlations

extracted $3\sigma/\mu$ of the threshold voltage, parasitic capacitance and resistance variation is 21%, 22% and 20%, respectively. For accurate Monte Carlo analysis of circuits, knowing the correlation between circuit parameters is essential. The scatter plot in Figure 3-13 shows a significant negative correlation ($\rho = -0.85$) between parasitic capacitance and resistance. The threshold voltage does not show notable correlation with other parameters. This correlation structure implies that the variation in physical dimension of poly-silicon load resistors is the major portion of the variation in total load resistance and output capacitance since the variation in the width of the poly-resistors causes the variation of resistance and parasitic capacitance to opposite directions.

Table 3.4 shows the sensitivity of each circuit parameter to the self-oscillation frequency at tested bias conditions. From the sensitivity analysis and the results from BPV analysis, 3σ deviation of V_{TH} , R_L and C_P from their nominal values causes the change of the self-oscillation frequency by 2.23, 5.30, and 4.55 GHz, respectively, at a normal bias condition (BC5). As V_{BIAS} increases, the self-oscillation frequency becomes less sensitive to V_{TH} , but the sensitivity to R_L , and C_P increases.

3.4.2 Decomposition of Frequency Divider Variation

For the characterization of active and passive devices in a manufacturing process, in-line electrical measurement of primitive test structures is typically performed using parametric testers. To keep track of the performance and DC characteristics of devices or other circuit elements, assorted test structures and conditions have been employed [53]. For example, FET devices of different sizes and layouts are designed and fabricated to monitor critical electrical parameters such as threshold voltage, drive current, and leakage current. Useful implications from the large collection of this heterogeneous in-line measurement data can be extracted using various statistical methods [28].

We propose a statistical method to analyze manufacturing in-line data to separate die-to-die variation and wafer-to-wafer variation from overall circuit performance variation. Using the proposed method and a set of pre-production manufacturing inline data collected from test structures built with a 65nm SOI CMOS technology, we evaluate the relative amount of systematic die-to-die and wafer-to-wafer variations in total in-line measurement data. Along with sensitivity analysis of circuit performance to the variation parameters, the contributions of systematic die-to-die and wafer-to-wafer variation can be evaluated separately. This method enables us to assess the effect of random variation, which cannot be explained by systematic variation components.

Constrained Principal Component Analysis

Principal Component Analysis (PCA) is a linear transformation of a set of random vectors to a new set of vectors, principal components (PCs) [50]. PCs are orthonormal and are ordered so that the first few PCs retain a large portion of the total variance in the set of original variables. The first PC can be visualized as the direction on which the variance of the projection of the original vector is maximized as expressed in Equation (3.6). Here, x is the original data vector, and w_i is the i^{th} PC. Subsequent PCs are defined in the same way except they need to be orthogonal to the previous

PCs.

$$w_{1} = \underset{||w||=1}{\operatorname{argmax}} \operatorname{var}(w^{T}x)$$

$$w_{k} = \underset{||w||=1, w \perp w_{k} \forall i=1, \dots, k-1}{\operatorname{argmax}} \operatorname{var}(w^{T}x), \ k \ge 2$$
(3.6)

In practice, PCA is often implemented with singular value decomposition of the covariance matrix of a given data set. PCA is a useful multivariate tool to reduce the dimension of the data set, to reduce noise, or to visualize the representative features of the given multidimensional data.

Here we discuss an enhanced approach appropriate to our need for understanding die-to-die and wafer-to-wafer variation. A constrained principal component analysis (CPCA) extracts constrained principal components (CPCs) which have the same properties as original PCs, but are constrained to a predefined subspace. Our goal is to extract the PCs of die-to-die or wafer-to-wafer variations separately for better understanding of the variation. In CPCA, CPCs can vary only in a guided set of dimensions which are consistent with die-to-die or wafer-to-wafer variation. Figure 3-14 exhibits the difference between CPCA and traditional PCA. Conceptually, PCA finds orthogonal coordinates which do not generally coincide with die and wafer variations. Therefore, the understanding of process variation using ordinary PCs would be perceptually difficult. On the other hand, CPCA guides the PCs to the die and wafer directions, leading to direct visualization of the variation dependencies in die-to-die and wafer-to-wafer. Only a few CPCs are examined for this purpose, because only a fraction of CPCs are sufficient to capture most of the information, as is also the case with PCs.

For our experiments using CPCA, 1109 in-line parameters from pre-production 65nm SOI CMOS technology wafers are used. A data set for each in-line parameter contains 520 samples (40 dies per wafer for 13 300mm wafers from the same lot). Various measurements from FET test structures (e.g. V_{TH} , I_{ON} , I_{OFF}), ring oscillators (ROs), SRAMs and capacitors are also obtained, as listed in Table 3.5.

Both ordinary PCA and CPCA were performed on an example data set for com-



Figure 3-14: Comparison of traditional PCA (a) vs CPCA (b).

Type	FET	RO	SRAM	Cap	Total
# of parameters	759	83	159	108	1109

Table 3.5: Categories of in-line parameters used in the CPCA analysis

parison. Figure 3-15 shows the cumulative variance which can be explained by the first 20 PCs and CPCs for PCA and CPCA, respectively. The first PC and CPC account for 31-34% of the total variance of the original data set. Using the first two CPCs, 57% can be explained, slightly less than the 61% for the ordinary PCs.

Experimental Results of CPCA on Frequency Divider Data

The CPC decomposition obtained from in-line data can be applied to the measured self-oscillation frequency (F_{SO}) of the CML frequency divider, to help understand spatial dependencies of F_{SO} . F_{SO} was measured from the same dies and wafers on which the in-line parameters used for the CPCA experiment were obtained. Figure 3-16 illustrates a sequence of projections of F_{SO} data onto successive CPCs in three dimensions, to visualize how F_{SO} can be reconstructed by adding one component at a time using an offset and the first four CPCs.

The bottom surface (z-axis) shows the measured F_{SO} from measured dies and wafers. The global offset shown at the top is an average of F_{SO} over all dies and wafers, thus a constant. The second surface from the top is the first CPC plus the offset, having only die-to-die (site dependent) variation. The next surface displays



Figure 3-15: Comparison of cumulative variance explained by PCA and CPCA, for in-line DC test data.

the added contribution of the second CPC (wafer-to-wafer variation) on top of the previous surface. The sequence of the surfaces demonstrates how the original data can be successively reconstructed from or, equivalently, decomposed into a few CPCs. Note that these CPCs are calculated from the in-line DC test data and not from the F_{SO} data which is being analyzed. A weight for each CPC is obtained by projecting F_{SO} data onto each CPC space. The first four CPCs retain 66% of all the information of F_{SO} variation, which is significant because the test data (frequency of an RF circuit) and the training data for CPC calculation (in-line DC measurement data) are quite different in nature. The physical mechanism of how each in-line device-level parameter affects complex RF circuitry such as the frequency divider is elusive and challenging to analyze. However, the proposed algorithm and experimental data show that significant portion of the performance variation is systematic, and therefore, the CPCs obtained from in-line measurement can explain a substantial portion of the process variation in complex RF circuits.



Figure 3-16: Decomposition of frequency divider self-oscillation frequency variation based on constrained principal component analysis.

3.5 Summary

In this chapter, we showed the statistical measurement results of the mm-wave VCO and frequency divider in a 65nm SOI CMOS technology. The implemented test vehicles have additional features for automated mm-wave performance measurement. The measurement data from 12 wafers and 76 dies from each wafer reveals trends in the variation of the VCO and frequency divider. The statistics of VCO measurement data show an improved frequency tuning range and parametric yield through careful device and circuit design. The variability of the divider performance is larger than that of the VCO, since the divider performance is determined by minimum sized devices which have more relative variation. A strong systematic wafer-level pattern of the variability in the self-oscillation frequency of the divider is observed.

We provide an extraction scheme for circuit parameter variation estimation from the measurement data of the self-oscillation frequency of the divider at multiple bias conditions. The extraction method is based on the sensitivity analysis of circuit performance to the deviation of circuit parameters with a back-propagation of variation approach. Experimental results demonstrate that the statistics of each circuit parameter and the important correlation between the parameters can be successfully extracted. Additionally, process variation in electrical parameters of circuit components can be decomposed into die-to-die and wafer-to-wafer variation using a constrained principal component analysis. Constrained principal components calculated from primitive test structures can explain a significant amount of variance in the selfoscillation frequency variation of the divider, suggesting that the variation of complex circuit performance in a new technology can be predicted by the measurement data of a primitive test structure set.

Chapter 4

Robust Optimization of Analog Circuits

While analog circuits account for only a small fraction of the total transistor count in mixed signal ICs, they consume a considerable amount of the area and power of the entire system. Furthermore, analog design demands intense design effort, since analog design procedures are still strongly dependent on manual parameter tuning. In many cases, analog building blocks are responsible for costly manufacturing respins in advanced technologies due to faults and parametric yield loss. As a result, analog circuit design becomes a bottleneck for the design, verification, and migration of mixed signal ICs.

In contrast to digital design, analog circuit designers must consider a complex multi-dimensional performance space to optimize circuit performance [91]. In digital design, functionality, timing, and power consumption of logic gates are the primary concerns, and in many cases, the problem reduces to a transistor size optimization problem to meet given constraints of required operating frequency and power consumption. However, in analog design, there are often a large number of performance variables to be considered, and the definition of the performance space is highly dependent on the type of the analog building block. Typical output performance variables include gain, delay, noise, voltage swing, impedance, and power, as noted in Figure 4-1. Conceptually, the robust optimization of analog circuits consists of adding a parametric yield consideration on top of the typical performance constraints.



Figure 4-1: Robust optimization of analog circuits considering process variability and parametric yield constraints.

Traditionally, analog circuit design has been intensively dependent on the experience of circuit designers. However, as the operating frequency and design complexity of analog blocks, increases due to the scaling impact, computer-aided analog design and optimization tools are desperately needed to automatically size the devices and to bias circuits for optimal performance and maximum yield. In particular, as the cost of re-spins in advanced technologies is extremely high, accurate estimation of parametric yield becomes essential in the development of analog design tools.

Unlike digital design, it is difficult to build a generic optimizer that can be used across the wide variety of analog building blocks since the behavioral characteristics of each analog circuit are extremely different from each other. Thus, instead of replacing analog designers, the optimization tool must help the designers by providing fast and accurate evaluation of circuit performance, and facilitating computationally efficient exploration over the multi-dimensional performance space. Furthermore, for the sake of high yield, there must be an efficient method to estimate parametric yield of analog circuits within a reasonable amount of time and computational resource.

In this chapter, we present a robust optimization methodology for analog circuit design, applied to our case study of VCO and frequency divider design. Section 4.1 introduces the background about static and dynamic optimization of parametric yield in analog circuit design. Section 4.2 analyzes the performance variability of the VCO and frequency divider, and identifies the primary causes of functional failure in the PLL front-end circuit due to variability. The feasibility of dynamic performance optimization in the PLL front-end is verified by showing how the parametric yield changes through tuning of the bias condition of the frequency divider. The underlying tradeoff in the dynamic tuning approach is also investigated. In Section 4.3, we describe a static yield optimization scheme of the PLL front-end, enabled by a numerically efficient performance evaluation model of the frequency divider. The parametric yield of the PLL front-end circuit can be optimized at the design stage by exploring over the entire design space within a reasonable amount of simulation time using the functional approximation.

4.1 Background: Yield Optimization of Analog Integrated Circuits

Various computer-aided design yield optimization methodologies for analog circuits have been suggested over the past several years [43] [63] [94]. Traditional yield improvement methodologies for analog circuits are based on *design centering*, which maximizes the margin of circuit parameters from acceptable boundaries to accommodate large-scale parameter variability. The design centering methodologies can be categorized based on the migrating parameter space: design parameter space centering and performance space oriented centering, as shown in Figure 4-2 [109]. Figure 4-2(a) shows the design parameter space centering method, which maximizes the minimum distance between a design point and the boundaries of an acceptable design space set by given performance constraints. A gradient-based deterministic optimization algorithm can be used to find the design point with maximum distance from the boundary. The computational cost of the search algorithm increases linearly with the number of design variables [5]. On the other hand, in the performance space oriented centering (Figure 4-2(b)), generalized distances between the lower and



(a) Design centering in design parameter space. (b) Design centering in performance variable space.

Figure 4-2: Design centering methodologies for indirect yield improvement.

upper specifications of output performance variables and simulation results are used to construct a suitable objective function to be optimized [66]. The performance centering methodology eliminates the costly search or generation of the acceptable boundary of the design parameter space, and provides computationally efficient operations. However, the relationship between design parameters and output performance variables are likely to be non-linear and discontinuous in analog circuits. Thus, without knowing the actual statistical distribution of the output performance variables, the centering in the performance space cannot achieve true yield maximization [109].

As the CMOS feature size scaling goes below 100nm, target performance of analog circuits becomes extremely high, and allowable performance margin of the high performance analog design has been reduced significantly. Thereby, the traditional statistical design methodologies based on design centering cannot provide properly optimized designs in advanced technologies. Instead, the effort of robust optimization in nano-scale technologies must better focus on circuit-level techniques to achieve large performance margin by using dynamic tunability and computationally efficient techniques for estimating parametric yield of complex analog circuits. We classify the robust optimization approaches of analog design as dynamic and static optimization methodologies. The following sections introduce the previous work related to the dynamic and static optimization of analog circuit yield.

4.1.1 Dynamic Optimization of Analog Circuit Yield

As industry standards for system performance become higher, analog blocks must meet more stringent performance metrics for bandwidth, gain, linearity, jitter, power, and chip area. To achieve the performance targets despite increasing variability, the degradation of performance in each building block or the operating range mismatch between these building blocks must be compensated with correction circuits that provide in-situ measurement and dynamic calibration of the circuit performance [10]. Finding an appropriate performance knob to provide a sufficient amount of tunability with allowable impact on other performance variables is a prerequisite for a dynamic compensation methodology. Design cost of adding the in-situ measurement and feedback circuitry to control the performance knob is also a critical concern.

In digital circuit design, adaptive control of power supply and body bias voltages (V_{SB}) can reduce the variation in delay and leakage power consumption of general logic gates significantly, as shown by Tschanz et al. in [103] and by Chen et al. in [26]. A similar technique can be adapted to analog circuits; for example, in differential CML comparators and summers, a DC offset arising from within-die variation can be corrected by measuring a tail current from each leg of the outputs and applying an offset current from a current digital-to-analog converter (IDAC) block to achieve constant and calibrated output values. A range of DC offset in voltage comparators can be reduced by a factor of a dozen using a low-resolution IDAC [52]. The area overhead of the IDAC circuitry is roughly proportional to the maximum range of the offset cancellation, assuming a fixed resolution. The additional chip area for the DC offset cancellation will be directly proportional to the variability, assuming that the IDAC is not affected by variability.

An adaptive post silicon tuning methodology to effectively reduce random device mismatches for analog circuits was proposed by Li et al. in [65]. A traditional approach to reduce random mismatch between supposedly identical components is simply increasing the physical size of the components. Following Pelgrom's model [85], the standard deviation of the mismatch is inversely proportional to the square root of the component area. However, increasing the size of the devices is counter to the scaling trend, and other performance variables such as speed and noise can be degraded. Instead, a pair of components can be decomposed into N sub-components and a sub-set of the sub-components that results in the best matching characteristic can be adaptively selected by a dynamic programming algorithm. The experimental results in [65] demonstrate that the standard deviation of device mismatch exponentially decreases as area increases using such a post-silicon tuning methodology.

4.1.2 Static Optimization of Analog Circuit Yield

As the variability of critical circuit parameters and design complexity of analog circuit increase, indirect yield optimization techniques such as design centering in the performance space can no longer provide true optimization of the parametric yield of analog circuits. For accurate estimation of parametric yield, statistical distributions of performance variables must be estimated at each design point, and the parametric yield must be calculated based on the estimated performance statistics. By employing the estimated yield as an objection function, direct yield optimization techniques, such as a gradient based search algorithm, can find the optimal design point achieving the highest yield. We refer to this procedure as a static or design-time optimization of analog circuits, as opposed to a dynamic or post-silicon optimization in the previous section.

For static optimization, the trade-off between the accuracy and computational cost of yield estimation is a critical concern. The accuracy of the yield estimation can be obtained simply by a large number of Monte Carlo iterations with full SPICE simulation, consuming a large amount of time and computational resources. To reduce the computational cost in performance evaluation, a response surface model (RSM) is a commonly used technique, as shown in Section 1.3.1. When the variation of circuit parameters is sufficiently small, a linear RSM can capture a significant portion of the performance variation. The statistics of output variables modeled by the linear RSM can then easily be calculated when the parameter variation is assumed to be Gaussian.
However, as the circuit parameter variability increases, the linear RSM can no longer explain the performance variability accurately. As an improvement, non-linear RSMs (e.g., quadratic polynomials) can be utilized to capture large-scale process variability. However, such models result in non-normal distributions for circuit performance, which are difficult to capture efficiently since the distribution model is unknown. Li et al. presented a computationally efficient estimation method to characterize the distribution of output variables modeled by non-linear RSMs in [64].

Estimating parametric yield at a large number of design points to find the maximum yielding point is a simple and practical approach for robust design. The calculation of the parametric yield can be accelerated by the RSM-based yield estimation shown above. Since the computational cost to estimate the yield over the entire design space is still expensive, a gradient-based search algorithm can be employed to avoid full exploration of the design space [109]. Alternatively, equation-based yield optimization described in Section 1.3.2 can be used for computational efficiency, based on the assumption that performance variables can be well-approximated by special functions (e.g., posynomial functions) [63] [108].

The static and dynamic optimization methodologies can be combined to maximize the overall yield improvement. For digital circuit design, a unified optimization methodology involving design-time gate-level sizing and post-silicon die-level body bias tuning and their impact of parametric yield were presented by Mani et al. in [69]. In the following sections, the yield benefit in the VCO and frequency divider circuits achievable through dynamic bias voltage tuning and static design parameter optimization is estimated, and an experimental result is shown in Section 4.3.3.

4.2 Dynamic Yield Optimization of PLL Front-End

To compensate for the variation of the PLL front-end circuit dynamically, the primary cause of parametric yield loss must be identified first. Figure 4.2 shows the compar-

ison between the variation of the VCO tuning range and the divider self-oscillation frequency over a 300mm wafer. The wafer-level die-to-die variation pattern is highly systematic for both the VCO and frequency divider; the cross correlation between different wafers is more than 90% on average, as discussed in Chapter 3. From the measurement of 65 dies (sites), $3\sigma/\mu$ of the VCO center frequency variation is 3.52%, and the variation of the self-oscillation frequency is 17%. The divider variation is around five times larger than the VCO variation, since the divider performance is primarily determined by the electrical parameters of the transistors, while the performance of the LC-VCO is mainly determined by passive components such as an inductor. The physical size of the inductor is much larger than that of the minimum channel length devices; the parameter variability of the passive components is much larger than that of the active devices.



Figure 4-3: Comparison of the variation of VCO minimum frequencies and frequency divider self-oscillation frequencies over 65 dies on one 300mm wafer.

Despite the high wafer-to-wafer correlations of the individual circuits is high, the spatial correlation between the two variations, as seen from the scattering diagram in Figure 4.2, is not significant ($\rho=0.17$). Thus the VCO and frequency divider operating ranges do not track each other in the presence of significant inter-die variation. This result suggests an important concern in the PLL front-end design: the divider operating range must cover the VCO tuning ranges in both of the worst case combinations (e.g., fast VCO-slow divider and slow VCO-fast divider) to guarantee high manufacturing yield.



Figure 4-4: Correlation between the variation of the VCO and frequency divider over 65 dies on one 300mm wafer.

From the comparison of the variation in the VCO and frequency divider, it becomes clear that variation in the divider is the dominant cause of functional failure in the operation of the VCO-divider combination. The frequency operating range of the divider must fully cover the tuning range of the VCO in order to divide the VCO output frequency properly. However, in the mm-wave band, it is difficult to achieve more than 15% for the frequency operating range of the frequency divider, for a 0 dBm reference input signal. From the measurement data, $17\% 3\sigma/\mu$ variation in the self-oscillation frequency of the divider was observed, and this can cause a serious yield loss in the PLL front-end circuit.

In this section, we analyze the failure mechanism of the PLL front-end circuit caused by the process variation in the divider. In Section 2.3.2, simulation results showed that the frequency operating range of the CML frequency divider can be tuned by the control of a bias current. We verify that this tunability can be utilized to compensate for the variation of the divider and improve the parametric yield of the PLL front-end circuit by measurement results. We discuss a performance trade-off between the parametric yield and phase noise performance of the divider output.

4.2.1 Failure Mechanism of PLL Front-End Circuit

The change of threshold voltage in the tail transistors (M_9, \ldots, M_{11}) in Figure 2-6) is nearly equivalent to the V_{BIAS} changes of those transistors. For a fixed V_{BIAS} , a lower threshold voltage gives higher gate overdriving voltage and results in a larger bias current in the CML buffers. The larger bias current increases the self-oscillation frequency and reduces the frequency operating range in the frequency divider. To analyze the process variation impact on the performance of the VCO and frequency divider, Figure 4-5 illustrates how the change of V_{BIAS} affects the operation of the VCO-divider combination.

In Figure 4-5, at $V_{BIAS,1}$, the input sensitivity curve of the frequency divider is well below the VCO output power level over the VCO tuning range; the frequency divider operates in the full VCO tuning range properly. As V_{BIAS} increases to $V_{BIAS,2}$, the self-oscillation frequency increases to shift the sensitivity curve toward the VCO tuning range, but the VCO output power becomes partly below the input sensitivity curve in the VCO tuning range since the sensitivity curves becomes steeper. That is to say, the VCO no longer provides enough output power in part of the tuning range to drive the divider at the desired output frequency, and the divided frequency falls between the self-oscillation and desired divided frequencies in the failure region. V_{BIAS} can be increased even more to set the self-oscillation frequency at the center of the VCO tuning range since the input sensitivity curve becomes extremely steepened. A small amount of variation in this condition can shift the sensitivity curve out of the VCO tuning range and cause a serious mismatch problem between the operating ranges of the VCO and frequency divider.



Figure 4-5: Failure analysis of the PLL front-end circuit in the presence of significant divider variation.

On the other hand, higher V_{BIAS} results in larger output swing and provides larger gate over-drive of input devices to stabilize the divider performance against threshold voltage variation. Therefore, to minimize the power consumption and functional failures in a fixed tuning range while obtaining reasonably large output swing for better phase noise characteristics, the optimal V_{BIAS} condition should be determined in the PLL front-end integration. Figure 4-6 shows the divider output frequency as V_{CTRL} changes from 0 to 1.2V for one of our 65 tested dies. When V_{BIAS} moves from 0.5 to 0.7V, the divider correctly generates the divided frequency. At $V_{BIAS}=0.8V$, the divided frequency is down-shifted slightly since the sensitivity curve becomes close the VCO output power and the self-oscillation frequency starts to be mixed into the divider output (injection-pulling). At $V_{BIAS}=0.9V$, the divider output departs from the desired output in the middle V_{CTRL} range and instead converges to the divider self-oscillation frequency, since the VCO output power is smaller than the minimum required input power of the divider to be locked at the desired frequency. When $V_{BIAS}=1.0$ or 1.1V, the input power is well below the sensitivity curves, and the divider generates constant self-oscillation frequencies rather than divided VCO output frequencies.



Figure 4-6: Divided output frequency of the PLL front-end circuit at different bias conditions of the frequency divider.

Figure 4-7 shows the output frequency of the PLL front-end circuit over the tuning range of the VCO at critical V_{BIAS} conditions. At $V_{BIAS}=0.8$ V, all 65 dies from a 300mm wafer functions properly in dividing the VCO output over the full tuning range. However, at $V_{BIAS}=0.9$ V, more than 50% of the dies fail to divide the VCO output over the full tuning range. Therefore, the optimal V_{BIAS} condition to achieve both maximum functional yield and the lowest phase noise characteristic must be between 0.8 and 0.9V.

4.2.2 Optimal Bias Condition for Maximum Yield

We define a sound PLL front-end die sample as one in which the divider operating range completely covers the tuning range of the VCO. Figure 4-8 shows the yield



Figure 4-7: VCO-divider outputs from 76 die samples over a wafer.

Type	Description
$VCODIV_1$	$VCO_1 + DIV_2$
$VCODIV_2$	$VCO_1 + DIV_2$
VCODIV ₃	$VCO_2 + DIV_2$
$VCODIV_4$	$VCO_2 + DIV_3$

Table 4.1: Test vehicles for the PLL front-end.

of the four different test vehicles of the PLL front-end when V_{BIAS} changes from 0.4 to 1.1V. The configurations of the four PLL front-end test vehicles are described in Table 4.1. The yield of VCODIV₁ and VCODIV₂ is less than VCODIV₃ and VCODIV₄ since the center frequency of VCO₁ is higher than that of VCO₂. Comparing VCODIV₃ and VCODIV₄, the addition of inductive peaking has some benefit when V_{BIAS} is at its highest. The layout optimization does not show notable benefit between VCODIV₁ and VCODIV₂. When V_{BIAS} is from 0.5 to 0.8V, the yield of VCODIV₃ and VCODIV₄ is 100%. Considering output phase noise, V_{BIAS} of 0.8V is the optimal yielding point for VCODIV₃ and VCODIV₄. For VCODIV₁ and VCODIV₂, V_{BIAS} of 0.7V gives the maximum yield.



Figure 4-8: Functional yield of the PLL front-end circuit at different bias conditions of the frequency divider over four VCO and divider design alternatives.

4.3 Static Yield Optimization of PLL Front-End

While the dynamic bias tuning of the frequency divider helps to improve the parametric yield of the PLL front-end circuit, the circuit parameters must initially be optimized at design time to maximize the manufacturing yield for given specifications. Several indirect and direct static yield optimization methodologies were introduced in Section 4.1.2. The static optimization methods are based on the estimation of the circuit parametric yield using the statistics of circuit parameters. Monte Carlo is a most commonly used technique to assess the parametric yield of a given circuit. Exploration of the design space to find the maximum yielding point using the Monte Carlo yield estimation is a simple and well-defined task. However, Monte Carlo includes a large number of iterations and becomes extremely time-consuming when the circuit is large and high accuracy is needed. Avoiding the simulation cost of Monte Carlo simulations while preserving the estimation accuracy has been a critical issue in CAD research.

In the PLL front-end, the frequency divider is the main concern in variabilityinduced functional failure, based on the variation statistics in Section 4.2.1. The distribution of the divider maximum operating frequency must be estimated accurately to calculate the parametric yield of the PLL front-end. However, the maximum operating frequency cannot be evaluated by a single transient simulation. Instead, it requires a sequence of transient analyses to sweep the input frequency of the divider with a small incremental step, in order to find the boundary of the divider operating range accurately.

Figure 4-9 shows the input sensitivity curve and the evaluation to find the maximum operating frequency at a given reference input power. To achieve high accuracy evaluation of the maximum operating frequency, a sequence of transient simulations is needed with a fine input frequency step to find the frequency at which the divider starts to fail the proper frequency division.

A common binary search can be helpful to reduce the number of transient simulations required for the maximum operating frequency evaluation. Figure 4-10 explains



Figure 4-9: Evaluation of maximum operating frequency of the frequency divider. Sequence of transient simulation with a fine frequency step is needed to evaluate the maximum operating frequency with high accuracy.

the binary search scheme. The frequency divider is simulated with a coarse input frequency step to set an initial frequency interval which includes the maximum operating frequency. The size of the frequency interval is reduced by half at each transient simulation to check the proper frequency division at the mid-point of the interval. The accuracy of the maximum operating frequency evaluation increases in logarithmically as the number of transient analyses increases.

Even though the circuit size of the frequency divider is small, each evaluation of the maximum operating frequency demands multiple transient simulations, and the total computational cost for estimating the parametric yield over the entire design space using Monte Carlo analysis will be extremely large. To bypass the full transient circuit simulation, we need a numerically efficient functional approximation of the maximum operating frequency. In the following, we suggest a new frequency model for a CML buffer-based ring oscillator, which is fitted by only DC circuit parameters. We extend the frequency model to explain the maximum operating frequency of the divider by adding a term based on an understanding of the locking range of the



Figure 4-10: Binary search scheme of the maximum operating frequency of the divider.

ring-ILFD discussed in Section 2.3.2.

4.3.1 Estimation of Frequency Divider Performance

The maximum operating frequency of the frequency divider, f_{MAX} , can be represented as follows.

$$f_{MAX} = f_{SO} + \Delta f, \tag{4.1}$$

where f_{SO} is the self-oscillation frequency and Δf is the frequency locking range of the divider.

In the self-oscillation mode, the divider is operating as a four-stage CML bufferbased ring oscillator (Section 2.3.2). Therefore, we first build a model for the oscillation frequency of a CML buffer-based ring oscillator.

Figure 4-11 shows a simple CML delay buffer. The delay buffer consists of a differential pair of CMOS devices (M_1 and M_2), resistor loads (R_1 and R_2), a tail current source (I_{TAIL}), and lumped output capacitance (C_1 and C_2). In a CML buffer, a full voltage swing at the output node is simply

$$V_{SW} = I_{TAIL} \cdot R_L. \tag{4.2}$$



Figure 4-11: A CML differential delay buffer.

Historically, a complete analytic model of the propagation delay even in a simple CMOS inverter has been extremely hard to obtain. The model in [12] provided a surprisingly complex analytic propagation delay model in a CMOS inverter, assuming a piecewise linear input signal. However, the complexity of the model prevented the practical use of the propagation delay model in conventional circuit design. It becomes even more complicated if the buffer is used as a stage of a ring oscillator, since the assumption that the shapes of input and output signals are identical except for inversion and time delay must be added in the calculation.

Typically, the propagation delay of a CML buffer is modeled by assuming the exponential decay of the output signal as follows [3].

$$t_d = \frac{C_L \cdot V_{SW} \cdot ln2}{I_{TAIL}} = \frac{I_{TAIL}R_L \cdot C_L \cdot ln2}{I_{TAIL}} = R_L C_L \cdot ln2$$
(4.3)

However, the major drawback of the model in Equation (4.3) is that it cannot explain the dependency of time delay on the tail current I_{TAIL} . The propagation delay of a CML buffer must be a function of I_{TAIL} since this current changes the transconductance of differential pair devices to give different strength of output load capacitor discharging. Appendix A shows the change of the output frequency of a ring-based



Figure 4-12: Transconductance function g(V) of the differential pair in Equation (4.4).

VCO to verify the change of the propagation delay through I_{TAIL} control.

To represent the propagation delay of the CML buffer as a stage of a ring oscillator, we first start from the node equations of the CML buffer. A key assumption in the equation building is that the input signal $V_I(t)$ and the output signal $V_O(t)$ are identical except for inversion and propagation delay t_d as follows:

$$V_O(t) = -V_I(t+t_d).$$

Using the assumption above, a final differential equation as a function of $V_O(t)$ can be developed as follows.

$$\frac{dV_O(t)}{dt} = -\frac{1}{R_L C_L} V_O(t) + \frac{g(V_O(t+t_d))}{C_L},$$
(4.4)

where g(V) is the transconductance of the differential pair as described in Figure 4-12. From observation, Equation (4.4) is non-causal and non-linear, and therefore, it is extremely hard or impossible to obtain an a closed-form solution of Equation (4.4).

Instead, we can build an RSM of the propagation delay t_d based on the circuit parameters of the CML buffer. We reviewed in Section 1.3.1 that the accuracy of an RSM can be improved by either increasing the complexity of the fitting model or introducing physical understanding of non-linear or discontinuous circuit behaviors. In this work, we do both to improve the modeling accuracy. Figure 4-13 shows the transient simulation result of the input and output signals in the CML buffer, assuming that the number of stages is sufficiently large for the signals to saturated to their minimum and maximum values. Since this is a differential operation, charging and discharging of output capacitance happen simultaneously. In an early part of the transition, discharging operation of a load capacitor dominates the transition, so that the signal slope can be approximated by g_m/C_L . In a later part of the transition, charging up operation from V_{DD} through the output resistor R_L becomes stronger, and the transition time is mainly determined by an R_LC_L time constant. Following this analysis, we define two time delay components as below.

$$t_1 = C_L/g_m \tag{4.5}$$
$$t_2 = R_L \cdot C_L$$





The RSM of t_d can be developed based on the two time delay components t_1 and t_2 in Equation (4.5). We use a quadratic RSM since it provides stable model coefficient calculation and captures second-order non-linearity of output variables [110]. The quadratic RSM can be formulated as follows.

$$\mathbf{x} = (t_1, t_2, \dots, t_k)^T,$$

$$t_d(\mathbf{x}) = \mathbf{x}^T \cdot \mathbf{A} \cdot \mathbf{x} + \mathbf{B} \cdot \mathbf{x} + c,$$

$$f_{SO} = \frac{1}{N \cdot t_d(\mathbf{x})},$$

(4.6)

where **A** and **B** are quadratic and linear terms of the RSM, respectively, and c is a constant term. $t_d(\mathbf{x})$ is the propagation delay of each CML buffer stage, and f_{SO} is the oscillation frequency of a CML ring oscillator with N stages. The coefficients of **A**, **B**,and c can be calculated from the evaluated f_{SO} at multiple design points. The sample design points must cover a sufficiently large portion of design space with reasonable density to achieve an accurate RSM over the design space. Each sample point provides a linear equation defining the quadratic RSM coefficients. Therefore, the calculation of the quadratic model coefficients is a simple matrix inversion problem as follows.

where $\mathbf{p} = (a_{11}, \ldots, a_{kk}, b_1, \ldots, b_k, c)$ is the model coefficients, $\mathbf{f} = (f_1, \ldots, f_n)$ is a vector of the simulated frequency at n sampled design points, \mathbf{S} is a matrix of quadratic terms of \mathbf{t} at the sampled design points, and $\mathbf{pinv}(\mathbf{S})$ is a pseudo-inversion of the matrix \mathbf{S} which gives the least mean-square error solution of the linear equations. Since the matrix \mathbf{S} is likely to be singular, singular-value decomposition (SVD) and minimum tolerance setting for the eigenvalues of \mathbf{S} must be conducted for the robust pseudo-inversion of \mathbf{S} .

In practice, we test the quadratic RSM using a 11-stage CML ring oscillator. The design space of the ring oscillator has four dimensions: the lumped capacitance at output node, bias current of the CML buffer, size of transistors, and load resistance. The model coefficients are calculated by the simulation results using more than 10 sample design points. Figure 4-14 shows the quadratic RSM fitting result. Figure 4-

14(a) shows the comparison between the simulated oscillation frequency over a full range of I_{TAIL} and the estimated frequency from the quadratic RSM. Figure 4-14(b) shows the scattering plot between the simulated and estimated frequencies at different R_L 's and device sizes. The estimation error in Figure 4-14(b) is 0.54%, which is much smaller than the 13.59% standard deviation of the simulated frequency.



Figure 4-14: Quadratic modeling fitting of the oscillation frequency of a CML bufferbased ring oscillator.

We next use the suggested quadratic RSM for the estimation of the self-oscillation frequency of a CML frequency divider. The time delay components of the RSM are as follows.

$$t_1 = C_L / g_{m,\text{DIFF}},$$

$$t_2 = C_L / g_{m,\text{LATCH}},$$

$$t_3 = R_L \cdot C_L,$$
(4.8)

where $g_{m,\text{DIFF}}$ and $g_{m,\text{LATCH}}$ are the transconductance values of the differential pair and latch devices, respectively. R_L and C_L are load resistance and lumped output capacitance, respectively. The quadratic RSM of the self-oscillation frequency is developed as a function of $\mathbf{x} = (t_1, t_2, t_3)$ and its fitting accuracy is shown in Figure 4-15(a).

In addition to the self-oscillation frequency, the frequency locking range of the

divider must be determined in order to calculate the maximum and minimum operating as in Equation (4.1). We review the analytic form of the divider locking range as follows (Equation (2.5) in Section 2.3.2).

$$\omega_L \approx k \cdot \frac{\omega_{SO}}{2Q} \cdot \frac{I_{INJ}}{I_{OSC}}$$

The divider locking range is a strong function of the self-oscillation frequency ω_{SO} , injected signal current swing I_{INJ} , quality factor of the oscillation loop Q, and current swing of the self-oscillation signal (bias current) I_{OSC} . Assuming that input signal swing is constant, I_{INJ} is proportional to the transconductance of the tail devices in Figure 2-6. ω_{SO} is already fitted to using t_1 , t_2 , and t_3 in Equation (4.8), and we add a new term as shown in Equation (4.9) to model the locking range ω_L .

$$t_4 = g_{m,TAIL} / I_{OSC}. \tag{4.9}$$

The loop quality factor Q is a complex function of $g_{m,DIFF}$, $g_{m,LATCH}$, R_L , and C_L [13]. We do not build a separate model for Q but still achieve a sufficient amount of fitting accuracy.

In our example, the coefficients of the RSMs for the self-oscillation frequency and locking range are calculated based on the simulation results over 125 sample design points. Similar to the CML ring oscillator, the design space covers the changes in the capacitance at output nodes, bias voltage, size of transistors, and load resistance. Figure 4-15 shows the fitting result of the self-oscillation and maximum operating frequency of the frequency divider for 0 dBm reference input signal. A 0.18μ m bulk CMOS model was used for the simulation. The fitting error for the divider performance is 0.90% and 1.47% for the self-oscillation and maximum operating frequencies, respectively. In the next section, we present experimental results based on using the quadratic RSM for the estimation of the parametric yield of the frequency divider in the presence of parameter variation in threshold voltage, channel length, parasitic capacitance, and sheet resistance.



Figure 4-15: Quadratic modeling fitting of the oscillation frequency of a CML bufferbased ring oscillator.

4.3.2 Estimation of Parametric Yield of PLL Front-End

Figure 4-16 shows the general concept of statistical analysis of circuit performance of the frequency divider. The design parameters to be considered are device sizes, load resistance, and bias conditions (e.g., V_{DD} , and V_{BIAS}). The variation parameters are sheet resistance R_{SH} , threshold voltage V_{TH} , and channel length deviation ΔL . The performance variables of interest are self-oscillation frequency, maximum operating frequency, and voltage swing of the output signal. For statistical analysis of circuit performance, the design and variation parameters must be provided to a simulation model for full performance evaluation. The values of the variation parameters are sampled from pre-defined distributions and correlation models. The accuracy of the yield estimation is dependent of the number of Monte Carlo runs including the variation parameter sampling and performance evaluation.

As described in Figure 4-9, the maximum operating frequency evaluation of the frequency divider is a time-consuming task. Instead, we use the proposed quadratic RSM developed in Section 4.3.1 for estimation of the maximum operating frequency and its statistical distribution. The quadratic RSM of the maximum operating frequency is fully based on DC circuit parameters such as transconductance, bias current, parasitic capacitance, and resistance. Thus, using the quadratic RSM can accelerate the yield estimation of the frequency divider significantly compared to the full SPICE



Figure 4-16: Estimation of the parametric yield of the frequency divider based on the statistics of variation parameters.

simulation-based estimation requiring transient analysis.

Figure 4-17 shows the procedure of the frequency divider yield estimation based on Monte Carlo simulation and DC analysis. First, a routine for the evaluation of the maximum operating frequency at sample design points is developed by a SKILL script in Spectre RF. We choose 125 design points to evaluate the self-oscillation and maximum operating frequencies and DC circuit parameters R_L , $g_{m,DIFF}$, $g_{m,LATCH}$, $g_{m,TAIL}$, I_{TAIL} , and C_L . The time delay components (t_1, \ldots, t_4) and their quadratic terms are calculated from the DC circuit parameters. The coefficients of the quadratic RSM are calculated from the evaluated self-oscillation and maximum operating frequencies at the sample design points. The tolerance for the pseudo-inversion is set to achieve minimum fitting and estimation RMS errors. Using the fitted RSM, the statistics of the self-oscillation and maximum operating frequencies are calculated from statistics of the DC circuit parameters, and the parametric yield of the divider can be estimated. The statistics of the DC circuit parameters are gathered from Monte Carlo simulation of DC analysis which is computationally efficient.

Figure 4-18 shows the experimental results of the self-oscillation and maximum operating frequency evaluation using the quadratic RSM. We introduce 20% $3\sigma/\mu$ variation for the threshold voltage of the devices, 15% for channel length, 21% for sheet resistance variation, and 21% for parasitic capacitance variation (interconnect to substrate capacitance). The $3\sigma/\mu$ variation of the self-oscillation and maximum



Figure 4-17: Procedure of yield estimation using DC Monte Carlo simulation results and functional approximation.

operating frequencies are 23.25% and 21.84%, respectively. The estimation errors are 0.58% and 1.02% for the self-oscillation and maximum operating frequencies, respectively, which are much smaller than the actual performance variation. This result verifies that the RSM-based yield estimation is fairly accurate enabling us to avoid full SPICE simulation.

The entire design space can be explored to investigate the design point with maximum yield by iterating the procedure in Figure 4-17 at a large number of design points. The design variables to be decided from the yield optimization are bias condition V_{BIAS} , load resistance R_L , and the size of differential pair devices W_1 . For each design point, we run 500 Monte Carlo runs to obtain a sufficiently small confidence interval of the estimated yield value. From experiments, the size of the 99% confidence interval for the estimates of μ and σ from Gaussian distribution is 0.01% and 0.09%, respectively. Figure 4-19 validates an approximately Gaussian assumption for



(b) Maximum operating frequency

Figure 4-18: Scattering plot of estimated and simulated self-oscillation and maximum operating frequencies in 500 Monte Carlo runs.



the self-oscillation and maximum operating frequency distributions.

Figure 4-19: Quantile-quantile normal plots of the simulated self-oscillation and maximum operating frequencies from 500 Monte Carlo runs.

In total, all possible combinations of nine different resistance values, six device sizes, and five bias conditions are simulated. For yield estimation at each combination, 500 Monte Carlo runs are performed. The elapsed time for the total simulation is less than five hours. Figure 4-20 shows the estimated parametric yield of the divider circuit over the design space for the given specifications in Table 4.2. Figure 4-20(a) shows the 3D yield surface of the yield over R_L and W_1 space at $V_{BIAS} = 1.05V$. The size of the latch and tail devices are adjusted based on W_1 . Figure 4-20(b) shows the yield surface estimated at $V_{BIAS} = 1.20V$. From the experimental results, 97% parametric yield can be achieved at $V_{BIAS} = 1.05V$, $R_L = 450 \Omega$, and $W_1 = 17 \mu$ m. The yield surface shows that the parametric yield is highly sensitive to the change of R_L .

Maximum Operating Frequency (F_{MAX})	13GHz
Minimum Operating Frequency (F_{MIN})	6GHz
Output Voltage Swing (V_{SW})	300 mV (0 dBm)

Table 4.2: Performance constraints of a 10GHz frequency divider in 0.18µm CMOS

4.3.3 Comparison of Dynamic and Static Yield Optimization

The parametric yield of the frequency divider has been optimized using dynamic and static methods in the previous sections. For the dynamic optimization, we showed that the bias condition of the frequency divider can be tuned for an individual die sample to maximize the number of dies satisfying given performance constraints. The design parameters can be optimized at design time to maximize the parametric yield by statistical yield analysis of circuit performance for given parameter uncertainties.

Both dynamic and static optimization methods can be combined to obtain additional yield benefit. From the simulation results of the static optimization in Section 4.3, the tunability of the bias condition can be adapted to observe the additional benefit on the parametric yield. Figure 4-21(a) shows the parametric yield of the frequency divider over the design space with dynamic tuning of V_{BIAS} . Figure 4-21(b) shows the comparison of the parametric yield with and without V_{BIAS} tuning. The dynamic V_{BIAS} tuning gives 2% additional yield benefit at the optimum design point obtained in Section 4.3.2. The benefit of dynamic V_{BIAS} tuning becomes large as the design point is further away from the static optimal point.

4.4 Summary

In this chapter, we discussed robust optimization methodologies for analog circuit design. We analyzed the performance variability of the VCO and frequency divider to investigate the main cause of functional failure by the variability. The trade-off in tuning the frequency operating range of the frequency divider was investigated in terms of parametric yield and performance degradation such as output phase noise. We showed how the parametric yield varies according to the change of the bias condition to validate the concept of dynamic yield optimization in the PLL front-end. We proposed an efficient numerical model for the evaluation of the maximum operating frequency of the frequency divider. Using the functional approximation of the frequency divider performance, the parametric yield of the PLL front-end circuit can be explored over the entire design space to optimize the parametric yield for the given parameter uncertainty. The static and dynamic optimization methods were jointly combined to evaluate additional yield benefit.



(b) $V_{BIAS} = 1.20V$

Figure 4-20: Functional yield of the frequency divider over the design space for the given specifications at $V_{BIAS} = 1.05V$ and $V_{BIAS} = 1.20V$.



(b) Comparison of w/and w/o dynamic compensation

Figure 4-21: Parametric yield of the frequency divider over the design space with dynamic V_{BIAS} tuning.

Chapter 5

Conclusion and Future Work

This thesis focuses on the design, analysis, and implementation of manufacturable mm-wave analog building blocks in a deeply scaled CMOS technology. We provided statistical measurement of circuit performance variation and robust optimization strategies to improve parametric yield. In this chapter, we summarize this thesis in Section 5.1. Ideas for future work as extensions of some of these contributions are discussed in Section 5.2.

5.1 Summary of Thesis

As the operating frequency of CMOS analog circuits reaches the mm-wave regime at 50GHz frequencies and beyond through the continuous scaling of MOS transistor feature size, parametric yield loss in analog circuit design by aggravated process variability becomes a serious concern. Multiple manufacturing steps contribute to the variation of CMOS device and interconnect parameters and overall circuit performance. A benchmarking task to characterize the performance-level variation properties of complex analog circuits is now required, in order to provide useful guidelines to the analog designer to build high yielding designs. To optimize parametric performance yield, statistical design methodologies have been suggested, whose goal is not only to meet the design specifications for nominal circuit performance, but also to achieve desired parametric yield in the face of process variability. In this thesis, we have contributed the design and implementation of a mm-wave PLL front-end circuit, containing a VCO and frequency divider as high-speed building blocks of a PLL. A 77GHz LC-VCO is designed to achieve a wide frequency tuning range for improved manufacturability by employing design features including a complementary topology for phase noise improvement and power reduction, an accumulation-mode varactor for a wide capacitance tuning ratio, relaxed pitch devices for reduced parasitic capacitance, and a high-Q inductor for good phase noise. A 2:1 frequency divider is designed to operate up to 90GHz of input frequency, using a ring-ILFD architecture for a wide frequency locking range. The self-oscillation frequency and frequency locking range of the divider are analyzed to obtain analytic forms of the divider performance, which are used for statistical optimization of the divider performance. The PLL front-end circuit is implemented in 65nm SOI CMOS with various technology features for high-speed analog circuit design.

The statistical measurement results of the VCO and frequency divider circuits were shown from a large number of high frequency measurements over multiple wafers. The measurement data for 12 wafers and 76 dies from each wafer reveals trends of variation in the VCO and frequency divider. The statistics of VCO measurement data show an improved frequency tuning range and parametric yield by the device and circuit optimization. A strong systematic wafer-level pattern of the variability in the performance of the frequency divider is observed. The performance variability of the VCO and frequency divider was analyzed to investigate the main cause of functional failure by the variability.

An extraction scheme to obtain circuit parameter variation from the frequency divider measurement data was proposed. The extraction method is based on the sensitivity analysis of the divider self-oscillation frequency to the deviation of critical circuit parameters such as threshold voltage, load resistance, and lumped parasitic capacitance. Experimental results show the statistics of each circuit parameter and the important correlation between the parameters. Furthermore, process variation in electrical parameters of circuit components can be decomposed into die-to-die and wafer-to-wafer variation using a constrained principal component analysis. Calculated constrained principal components from primitive test structures can explain a significant amount of variance in the self-oscillation frequency of the divider, showing that the variation of complex circuit performance in a new technology can be predicted by the measurement data of a primitive test structure set.

We discussed general robust optimization methodologies for analog circuit design, and specialized static and dynamic optimization schemes for the VCO and frequency divider. As an example of dynamic optimization, we verified that the parametric yield of the PLL front-end can be improved by tuning the bias condition of the frequency divider, while there exists a trade-off between the parametric yield and phase noise performance. To enable static optimization, we propose functional approximation of the maximum operating frequency of the frequency divider. Using the functional approximation, the parametric yield of the PLL front-end circuit can be estimated over the entire design space within a reasonable amount of simulation time, and the optimal design point can be investigated.

5.2 Future Work

As noted in the ITRS road map, sub-100nm CMOS technologies have become one of the viable solutions for the implementation of mm-wave applications [8]. Compared to SiGe and III-V technologies which have been conventionally used for mm-wave applications, CMOS has a strong advantage in low cost and high density integration. However, there are negative scaling impacts on analog circuit design as well, and in particular, aggravated process variability of the device and interconnect performance causing parametric yield loss becomes one of the main design concerns [77].

To achieve high yielding analog circuits as building blocks of high frequency mixedsignal applications, parametric yield of such analog circuits must be predicted at the design stage and proper variability mitigation techniques must be utilized. In this thesis, we showed successful integration of mm-wave analog building blocks in a CMOS technology and verified their manufacturability based on high frequency measurement data over multiple wafers. Among the contributions of this thesis, the methodology used for extraction of important circuit parameters could be extended to provide a general benchmarking scheme for fast characterization of analog design aspects of new technologies. Furthermore, an abundant resource of digital processing in scaled CMOS technologies can provide opportunities for the dynamic compensation of performance variability in analog circuits. Here, we discuss the future trend of process variation in analog circuits, and suggest analog technology benchmarking and digitally-assisted robust analog design as future work for robust analog circuit design.

5.2.1 Future Trend of Process Variation in Analog Circuits

The enhancement of performance and power consumption in CMOS circuits driven by the feature size scaling will continue in analog and digital circuits. Together with increasing manufacturing cost in high-end CMOS processes, process variability and corresponding parametric yield loss will continue to be an increasingly challenging issue in future process nodes. In digital circuit design, it is well known that the leakage power consumption will be a dominant factor in the variation in total power consumption, due to the exponential dependency of leakage current to threshold voltage variation. However, it is difficult to define a general future trend of variation in analog circuits since analog circuits have a large number of performance variables to be considered, and the variation of each performance is a mixture of multiple sources of variation with different physical characteristics and statistical properties.

Generally, the physical sources of variation can be categorized into systematic and random components. From the measurement results and statistical analysis of the VCO and frequency divider in Chapter 3, we verified that the performance variation has a large systematic component. Typically, wafer-level variation in critical dimension (CD) of transistor gates and back-end variation by etching and CMP nonuniformity are primary causes of the systematic variation in circuit performance. While the amount of systematic variation is largely determined by the quality of manufacturing control, various design-level regularization techniques have been used to reduce the systematic variation significantly (e.g., etch and CMP dummies in layout). Restrictive design rules (e.g., uni-directional poly-silicon gates, layout density rules, etc.) also help to improve the regularity of the active and passive device performance across dies and between wafers. As a result, together with the improvement in physical manufacturing control, one can expect that the absolute amount of die-todie variation of passive component oriented circuits (e.g., LC-VCO) will be reduced by the regularization techniques. Also, the reduction of wafer-level CD variation can mitigate the die-to-die variation of channel length dependent device and circuit characteristics (e.g., frequency divider performance when $V_{BIAS} \gg V_{TH}$). However, as the operating frequency and required precision of analog circuits increases in advanced applications, the relative impact of the systematic variation can also increase.

On the other hand, the random portion of process variation generally keeps increasing as geometric feature size shrinks down. Typically, threshold voltage variation by random dopant fluctuation and line-edge roughness in transistor gates and metal wires are major sources of the random variation in circuit performance. The theoretical degradation of threshold voltage matching between two devices with smaller feature size is well predicted in Pelgrom's model. The impact of the line-edge roughness will become more significant as the physical dimension of the device and metal wire shrinks, because the gap between the wavelength of photo-lithography and actual feature size has increased since the 180nm process node and likely will keep increasing [102]. The portion of the random variation in total performance variation has increased significantly and the trend will continue in future process nodes [81]. Therefore, the parametric yield loss of match sensitive analog circuits (e.g., SRAM cells, analog-to-digital converters, phase noise of ring oscillators, etc.) will become an increasingly challenging issue in future process nodes.

Alternatively, the process variability of circuit performance can be improved dramatically by the introduction of new technologies. For example, the introduction of metal gates in a CMOS technology reduces threshold mismatch due to variations in gate doping, while improving the maximum oscillation frequency F_{MAX} by reduced gate resistance. The measurement results in [56] show that the process variability of CMOS ring-oscillator performance has been reduced in 45nm high-k metal gate technology compared to 65nm poly-silicon gate technology. Metal wire resistance variation after CMP has also been significantly mitigated by the improvement of process control in etching and CMP in the 45nm technology. Though new technologies will be able to provide discrete improvement in variation trends, the feature size scaling based variation trend will continue after each introduction of the new technology.

5.2.2 Technology Benchmarking for Analog Circuit Design

The demand for rapid time-to-market of high performance mixed-signal applications has become more stringent and challenging due to the fast evolution of technology. Though innovative circuit ideas are still a considerable contributor of circuit performance enhancement, the improvement in the device and interconnect performance by technology evolution (e.g., scaling, or stress-induced mobility enhancement) becomes a more dominant factor in today's semiconductor manufacturing.

Conventionally, designers need model parameters for active and passive devices to verify performance benefits and re-optimize previous circuit designs in a new technology. However, characterizing a complete set of the model parameters (e.g., in BSIM4 or PSP compact models) is an extremely time-consuming task, and it can be highly costly given the shortened life cycles of today's process nodes. Instead, in advance of the complete characterization of the model parameters, the new technology can be benchmarked by using well-designed test structures whose measurement data can enable early and accurate prediction of circuit performance enhancement. Furthermore, potential problems such as aggravated process variability in certain circuit performance variables can be characterized in advance and feedback can be provided for fine-tuning of the manufacturing process.

Technology benchmarking has been conventionally used in digital circuit design. Table 5.1 shows common metrics used in the characterization of a digital microprocessor manufacturing process. An array of test devices in different configurations in terms of size, device option, and bias condition are implemented in test fabrication runs to measure the DC performance of devices. For test of delay and power consumption in digital logic, ring oscillators are commonly used as a benchmarking unit.

	Parameter	Description	
Devices	I _{DS,LIN}	Drain current in the linear region	
	I _{GATE}	Gate leakage current	
	I _{OFF}	Off current (subthreshold leakage)	
	I _{ON}	Saturation current when)	
	$V_{TH, LIN}$	Threshold voltage in the linear region	
	$V_{TH, SAT}$	Threshold voltage in the saturation region	
Ring oscillators	I _{DDA}	Active current	
	I _{DDQ}	Quiescent current	
	Frequency	Oscillation frequency	

Table 5.1: Common metrics for the characterization of a digital microprocessor manufacturing process.

The oscillation frequency of the ring oscillator for different configurations indicates the time delay of logic gates, including the interconnect impact. Active and quiescent currents of the ring oscillator provide information to calculate the active and leakage power consumption of digital logic and microprocessors.

In contrast to digital circuits, analog circuits have a more diverse set of performance variables, as discussed in Chapter 4. Even though some of the metrics in Table 5.1 can be shared for analog benchmarking, a unique set of metrics to compare the benefits of new technology nodes in analog circuit design are strongly required since there are more performance variables beyond delay and power to be considered in analog circuit design. Traditionally, cutoff frequency (f_T) and maximum oscillation frequency (f_{MAX}) are the metrics used to compare the performance capability of analog manufacturing processes. RF performance defined by f_T and f_{MAX} of CMOS devices and interconnects in 65nm and 45nm SOI CMOS was reported in [61] and [60], respectively.

The list below shows several important metrics for analog circuit design.

- Cut-off frequency f_T and f_{MAX} of the devices
- Transconductor efficiency g_m/I_D
- Available voltage swing
- Intrinsic gain g_m/g_{ds}

- Noise figure NF
- Input offset ΔV_{IN}
- Temperature coefficient

For benchmarking of the metrics above, well-designed test structures for efficient measurement of the analog performance metrics are needed. Several example test structures to measure the metrics above were suggested in [91]. In this thesis, we showed that the self-oscillation frequency of the divider can be used as a variation probe for high frequency circuit parameters. As shown in Section 2.3.2, the analytic form of the divider self-oscillation frequency g_m/C_P implies the cut-off frequency of the device $(f_T=g_m/C_{gs})$ while C_P is a lumped sum of the C_{gs} of differential pair and latch devices and C_{wire} . While direct measurement of multi-hundred gigahertz f_T is extremely difficult, the variability of f_T can be extrapolated from the self-oscillation oscillation frequency measurement statistics.

In addition, we verified in Section 3.4.1 that the variability of critical circuit parameters can be extracted by using the self-oscillation frequency measurement data at multiple divider bias conditions. The number of parameters to be characterized can be increased by using multiple configurations of the divider circuit to reduce the co-linearity between the deviations of some parameters, which introduces a singularity problem during the matrix inversion in Equation (3.4).

Obtaining performance characteristics and their variation statistics using mmwave measurement equipment is still a costly task. To reduce the cost of measurement, on-chip digital measurement interfaces are needed for efficient characterization of mmwave circuit performance. In this work, the oscillation frequency can be divided down further by using deep sub-dividers until it can be measured by conventional digital counters. Such digital interfaces can enable an array of mm-wave test structures to characterize within-die variation of the VCO and frequency divider performance; understanding this within-die variation is likely to be important in the implementation of mm-wave imaging system which use an array of PLLs within each chip for fineresolution of images.

5.2.3 Digitally-Assisted Robust Analog Design

Before it became a critical problem in digital circuit design, process variability was a well-known problem in analog circuit design. A wide range of analog circuits are highly matching sensitive, and within-die variation in active and passive components can cause serious performance degradation such as a voltage offset and non-linearity. Various techniques have been proposed and used to reduce the variability impact on analog circuit performance (e.g., symmetric layout, analog or digital calibration, dynamic element matching, larger device area).

While the speed and power performance of CMOS devices have been continuously improved, the matching property has been degraded by technology scaling. On the other hand, as the cost of complex digital functionality has been reduced by scaling, the design-overhead of utilizing digital calibration in analog circuits has also decreased. As a result, digitally-assisted analog design becomes a substantial trend in high performance and high precision analog circuit design [47]. In precision oriented analog circuits such as analog-to-digital converters, digital calibration techniques for the mismatch between devices and capacitors have been intensively studied for decades [58] [78].

In the PLL front-end circuit in this work, we verified that adaptive tuning of the bias voltage (V_{BIAS}) of the frequency divider can improve overall parametric yield. For actual implementation of the adaptive bias tuning, digital calibration is a promising design option for its low cost and versatile usage. Figure 5-1 shows the concept of digital calibration of V_{BIAS} using in-situ frequency measurement circuitry, a digital processor, and digital-to-analog converters (DACs). The digital processor must be equipped with an efficient algorithm to find an optimal V_{BIAS} by measuring the VCO and divider characteristics at different conditions of V_{BIAS} . The output frequency of the frequency divider must be deeply divided down using a chain of frequency dividers to be measurable using a digital counter. Thermal and flicker noise of DAC outputs must be carefully considered, since these are directly related to the phase noise of the divider output. Digital calibration can provide additional benefits such as the



Figure 5-1: Digital feedback control of V_{CTRL} and V_{BIAS} and divided frequency measurement to set optimized V_{BIAS} voltage.

compensation of aging impact on circuit performance by low frequency calibration. The output performance variables to be tuned by digital calibration can be extended to phase noise, power consumption, I/Q mismatch.
Appendix A

Process Variability and Statistical Yield Analysis of High-Speed Ring Oscillators

The design and test of analog RF building blocks in SoC applications have been challenging due to the high-frequency operation and broad specifications of analog RF circuits. Considering the *chip-limited yield* (CLY) failure due to analog performance shortfall, analog *design-for-test* (DFT) methodologies for RF test are essential to improve testability and yield estimation of analog building blocks.

The main difficulty of analog design for RF DFT lies in signal speed. In a mixedsignal SoC, analog and RF function blocks are driven to the maximum speed to provide clocks in digital systems and carrier frequencies in communication channels. The operating frequency of the analog front-end circuits is higher than the affordable speed of digital interfaces. Therefore, analog RF circuits are unlikely to have systematic on-chip assistance to measure the performance variables accurately. Severe signal attenuation in high frequency bands is another reason for the difficulty of measurement unless proper RF buffering is provided. In total, it is difficult to have a standardized digital interface as used in digital DFT, and the test equipment and cables for RF test must be highly customized depending on circuit operating conditions and output variables. Furthermore, as CMOS scaling continues to enhance the system performance and power, process variation in the performance of nano-scale device and interconnect becomes a serious issue [79]. In particular, for SoC applications, analog and RF building blocks can be a critical reason for limited overall system yield since the analog and RF blocks are operating at the highest frequency of the system, as they are sensitive to die-to-die and within-die variations in the deeply scaled CMOS process [14]. Therefore, the variation of analog and RF circuit performance must be fully characterized using dedicated test structures for accurate yield learning [91]. Systematic approaches are needed to test the various output performance variables (e.g., gain, noise, power, frequency range, and etc.) of analog and RF circuits over a number of samples within a short period of time.

The main contributions of the RF DFT in this chapter are as follows.

- The integration of the analog design and automated RF test and measurement plan for multiple performance variables at an early design stage.
- Implementation of assistance circuitry for reliable automated RF measurements over a number of analog circuit samples.





Figure A-1: Process flow diagram of analog design for RF test platform.

	Digital DFT	RF DFT		
Motivation	Structural test	Test feasibility		
Problem	Complexity	Speed		
Design	System	Ckt.+Env.		
Test setup	Standard	Custom		
Test process	Test and verify	Verify while test		
Test case	Automatic	Manual		

Table & 1. Digital DET are DE DET

entire wafer.

• Yield analysis of a high-speed VCO based on measured statistics of phase noise and output frequency.

The RF DFT process flow in Figure A-1 captures the framework and methodology that produce test-friendly RF circuits and reliable measurements. The circuit design, test setup, verification, and qualification to prepare scalable statistical tests are described in Section A.2. The statistical test of the performance of a 6GHz widely tunable VCO is demonstrated as an example in Section A.3. The yield model of the VCO in a 130nm CMOS technology is developed, and the circuit yield in the presence of phase noise and frequency variability is estimated in Section A.4. The analog design and test plan are interactively used through the whole RF DFT process.

Analog Design for RF Test A.1

Digital DFT and RF DFT have common and distinct characteristics as depicted in Table A.1. The digital DFT is focused on the functional test of logic blocks or the characterization of maximum operating frequency and power consumption. The analog DFT mainly considers feasibility of circuit performance measurement by using on-chip test circuitry or off-chip equipment.

Previously, analog DFT has been considered to focus on detecting soft failures

in analog circuits by using various fault sensors, or measuring specific performance variables using on-chip test circuitry. In [6], oscillation-based fault detection is implemented. The variation sensor outputs must be efficient to measure, and the decision is based on perfect correlation between the sensors and actual circuits. However, recent characterization results show that the within-die variation of threshold voltage is spatially uncorrelated, and therefore, the sensor cannot provide accurate estimation of circuit performance. Dedicated on-chip measurement circuitry can provide efficient measurement of analog performance (e.g., phase noise measurement of PLL in [51]). However, the measurement circuit must be highly customized and cannot be applied to multiple performance variables.

The RF DFT engages analog circuit design, specifications, and test strategies at an early design stage. The design must include circuit schematic, layout, interface, and test environment. For testability, the interfaces are customized using output buffers, pad sets, RF+DC probes, cables, and measurement equipment. The test case is manually generated according to the design specifications and test environment. The pass or fail of functional measurements is determined by pre-defined performance parameter ranges.

The RF DFT is applied to the design and test of a VCO for a phase-locked loop (PLL) in microprocessors as illustrated in Figure A-2. The VCO has an 11-stage delay chain to reduce process-induced variability by averaging. Each delay stage consists of a current-mode-logic (CML) differential amplifier with tail current control for frequency tuning. The VCO has been implemented in 130nm CMOS on a 200mm wafer. The oscillation frequency is controlled by the reference current input I_B . The VCO determines the CLY of the entire chip, and the *frequency tuning range* (FTR) must be wide enough to meet the specification frequency in the presence of process variation. Also, the phase noise of output signals needs to be lower than a specification limit.

The VCO specifications are listed in Table A.2. The specifications include static power consumption, FTR, dynamic power consumption, phase noise, and *figures-ofmerit* (FoM). The FTR and phase noise are the most critical performance variables



Figure A-2: A schematic diagram of 11-stage CML VCO implemented in 130nm CMOS.

for the VCO and overall system operation. The FOM is used for fair comparison of the VCO phase noise.

In the RF DFT, several design modifications and environmental variables must be applied to support automated quality measurements: 1) RF and DC interfaces, 2) pad pitch and size, 3) probe set, 4) RF buffer circuit, 5) on/off-chip DC decoupling, 6)



Figure A-3: VCO layout with testing environment considering RF DFT.

Entity	Unit	Equipment	Specification	
Active current	mA	SMU	40mA	
Quiescent current	mA	SMU	$1 \mathrm{mA}$	
Frequency	GHz	Spectrum	$f_0=5.5\mathrm{GHz}$	
FTR (%)	%	analyzer	FTR=40%	
Output power	dBm		-20dBm	
Phase noise	dBc/Hz	PLL	-120	
FoM	@1MHz	analyzer	-160	

Table A.2: VCO Performance specification for RF DFT

cable connection, 7) measurement equipment and system, and 8) test code for automatic measurement. Especially, RF buffer and DC decoupling are important for phase noise and FoM measurements. This is because the *circuit-under test* (CUT) is surrounded with and connected to noise sources, such as a DC *source-and-measurement unit* (SMU), a control PC, and a wafer stepping tester.

Figure A-3 shows the layout and pad assignment and design assistants for efficient automated probing of the VCO circuit. The CUT requires two DC inputs and one RF output, and 150μ m-pitch pads and probe sets are used. All RF pads are assigned on the right side to reduce the number of high-frequency probe sets. Since the contact between automatically-controlled probes and pads are not as solid as wire bonding, on-chip high-density bypass capacitors (CB1 and CB2) were inserted extensively to reduce the power supply noise induced by the weak contact. A common source RF buffer is added to improve output signal power. The VCO CUT is testable at the M3 level, reducing the RF DFT process cycle time for circuit debugging.

A.2 Measurement Setup

There are several practical issues to match the CUT well with the external test environment, and the issues must be resolved in the design stage. The instrument measurement ranges must be carefully inspected to meet the circuit specifications from simulation results. The availability of test instrument must be checked in time.



Figure A-4: Measurement setup diagram for the VCO.



Figure A-5: Manual screen captures of VCO output spectrum with 50MHz span and the phase noise at 6.94GHz.

Estimated signal losses in on-chip interconnects, probe contact resistance, probe sets, cables, and a bias-T must be considered. Custom probes are fabricated if necessary, and the wear-out of the probe set, RF connector, cable, adapter, and bias-T must be checked in advance. Having spares of fragile RF test equipment is recommended since delayed yield learning of product designs can be costly. The automated test code must be prepared and debugged based on the simulation results.

Figure A-4 shows a test setup diagram for the VCO. DC SMUs provide operating conditions to the CUT through cable and probe set. The probe set is held by a wafer tester, and the tester performs robotic movements across the wafer. The RF output of the CUT is collected by the RF probe and bias-T. The power splitter divides the



Figure A-6: VCO output frequency for input current I_B and the output signal power. signal to the spectrum analyzer and the PLL analyzer. The PLL analyzer shows faster and more accurate phase noise measurement in real time than a spectrum analyzer. A spectrum analyzer is used for signal verification and oscillation frequency measurement. All of the instruments are connected through GPIB, and a PC controls instrument operations. The measurement results are transferred to the PC and their validity is evaluated with algorithms based on the VCO spectrum and phase noise models. For example, an accurate oscillation frequency is reliably determined by zooming in the measuring frequency range iteratively. The test results are recorded in a central test database, containing wafer and site information, and test conditions.

The measured VCO output spectrum and phase noise screen shots are shown in Figure A-5. The initial test shows that the VCO and the test setup are functional. The phase noise plot does not show noticeable noise modulation. In practice, several factors such as the tester condition and global ground should be modified to isolate noise sources. Once the minimal circuit function is verified, a set of repetitive tests are performed based on the pre-determined test plan. The FTR and the RF output power of the VCO are obtained with automated test code as plotted in Figure A-6. The plots show the VCO output frequency versus input current I_B and RF output power, all sampled by a spectrum analyzer. The measured VCO minimum frequency is 4.6GHz

at $I_B=1$ mA, and the maximum is 7.4GHz at $I_B=25$ mA. The FTR is 2.8GHz, or 46.7% around a center frequency.

The other important CUT specification is phase noise, and the phase noise curves at different I_B s are shown in Figure A-7. The plot shows that the phase noise curves have regular slope overall. At the minimum oscillation frequency, the VCO output tends to have worse phase noise as plotted in Figure A-8. There are more abnormalities involved at 1MHz offset in the mid- I_B region as in Figure A-8. The plots suggest that the phase noise should be sampled at larger offset frequencies such as 2, 5, and 10MHz for reliable measurement. The PLL analyzer noise floor is below the minimum phase noise at 10MHz, verifying that the measurement result at 10MHz offset is actual VCO phase noise, not the instrument noise. Moreover, the phase noise must be measured at both the highest and lowest frequencies, as in Figure A-9, to verify that the VCO meets the phase noise specification over the operating range. The phase noise is assumed to change monotonically with I_B . The phase noise measurements tend to have a noisy curve, and there is a trade-off between the measurement time and the averaging, with respect to the test scalability.

The test procedure must be iterated to modify the parameter capturing algorithm and to improve the reliability of the measurement against environmental variation



Figure A-7: VCO phase noise output with input current I_B sweep.



Figure A-8: VCO phase noise at 1, 2, 5, and 10MHz offset with I_B sweep.

and anomalies. When confidence is established in the circuit operation, environment, control, and algorithm, the test set up is qualified for statistical tests.



Figure A-9: VCO phase noise at $I_B=1$ mA and 15mA.

A.3 Statistical RF Test

The test qualification for statistical measurement must be conducted on several sampled sites on a single or multiple wafers, before proceeding to massive (high data volume) measurements over a number of wafers. The test results must be qualified based on acceptance criteria assuming possible errors during the large number of measurements. The statistical tests were performed on a 200mm wafer including 46 dies.

Figure A-10 shows maximum and minimum tunable frequencies (F_{max} and F_{min}) of all 46 dies. The data is sorted on a descending order of the calculated FTRs. This measurement result shows the statistics of F_{max} at $I_B=12$ mA and F_{min} at $I_B=2$ mA including mean, standard deviation, and correlation between each other. The size of a common FTR over all dies is also estimated. The VCO yield defined by the minimum FTR specification (40%) can be calculated based on the measurement data and suggested yield model in Section A.4.

The measured phase noise values at 10MHz offset when $I_B=2$ and 12mA over all dies are shown in Figure A-11. The site index is the same as in Figure A-10.



Figure A-10: Statistical measurement on VCO maximum and minimum tunable frequency.

Table A.3: CML VCO RF Performance Statistics							
Entity	Average	σ/μ (%)	Median	Minimum	Maximum		
$I_A @F_{max} (mA)$	53.2	5.86	53.6	39.8	56.8		
$I_Q (\mathrm{mA})$	1.61	15.8	1.55	1.44	2.85		
F_{min} (GHz)	4.23	4.33	4.24	3.83	4.58		
F_{max} (GHz)	6.35	5.58	6.35	5.65	7.06		
FTR (GHz)	2.12	9.76	2.14	1.71	2.50		
FTR (%)	40.0	6.09	40.7	32.5	43.3		
$P_{RF}@F_{max}$ (dBm)	-4.40	15.7	-4.24	-6.90	-3.50		
Phase noise	-123.5	2.88	-124.3	-129.5	-113.0		
(dBc/Hz@10MHz,							
$I_B=2\text{mA})$							
Phase noise	-129.8	1.20	-129.7	-132.6	-126.9		
(dBc/Hz@10MHz,							
$I_B=12\text{mA})$							
FoM (dBc/Hz)	-166.1	0.91	-165.9	-169.1	-163.3		

The phase noise at $I_B=2\text{mA}$ shows stronger fluctuation and the phase noise variation does not show strong correlation to the variation of FTR at both I_B conditions. The mean and standard deviation are used to construct a phase noise yield model. The statistical RF test result is summarized in Table A.3.

A.4 Yield Analysis

We consider the frequency tuning range and phase noise to define sound samples for yield calculation. Figure A-12 shows the distribution of phase noise at 10MHz offset and output frequency variations at F_{min} and F_{max} of the VCO. At F_{min} ($I_B=2$ mA), the yield is limited by the phase noise specification since the phase noise shows a wide spread. On the other hand, at F_{max} ($I_B=12$ mA), the output frequency variation shows a wide spread and becomes a critical concern for yield estimation.

We provide the estimation of circuit yield against frequency variation at F_{max} as an example. To estimate the circuit parametric yield in manufacturing, the variation must be decomposed to within-die, die-to-die, wafer-to-wafer, and lot-to-lot variation. For most mixed-signal SoC applications, the die area consumed by analog building blocks is extremely small compared to digital circuits. Therefore, the within-die variation is not considered to be a serious issue in analog circuits, unless the circuit is a repetitive large array and the performance is highly sensitive to device mismatch, such as in SRAM cells.

The die-to-die variation over a wafer is one of the main concerns in overall variation. The slowest and fastest dies in a 200mm wafer in our example show 22.2% range of F_{max} , and the $3\sigma/\mu$ of F_{max} is 16.74%. The die-to-die frequency variation over a wafer is seen to be highly systematic; the cross-correlation between different wafers is more than 90% in most cases [67]. We model the variation such that there is a mean-shift between wafers caused by the wafer-to-wafer variation, while σ/μ of each wafer is preserved.

The cause of wafer-to-wafer variation can be either temporal or spatial. The temporal wafer-to-wafer variation is caused by the drift of process conditions in single wafer processing. While the wafer-level variation is likely to be systematic, the waferto-wafer variation is time and equipment dependent and we assume it as random in this analysis.

By using the RF DFT, the variation pattern of the VCO output frequency tuning range over a 200mm wafer has been fully characterized as in Table A.3. Figure A-13



Figure A-11: Statistical measurement on VCO phase noise with I_B =2mA and 12mA @10MHz offset.



Figure A-12: VCO F_{max} ($I_B=12$ mA) and F_{min} ($I_B=2$ mA) versus phase noise distribution.

shows our variation model including die-to-die variation over a single wafer and waferto-wafer variation. For each wafer, $3\sigma/\mu$ is preserved at 16.74%. The mean shift of each wafer is assumed to come from a normal distribution with zero mean and σ_{W2W} standard deviation, and is added to all dies in a wafer to model the wafer-to-wafer variation.

The probability of the VCO maximum tunable frequency (F_{max}) not satisfying the minimum requirement $F_{MAX, MIN}$ can be calculated as follows.

$$P(f < \mathcal{F}_{\text{MAX, MIN}}) = \int_{-\infty}^{\infty} Q\left(\frac{\mathcal{F}_{\text{MAX, MIN}} - m(w)}{\sigma(w)}\right) f_{\sigma_{w2w}}(w) dw$$
(A.1)

where $Q(x) = \int_{-\infty}^{x} \frac{1}{\sqrt{2\pi}} e^{-\frac{x^2}{2}} dx$, $f_{\sigma}(w) = \frac{1}{\sqrt{2\pi\sigma}} e^{-\frac{w^2}{2\sigma^2}}$, $m(w) = m_{ref} - w$, and $\sigma(w) = m(w) \cdot \frac{\sigma_{ref}}{m_{ref}}$. Here, m_{ref} and σ_{ref} are the mean and standard deviation of the reference wafer with zero mean shift. The yield can be calculated numerically using the characterization results from the measurement. Assuming the minimum $F_{MAX, MIN}$ specification is 6.35GHz for 40% FTR, the calculated circuit yield is 51%. Similar analysis can be applied to the phase noise constraint at F_{min} .

A.5 Summary

A framework for analog design for RF test has been developed for the statistical measurement of yield-limiting analog building blocks in mixed-signal SoC applications. A systematic measurement plan for high-speed analog circuits is critical for fast and accurate yield learning. A 6GHz ring-based CML VCO was implemented in a 130nm CMOS technology with circuit assistance for automated measurements. The frequency tuning range, phase noise, and power consumption of the VCO and their statistical properties were characterized over a 200mm wafer. The measurement results show an important trade-off between the variability of phase noise and output frequency at different bias conditions of the VCO. The yield of the VCO circuit for given phase noise and FTR specifications is estimated based on the proposed variation model.



Figure A-13: Variation model including wafer-level die-to-die variation and random wafer-to-wafer variation.

Bibliography

- RF MEMS for automotive radar (77GHz), KTH Royal Institute of Technology. http://www.s3.kth.se/mst/research/.
- [2] Lumera Corporation. http://www.lumera.com, 2007.
- [3] A.A. Abidi. Phase noise and jitter in CMOS ring oscillators. IEEE Journal of Solid-State Circuits, 41(8):1803–1816, August 2006.
- [4] R. Adler. A study of locking phenomena in oscillators. Proceedings of the IEEE, 61(10):1380–1385, October 1973.
- [5] K.J. Antreich, H.E. Graeb, and C.U. Wieser. Circuit analysis and optimization driven by worst-case distances. *IEEE Transactions on Computer-Aided Design* of Integrated Circuits and Systems, 13(1):57-71, Jan 1994.
- [6] K. Arabi and B. Kaminska. Oscillation-test strategy for analog and mixedsignal integrated circuits. In *Proceedings of IEEE VLSI Test Symposium*, pages 476–482, 1996.
- [7] A. Asenov, A.R. Brown, J.H. Davies, S. Kaya, and G. Slavcheva. Simulation of intrinsic parameter fluctuations in decananometer and nanometer-scale MOS-FETs. *IEEE Transactions on Electron Devices*, 50(9):1837–1852, September 2003.
- [8] S.I. Association. International Technology Roadmap for Semiconductors (ITRS). http://www.itrs.net, 2007.

- [9] H.S. Bennett, R. Brederlow, J.C. Costa, P.E. Cottrell, W.M. Huang, Jr. Immorlica, A.A., J.-E. Mueller, M. Racanelli, H. Shichijo, C.E. Weitzel, and B. Zhao. Device and technology evolution for Si-based RF integrated circuits. *IEEE Transactions on Electron Devices*, 52(7):1235–1258, July 2005.
- [10] K. Bernstein, D.J. Frank, A.E. Gattiker, W. Haensch, B.L. Ji, S.R. Nassif, E.J. Nowak, D.J. Pearson, and N.J. Rohrer. High-performance CMOS variability in the 65-nm regime and beyond. *IBM J. Res. Dev.*, 50(4/5):433-449, 2006.
- [11] R.E. Best. Phase-locked Loops: Design, Simulation and Applications. McGraw-Hill, 2003.
- [12] L. Bisdounis, S. Nikolaidis, and O. Loufopavlou. Propagation delay and shortcircuit power dissipation modeling of the CMOS inverter. *IEEE Transactions on Circuits and Systems I: Fundamental Theory and Applications*, 45(3):259–270, Mar 1998.
- [13] A. Bonfanti, A. Tedesco, C. Samori, and A.L. Lacaita. A 15-GHz broad-band ÷/2 frequency divider in 0.13μm CMOS for quadrature generation. *IEEE Mi*crowave and Wireless Components Letters, 15(11):724–726, November 2005.
- [14] J. Bordelon, B. Tranchina, V. Madangarli, and M. Craig. A stratagy for mixedsignal yield improvement. *IEEE Design & Test of Computers*, 19(3):12–21, May/June 2002.
- [15] S. Borkar, T. Karnik, S. Narendra, J. Tschanz, A. Keshavarzi, and V. De. Parameter variations and impact on circuits and microarchitecture. In *Proceedings* of the ACM/IEEE Design Automation Conference, pages 338–342, June 2003.
- [16] K.A. Bowman, S.G. Duvall, and J.D. Meindl. Impact of die-to-die and withindie parameter fluctuations on the maximum clock frequency distribution for gigascale integration. *IEEE Journal of Solid-State Circuits*, 37(2):183–190, February 2002.

- [17] G.E.P. Box and N.R. Draper. Empirical model-building and response surface. John Wiley & Sons, Inc., New York, NY, USA, 1986.
- [18] S. Boyd, S.-J. Kim, L. Vandenberghe, and A. Hassibi. A tutorial on geometric programming. Optimization and Engineering, 8(1):67–127, 2007.
- [19] A. Cabrini, D. Cantarelli, P. Cappelletti, R. Casiraghi, A. Maurelli, M. Pasotti, P.L. Rolandi, and G. Torelli. A test structure for contact and via failure analysis in deep-submicrometer CMOS technologies. *IEEE Transactions on Semiconductor Manufacturing*, 19(1):57–66, February 2006.
- [20] H. Cai. Modeling of Pattern Dependencies in the Fabrication of Multilevel Copper Metallization. PhD thesis, Massachusetts Institute of Technology, June 2007.
- [21] C. Cao and K.K. O. A 90-GHz voltage-controlled oscillator with a 2.2-GHz tuning range in a 130-nm CMOS technology. In *IEEE Symposium on VLSI Circuits Digest of Technical Papers*, pages 242–243, June 2005.
- [22] C. Cao and K.K. O. Millimeter-wave voltage-controlled oscillators in 0.13μm CMOS technology. *IEEE Journal of Solid-State Circuits*, 41(6):1297–1304, 2006.
- [23] C. Cao, E. Seok, and K.K. O. 192 GHz push-push VCO in 0.13 μ m CMOS. Electronics Letters, 42(4):208-210, February 2006.
- [24] E. Cassan, P. Dollfus, S. Galdin, and P. Hesto. Calculation of direct tunneling gate current through ultra-thin oxide and oxide/nitride stacks in MOSFETs and h-MOSFETs. *Microelectronics Reliability*, 40:585–588, April 2000.
- [25] F. Chen, F. Ungar, A.H. Fischer, J. Gill, A. Chinthakindi, T. Goebel, M. Shinosky, D. Coolbaugh, V. Ramachandran, Y.K. Siew, E. Kaltalioglu, S.O. Kim, and K. Park. Reliability characterization of BEOL vertical natural capacitor

using copper and low-k SiCOH dielectric for 65nm RF and mixed-signal applications. *Proceedings of IEEE International Reliability Physics Symposium*, pages 490–495, March 2006.

- [26] T. Chen and S. Naffziger. Comparison of adaptive body bias (ABB) and adaptive supply voltage (ASV) for improving delay and leakage under the presence of process variation. *IEEE Transactions on Very Large Scale Integration Systems*, 11(5):888–899, October 2003.
- [27] C.S. Chiu, G.W. Huang, D.Y. Chiu, K.M. Chen, M.H. Cho, and S.C. Wang. A novel process-controlled-monitor structure suitable for RF CMOS characterization. In *IEEE MTT-S International Microwave Symposium Digest of Technical Papers*, pages 1299–1302, June 2005.
- [28] C. Cho, D. Kim, J. Kim, J.-O. Plouchart, and R. Trzcinski. Statistical framework for technology-model-circuit co-design and convergence. In *Proceedings of* the ACM/IEEE Design Automation Conference, pages 503–508, June 2007.
- [29] M.-H. Cho, R. Lee, A.-S. Peng, D. Chen, C.-S. Yeh, and L.-K. Wu. Miniature RF test structure for on-wafer device testing and in-line process monitoring. *IEEE Transactions on Electron Devices*, 55(1):462–465, January 2008.
- [30] J.A. Croon, G. Storms, S. Winkelmeier, I. Pollentier, M. Ercken, S. Decoutere, W. Sansen, and H.E. Maes. Line edge roughness: characterization, modeling and impact on device behavior. In *Proceedings of International Electron Devices Meeting (IEDM)*, pages 307–310, 2002.
- [31] J.P. de Gyvez and H.P. Tuinhout. Threshold voltage mismatch and intra-die leakage current in digital CMOS circuits. *IEEE Journal of Solid-State Circuits*, 39(1):157–168, 2004.
- [32] A. Dharchoudhury and S.M. Kang. Worst-case analysis and optimization of vlsi circuit performances. *IEEE Transactions on Computer-Aided Design of Integrated Circuits and Systems*, 14(4):481–492, April 1995.

- [33] C.H. Doan, S. Emami, A.M. Niknejad, and R.W. Brodersen. Millimeter-wave CMOS design. *IEEE Journal of Solid-State Circuits*, 40(1):144–155, January 2005.
- [34] N. Drego, A. Chandrakasan, and D. Boning. A test-structure to efficiently study threshold voltage variation in large MOSFET arrays. In Proceedings of the International Symposium on Quality Electronic Design, pages 281–286, 2007.
- [35] P.G. Drennan and C.C. McAndrew. Understanding MOSFET mismatch for analog design. *IEEE Journal of Solid-State Circuits*, 38(3):450-456, March 2003.
- [36] R.W. Dutton, D.A. Divekar, A.G. Gonzalez, S.E. Hansen, and D.A. Antoniadis. Correlation of fabrication process and electrical device parameter variations. *IEEE Journal of Solid-State Circuits*, 12(4):349–355, Aug 1977.
- [37] N. Fong, J. Kim, J.-O. Plouchart, N. Zamdmer, D. Liu, L. Wagner, C. Plett, and G. Tarr. A low-voltage 40-GHz complementary VCO with 15% frequency tuning range in SOI CMOS technology. *IEEE Journal of Solid-State Circuits*, 39(5):841-846, May 2004.
- [38] N.H.W. Fong, J.-O. Plouchart, N. Zamdmer, D. Liu, L.F. Wagner, C. Plett, and N.G. Tarr. Design of wide-band CMOS VCO for multiband wireless LAN applications. *IEEE Journal of Solid-State Circuits*, 38(8):1333–1342, August 2003.
- [39] L.M. Franca-Neto, R.E. Bishop, and B.A. Bloechel. 64 GHz and 100 GHz VCOs in 90 nm CMOS using optimum pumping method. In *Proceedings of IEEE International Solid-State Circuits Conference (ISSCC)*, pages 444–538, February 2004.
- [40] P. Friedberg, Y. Cao, J. Cain, R. Wang, J. Rabaey, and C. Spanos. Modeling within-die spatial correlation effects for process-design co-optimization. In Pro-

ceedings of the International Symposium on Quality of Electronic Design, pages 516–521, March 2005.

- [41] M. Garg, S.S. Suryagandh, and J. C.S. Woo. Scaling impact on analog performance of sub-100nm MOSFETs for mixed mode applications. In *Proceedings of Conference on European Solid-State Device Research*, pages 371–374, September 2003.
- [42] K.G.M.V. Gettings. Extraction of variation sources due to layout practices. Master's thesis, Massachusetts Institute of Technology, June 2002.
- [43] G.G.E. Gielen and R.A. Rutenbar. Computer-aided design of analog and mixedsignal integrated circuits. *Proceedings of the IEEE*, 88(12):1825–1854, Dec 2000.
- [44] A. Hajimiri and T.H. Lee. Design issues in CMOS differential LC oscillators. IEEE Journal of Solid-State Circuits, 34(5):717–724, May 1999.
- [45] D.A. Hitko, T. Hussain, J.F. Jensen, Y. Royter, S.L. Morton, D.S. Matthews, R.D. Rajavel, I. Milosavljevlc, C.H. Fields, S. Thomas III, A. Kurdoghllan, Z. Lao, K. Elliott, and M. Sokolfch. A low power (45mW/latch) static 150GHz CML divider. In Proceedings of IEEE Compound Semiconductor Integrated Circuit Symposium, pages 167–170, October 2004.
- [46] K.-L. Hsiung, S.-J. Kim, and S. Boyd. Tractable approximate robust geometric programming. Optimization and Engineering, 9(2):95–118, 2008.
- [47] C.-M. Hsu, M.Z. Straayer, and M.H. Perrott. A low-noise, wide-BW 3.6GHz digital $\Delta \Sigma$ fractional-N frequency synthesizer with a noise-shaping time-todigital converter and quantization noise cancellation. In *Proceedings of IEEE International Solid-State Circuits Conference (ISSCC)*, February 2008.
- [48] P.-C. Huang, R.-C. Liu, H.-Y. Chang, C.-S. Lin, M.-F. Lei, H. Wang, C.-Y. Su, and C.-L. Chang. A 131 GHz push-push VCO in 90-nm CMOS technology. In Proceedings of IEEE Radio Frequency integrated Circuits (RFIC) Symposium Digest of Papers, pages 613–616, June 2005.

- [49] P.-C. Huang, M.-D. Tsai, G.D. Vendelin, H. Wang, C.-H. Chen, and C.-S. Chang. A low-power 114-GHz push.push CMOS VCO using LC source degeneration. *IEEE Journal of Solid-State Circuits*, 42(6):1230–1239, 2007.
- [50] I.T. Jolliffe. Principal Component Analysis. Springer-Verlag New York, Inc., 2002.
- [51] W. Kahlil, B. Bakkaloglu, and S. Kiaei. A self-calibrated on-chip phase noise measurement circuit with -76dBc single-tone sensitivity at 100kHz offset. In Proceedings of IEEE International Solid-Sate Circuits Conference, pages 546– 547, February 2007.
- [52] M. Kayal, R.T.L. Saez, and M. Declercq. An automatic offset compensation technique applicable to existing operational amplifier core cell. In *Proceedings* of the IEEE Custom Integrated Circuits Conference, pages 419–422, May 1998.
- [53] M. Ketchen, M. Bhushan, and D.J. Pearson. High speed test structures for in-line process monitoring and model calibration. In *Proceedings of the IEEE International Conference on Microelectronic Test Structures*, pages 33–38, 2005.
- [54] D. Kim, J. Kim, J.-O. Plouchart, C. Cho, R. Trzcinski, S. Lee, M. Kumar, C. Norris, J.-S. Rieh, G. Freeman, and D. Ahlgren. Manufacturable parasiticaware circuit-level fets in 65-nm SOI CMOS technology. *Electron Device Letters*, *IEEE*, 28(6):520–522, June 2007.
- [55] D.D. Kim, C. Cho, and J. Kim. Scalable statistical measurement and estimation of a mmwave cml static divider sensitivity in 65nm soi cmos. In Proceedings of IEEE Radio Frequency Integrated Circuits (RFIC) Symposium, pages 625–628, April 2008.
- [56] K.J. Kuhn. Reducing variation in advanced logic technologies: Approaches to process and design for manufacturability of nanoscale cmos. *IEEE International Electron Devices Meeting*, pages 471–474, December 2007.

- [57] E. Laskin, S.T. Nicolson, P. Chevalier, A. Chantre, B. Sautreuil, and S.P. Voinigescu. Low-power, low-phase noise SiGe HBT static frequency divider topologies up to 100 GHz. *Bipolar/BiCMOS Circuits and Technology Meeting*, pages 1–4, October 2006.
- [58] H.-S. Lee, D.A. Hodges, and P.R. Gray. A self-calibrating 15 bit CMOS A/D converter. *IEEE Journal of Solid-State Circuits*, 19(6):813–819, Dec 1984.
- [59] J. Lee and B. Razavi. A 40 GHz frequency divider in 0.18µm CMOS technology. IEEE Journal of Solid-State Circuits, 39(4):594–601, 2004.
- [60] S. Lee, B. Jagannathan, S. Narasimha, A. Chou, N. Zamdmer, J. Johnson, R. Williams, L. Wagner, J. Kim, J.-O. Plouchart, J. Pekarik, S. Springer, and G. Freeman. Record RF performance of 45-nm SOI CMOS technology. In Proceedings of International Electron Devices Meeting (IEDM), pages 255–258, December 2007.
- [61] S. Lee, J. Kim, D. Kim, B. Jagannathan, C. Cho, J. Johnson, B. Dufrene, N. Zamdmer, L. Wagner, R. Williams, D. Fried, K. Rim, J. Pekarik, S. Springer, J.-O. Plouchart, and G. Freeman. SOI CMOS technology with 360GHz f_T NFET, 260GHz f_T PFET, and record circuit performance for millimeter-wave digital and analog System-on-Chip applications. In *Proceedings of IEEE Symposium* on VLSI Technology, pages 54–55, June 2007.
- [62] H. Li and H.-M. Rein. Millimeter-wave VCOs with wide tuning range and low phase noise, fully integrated in a SiGe bipolar production technology. *IEEE Journal of Solid-State Circuits*, 38(2):184–191, February 2003.
- [63] X. Li, P. Gopalakrishnan, Y. Xu, and L.T. Pileggi. Robust analog/RF circuit design with projection-based performance modeling. *IEEE Transactions on Computer-Aided Design of Integrated Circuits and Systems*, 26(1):2–15, January 2007.

- [64] X. Li, J. Le, P. Gopalakrishnan, and L.T. Pileggi. Asymptotic probability extraction for nonnormal performance distributions. *IEEE Transactions on Computer-Aided Design of Integrated Circuits and Systems*, 26(1):16–37, January 2007.
- [65] X. Li, B. Taylor, Y. Chien, and L.T. Pileggi. Adaptive post-silicon tuning for analog circuits: concept, analysis and optimization. In Proceedings of the IEEE/ACM International Conference on Computer-Aided Design, pages 450– 457, 2007.
- [66] X. Li, J. Wang, L.T. Pileggi, T.-S. Chen, and W. Chiang. Performance-centering optimization for system-level analog design exploration. In *Proceedings of the IEEE/ACM International Conference on Computer-Aided Design*, pages 422– 429, 2005.
- [67] D. Lim, J. Kim, J.-O. Plouchart, C. Cho, D. Kim, R. Trzcinski, and D. Boning. Performance variability of a 90GHz static CML frequency divider in 65nm SOI CMOS. In Proceedings of IEEE International Solid-State Circuits Conference (ISSCC), pages 542-621, February 2007.
- [68] R.-C. Liu, H.-Y. Chang, C.-H. Wang, and H. Wang. A 63 GHz VCO using a standard 0.25 μm CMOS process. In Proceedings of IEEE International Solid-State Circuits Conference (ISSCC), pages 446–447, February 2004.
- [69] M. Mani, A. Singh, and M. Orshansky. Joint design-time and post-silicon minimization of parametric yield loss using adjustable robust optimization. In Proceedings of the IEEE/ACM International Conference on Computer-Aided Design, pages 19-26, November 2006.
- [70] A. Matsuzawa. Nano-scale CMOS and low voltage analog to digital converter design challenges. In Proceedings of International Conference on Solid-State and Integrated Circuit Technology, pages 1676–1679, 2006.

- [71] V. Mehrotra, S. Nassif, D. Boning, and J. Chung. Modeling the effects of manufacturing variation on high-speed microprocessor interconnect performance. In *Proceedings of International Electron Devices Meeting (IEDM)*, pages 767–770, Dec 1998.
- [72] V. Mehrotra, S.L. Sam, D. Boning, A. Chandrakasan, R. Vallishayee, and S. Nassif. A methodology for modeling the effects of systematic within-die interconnect and device variation on circuit performance. In *Proceedings of the* ACM/IEEE Design Automation Conference, pages 172–175, 2000.
- [73] C. Michael and M. Ismail. Statistical modeling of device mismatch for analog MOS integrated circuits. *IEEE Journal of Solid-State Circuits*, 27(2):154–166, February 1992.
- [74] A. Mirzaei, M. Heidari, R. Bagheri, S. Chehrazi, and A.A. Abidi. Injectionlocked frequency dividers based on ring oscillators with optimum injection for wide lock range. In *Proceedings of IEEE Symposium on VLSI Circuits Digest* of Technical Papers, pages 174–175, June 2006.
- [75] A. Mirzaei, M.E. Heidari, R. Bagheri, and A.A. Abidi. Multi-phase injection widens lock range of ring-oscillator-based frequency dividers. *IEEE Journal of Solid-State Circuits*, 43(3):656–671, March 2008.
- [76] M. Mokhtari, C. Fields, R.D. Rajavel, M. Sokolich, J.F. Jensen, and W.E. Stanchina. 100+ GHz static divide-by-2 circuit in InP-DHBT technology. *IEEE Journal of Solid-State Circuits*, 38(9):1540–1544, 2003.
- [77] B. Murmann. Analog design in sub-100nm technologies. IEEE Half-Day Symposiumon Process, Device and Design Challenges in Analog and Mixed Signal Applications, May 2006.
- [78] B. Murmann. A/D converter trends: Power dissipation, scaling and digitally assisted architectures. In Proceedings of IEEE Custom Integrated Circuits Conference, September 2008.

- [79] S. Nassif. Modeling and analysis of manufacturing variations. In Proceedings of IEEE Custom Integrated Circuits Conference, pages 223–228, May 2001.
- [80] P. Oldiges, Q. Lin, K. Petrillo, M. Sanchez, M. Ieong, and M. Hargrove. Modeling line edge roughness effects in sub 100 nanometer gate length devices. *International Conference on Simulation of Semiconductor Processes and Devices*, pages 131-134, 2000.
- [81] M. Orshansky, S.R. Nassif, and D. Boning. Design for Manufacturability and Statistical Design, A Constructive Approach. Springer, 2008.
- [82] M. Orshansky, C. Spanos, and C. Hu. Circuit performance variability decomposition. In Proceedings of International Workshop on Statistical Metrology, pages 10–13, 1999.
- [83] L.-T. Pang and B. Nikolic. Impact of layout on 90nm CMOS process parameter fluctuations. In Proceedings of IEEE Symposium on VLSI Circuits Digest of Technical Papers, pages 69–70, 2006.
- [84] L.C. Parrillo, G.W. Reutlinger, R.S. Payne, A.R. Tretola, and R.T. Kraetsch. The sensitivity of transistor gain to processing variations in an all implanted bipolar technology. In *Proceedings of International Electron Devices Meeting* (*IEDM*), volume 23, pages 265–265, 1977.
- [85] M.J.M. Pelgrom, A.C.J. Duinmaijer, and A.P.G. Welbers. Matching properties of mos transistors. *IEEE Journal of Solid-State Circuits*, 24(5):1433–1439, October 1989.
- [86] L.T. Pileggi, H. Schmit, A.J. Strojwas, P. Gopalakrishnan, V. Kheterpal, A. Koorapaty, C. Patel, V. Rovner, and K.Y. Tong. Exploring regular fabrics to optimize the performance-cost trade-off. In *Proceedings of the ACM/IEEE Design Automation Conference*, pages 782–787, 2003.
- [87] J.-O. Plouchart, J. Kim, V. Karam, R. Trzcinski, and J. Gross. Performance variability of a 66GHz static CML divider in 90nm CMOS. In *Proceedings of*

IEEE International Solid-State Circuits Conference (ISSCC), pages 2142–2151, February 2006.

- [88] J.-O. Plouchart, J. Kim, H. Recoules, N. Zamdmer, Y. Tan, M. Sherony, A. Ray, and L. Wagner. A power-efficient 33 GHz 2:1 static frequency divider in 0.12μm SOI CMOS. In Proceedings of IEEE Radio Frequency Integrated Circuits (RFIC) Symposium, pages 329–332, June 2003.
- [89] H.R. Rategh and T.H. Lee. Superharmonic injection-locked frequency dividers. *IEEE Journal of Solid-State Circuits*, 34(6):813–821, Jun 1999.
- [90] H.R. Rategh, H. Samavati, and T.H. Lee. A 5 GHz, 1 mW CMOS voltage controlled differential injection locked frequency divider. In *Proceedings of the IEEE Custom Integrated Circuits Conference*, pages 517–520, 1999.
- [91] B. Razavi. CMOS technology characterization for analog and RF design. IEEE Journal of Solid-State Circuits, 34(3):268-276, March 1999.
- [92] B. Razavi. A study of injection locking and pulling in oscillators. IEEE Journal of Solid-State Circuits, 39(9):1415–1424, September 2004.
- [93] A. Rylyakov and T. Zwick. 96-GHz static frequency divider in SiGe bipolar technology. *IEEE Journal of Solid-State Circuits*, 39(10):1712–1715, October 2004.
- [94] A. Seifi, K. Ponnambalam, and J. Vlach. A unified approach to statistical design centering of integrated circuits with correlated parameters. *IEEE Transactions* on Circuits and Systems I: Fundamental Theory and Applications, 46(1):190– 196, Jan 1999.
- [95] U. Singh and M. Green. Dynamics of high-frequency CMOS dividers. In Proceedings of IEEE International Symposium on Circuits and Systems, volume 5, pages 421–424, 2002.

- [96] K. Skadron, M.R. Stan, W. Huang, S. Velusamy, K. Sankaranarayanan, and D. Tarjan. Temperature-aware microarchitecture. SIGARCH Computer Architecture News, 31(2):2–13, 2003.
- [97] B.E. Stine, D.S. Boning, and J.E. Chung. Analysis and decomposition of spatial variation in integrated circuit processes and devices. *IEEE Transactions on Semiconductor Manufacturing*, 10(1):24–41, February 1997.
- [98] H. Su, F. Liu, A. Devgan, E. Acar, and S. Nassif. Full chip leakage estimation considering power supply and temperature variations. In Proceedings of the International Symposium on Low power Electronics and Design, pages 78–83, 2003.
- [99] M. Tiebout, H.-D. Wohlmuth, and W. Simburger. A 1 V 51GHz fully-integrated VCO in 0.12 μm CMOS. In Proceedings of IEEE International Solid-State Circuits Conference (ISSCC), pages 300–468, February 2002.
- [100] K.Y. Tong, V. Rovner, L.T. Pileggi, and V. Kheterpal. Design methodology of regular logic bricks for robust integrated circuits. In *International Conference* on Computer Design, pages 162–167, October 2006.
- [101] S. Trotta, H. Knapp, T.F. Meister, K. Aufinger, J. Bock, W. Simburger, and A.L. Scholtz. 110-GHz static frequency divider in SiGe bipolar technology. In Proceedings of IEEE Compound Semiconductor Integrated Circuit Symposium, pages 291-294, October 2005.
- [102] J. Tschanz, K. Bowman, and V. De. Variation-tolerant circuits: circuit solutions and techniques. In Proceedings of the ACM/IEEE Design Automation Conference, pages 762-73, June 2005.
- [103] J.W. Tschanz, J.T. Kao, S.G. Narendra, R. Nair, D.A. Antoniadis, A.P. Chandrakasan, and V. De. Adaptive body bias for reducing impacts of die-to-die and within-die parameter variations on microprocessor frequency and leakage. *IEEE Journal of Solid-State Circuits*, 37(11):1396–1402, November 2002.

- [104] H. Wang, K.W. Chang, L.T. Tran, J.C. Cowles, T.R. Block, E.W. Lin, G.S. Dow, A.K. Oki, D.C. Streit, and B.R. Allen. Low phase noise millimeter-wave frequency sources using InP-based HBT MMIC technology. *IEEE Journal of Solid-State Circuits*, 31(10):1419–1425, October 1996.
- [105] Z. Wang and S.W. Director. An efficient yield optimization method using a two step linear approximation of circuit performance. In *Proceedings of the IEEE European Design and Test Conference*, pages 567–571, March 1994.
- [106] X. Xie. Physical Understanding and Modeling of Chemical Mechanical Planarization in Dielectric Materials. PhD thesis, Massachusetts Institute of Technology, June 2007.
- [107] J. Xiong, V. Zolotov, and L. He. Robust extraction of spatial correlation. IEEE Transactions on Computer-Aided Design of Integrated Circuits and Systems, 26(4):619-631, April 2007.
- [108] Y. Xu, X. Li, K.-L. Hsiung, S. Boyd, and I. Nausieda. OPERA: optimization with ellipsoidal uncertainty for robust analog ic design. In *Proceedings of the* ACM/IEEE Design Automation Conference, pages 632–637, June 2005.
- [109] J.C. Zhang and M.A. Styblinski. Yield and Variability Optimization of Integrated Circuits. Kluwer Academic Publishers, 1995.
- [110] L. Zhang, W. Chen, Y. Hu, J.A. Gubner, and C.C.-P. Chen. Correlationpreserved non-Gaussian statistical timing analysis with quadratic timing model. In Proceedings of the ACM/IEEE Design Automation Conference, pages 83–88, June 2005.